### IN THE SUPREME COURT OF THE STATE OF NEVADA

JAMES J. COTTER, JR., derivatively on behalf of Reading International, Inc.,  Appellant, v.	Electronically Filed Aug 30 2019 12:34 p.m Supreme Collita Beste No B750/53 Consolidate Glorito Case None Court 76981, 77648 & 77733
DOUGLAS MCEACHERN, EDWARD KANE, JUDY CODDING, WILLIAM GOULD, MICHAEL WROTNIAK, and nominal defendant READING INTERNATIONAL, INC., A NEVADA CORPORATION  Respondents.	District Court Case No. A-15-719860-B  Coordinated with: Case No. P-14-0824-42-E

Appeal (77648 & 76981)

Eighth Judicial District Court, Dept. XI The Honorable Elizabeth G. Gonzalez

JOINT APPENDIX TO OPENING BRIEFS FOR CASE NOS. 77648 & 76981 Volume XIII JA3059 – JA3308

Steve Morris, Esq. (NSB #1543) Akke Levin, Esq. (NSB #9102) Morris Law Group 411 E. Bonneville Ave., Ste. 360 Las Vegas, NV 89101 Telephone: (702) 474-9400

Attorneys for Appellant James J. Cotter, Jr.

Date	Description	Vol.#	Page Nos.
2015-06-12	Complaint	I	JA1-JA31
2015-06-18	Amended AOS - Douglas	I	JA32-JA33
	McEachern		***
2015-06-18	Amended AOS - Edward Kane	I	JA34-JA35
2015-06-18	Amended AOS - Ellen Cotter	I	JA36-JA37
2015-06-18	Amended AOS - Guy Adams	I	JA38-JA39
2015-06-18	Amended AOS - Margaret Cotter	I	JA40-JA41
2015-06-18	Amended AOS - RDI	I	JA42-JA43
2015-06-18	Amended AOS – Timothy Storey	I	JA44-JA45
2015-06-18	Amended AOS – William Gould	I	JA46-JA47
2015-08-10	Motion to Dismiss Complaint	I	JA48-JA104
2015-08-20	Reading International, Inc. ("RDI")'s Joinder to Margaret Cotter, Ellen Cotter, Douglas McEachern, Guy Adams, & Edward Kane ("Individual Defendants") Motion to Dismiss Complaint	I	JA105-JA108
2015-08-28	T2 Plaintiffs' Verified Shareholder Derivative Complaint	Ι	JA109-JA126
2015-08-31	RDI's Motion to Compel Arbitration	Ι	JA127-JA148
2015-09-03	Individual Defendants' Motion to Dismiss Complaint	Ι	JA149-JA237
2015-10-06	Transcript of 9-10-15 Hearing on Defendants' Motion to Dismiss & Plaintiff Cotter Jr. ("Cotter Jr.")'s Motion for Preliminary Injunction	I, II	JA238-JA256
2015-10-12	Order Denying RDI's Motion to Compel Arbitration	II	JA257-JA259
2015-10-19	Order Re Motion to Dismiss Complaint	II	JA260-JA262
2015-10-22	First Amended Verified Complaint	II	JA263-JA312
2015-11-10	Scheduling Order and Order Setting Civil Jury Trial, Pre-Trial Conference and Calendar Call	II	JA313-JA316

Date	Description	Vol.#	Page Nos.
2016-02-12	T2 Plaintiffs' First Amended Complaint	II	JA317-JA355
2016-02-23	Transcript of 2-18-16 Hearing on Motion to Compel & Motion to File Document Under Seal	II	JA356-JA374
2016-03-14	Individual Defendants' Answer to Cotter's First Amended Complaint	II	JA375-JA396
2016-03-29	RDI's Answer to Cotter, Jr.'s First Amended Complaint	II	JA397-JA418
2016-03-29	RDI's Answer to T2 Plaintiffs' First Amended Complaint	II	JA419-JA438
2016-04-05	Codding and Wrotniak's Answer to T2 Plaintiffs' First Amended Complaint	II	JA439-JA462
2016-06-21	Stipulation and Order to Amend Deadlines in Scheduling Order	II	JA463-JA468
2016-06-23	Transcript of 6-21-16 Hearing on Defendants' Motion to Compel & Motion to Disqualify T2 Plaintiffs	II	JA469-JA493
2016-08-11	Transcript of 8-9-16 Hearing on Cotter Jr.'s Motion for Partial Summary Judgment, Motion to Compel & Motion to Amend	II, III	JA494-JA518
2016-09-02	Cotter Jr.'s Second Amended Verified Complaint	III	JA519-JA575
2016-09-23	Defendant William Gould ("Gould")'s MSJ	III, IV, V, VI	JA576-JA1400
2016-09-23	MIL to Exclude Expert Testimony of Steele, Duarte-Silva, Spitz, Nagy, & Finnerty	VI	JA1401-JA1485
2016-09-23	Individual Defendants' Motion for Partial Summary Judgment (No. 1) Re: Plaintiff's Termination and Reinstatement Claims ("Partial MSJ No. 1)	VI, VII, VIII, IX	JA1486-JA2216 (FILED UNDER SEAL JA2136A-D)

Date	Description	Vol.#	Page Nos.
2016-09-23	Individual Defendants' Motion for Partial Summary Judgment (No. 2) Re: The Issue of Director Independence ("Partial MSJ No. 2")	IX, X	JA2217-JA2489 (FILED UNDER SEAL JA2489A-HH)
2016-09-23	Individual Defendants' Motion for Partial Summary Judgment (No. 3) On Plaintiff's Claims Related to the Purported Unsolicited Offer ("Partial MSJ No. 3")	X, XI	JA2490-JA2583
2016-09-23	Individual Defendants' Motion for Partial Summary Judgment (No. 4) On Plaintiff's Claims Related to the Executive Committee ("Partial MSJ No. 4")	XI	JA2584-JA2689
2016-09-23	Individual Defendants' Motion for Partial Summary Judgment (No. 5) On Plaintiff's Claims Related to the Appointment of Ellen Cotter as CEO ("Partial MSJ No. 5")	XI, XII	JA2690-JA2860
2016-09-23	Individual Defendants' Motion for Partial Summary Judgment (No. 6) Re Plaintiff's Claims Re Estate's Option Exercise, Appointment of Margaret Cotter, Compensation Packages of Ellen Cotter and Margaret Cotter, and related claims Additional Compensation to Margaret Cotter and Guy Adams ("Partial MSJ No. 6")	XII, XIII, XIV	JA2861-JA3336
2016-09-23	Cotter Jr.'s Motion for Partial Summary Judgment ("MPSJ")	XIV, XV	JA3337-JA3697
2016-10-03	Order Granting Cotter Jr.'s Motion to Compel Production of Documents & Communications Re the Advice of Counsel Defense	XV	JA3698-JA3700

Date	Description	Vol.#	Page Nos.
2016-10-03	Order Re Cotter Jr.'s Motion to Permit Certain Discovery re Recent "Offer"	XV	JA3701-JA3703
2016-10-03	RDI's Joinder to MIL to Exclude Expert Testimony	XV	JA3704-JA3706
2016-10-03	RDI's Joinder to Individual Defendants' Partial-MSJ No. 1	XV	JA3707-JA3717
2016-10-03	RDI's Joinder to Individual Defendants' Partial MSJ No. 2	XV	JA3718-JA3739
2016-10-03	RDI's Joinder to Individual Defendants' Partial MSJ No. 3	XV	JA3740-JA3746
2016-10-03	RDI's Joinder to Individual Defendants' Partial MSJ No. 4	XV	JA3747-JA3799
2016-10-03	RDI's Joinder to Individual Defendants' Partial MSJ No. 5	XV	JA3800-JA3805
2016-10-03	RDI's Joinder to Individual Defendants' Partial MSJ No. 6	XV, XVI	JA3806-JA3814
2016-10-13	Individual Defendants' Opposition to Cotter Jr.'s MPSJ	XVI	JA3815-JA3920
2016-10-13	RDI's Joinder to Individual Defendants' Opposition to Cotter Jr.'s MPSJ	XVI	JA3921-JA4014
2016-10-13	Cotter Jr.'s Opposition to Gould's MSJ	XVI	JA4015-JA4051
2016-10-13	Cotter Jr.'s Opposition to Partial MSJ No. 1	XVI, XVII	JA4052-JA4083
2016-10-13	Cotter, Jr.'s Opposition to Partial MSJ No. 2	XVII	JA4084-JA4111
2016-10-13	Cotter, Jr.'s Opposition to Partial MSJ No. 6	XVII	JA4112-JA4142
2016-10-17	Cotter Jr.'s Appendix of Exhibits ISO Opposition to Individual Defendants' Partial MSJ No. 1	XVII, XVIII	JA4143-JA4311 (FILED UNDER SEAL JA4151A-C)

Date	Description	Vol.#	Page Nos.
2016-10-17	Cotter Jr.'s Appendix of Exhibits ISO Opposition to Individual Defendants' Partial MSJ No. 2	XVIII	JA4312-JA4457
2016-10-17	Cotter Jr.'s Appendix of Exhibits ISO Opposition to Gould's MSJ	XVIII	JA4458-JA4517
2016-10-21	Individual Defendants' Reply ISO of Partial MSJ No. 1	XVIII	JA4518-JA4549
2016-10-21	Individual Defendants' Reply ISO Partial MSJ No. 2	XVIII, XIX	JA4550-JA4567
2016-10-21	RDI's Reply ISO Individual Defendants' Partial MSJ No. 1	XIX	JA4568-JA4577
2016-10-21	RDI's Reply ISO Individual Defendants' Partial MSJ No. 2	XIX	JA4578-JA4588
2019-10-21	RDI's Consolidated Reply ISO Individual Defendants' Partial MSJ Nos. 3, 4, 5 & 6	XIX	JA4589-JA4603
2016-10-21	RDI's Reply ISO Gould's MSJ	XIX	JA4604-JA4609
2016-10-21	Gould's Reply ISO MSJ	XIX	JA4610-JA4635
2016-10-21	Declaration of Bannett ISO Gould's Reply ISO MSJ	XIX	JA4636-JA4677
2016-10-21	Individual Defendants' Reply ISO Partial MSJ Nos. 3, 4, 5, and 6	XIX	JA4678–JA4724
2016-10-26	Individual Defendants' Objections to Declaration of Cotter, Jr. Submitted in Opposition to Partial MSJs	XIX	JA4725-JA4735
2016-11-01	Transcript of 10-27-16 Hearing on Motions	XIX, XX	JA4736-JA4890
2016-12-20	RDI's Answer to Cotter Jr.'s Second Amended Complaint	XX	JA4891-JA4916
2016-12-21	Order Re Individual Defendants' Partial MSJ Nos. 1–6 and MIL to Exclude Expert Testimony	XX	JA4917-JA4920
2016-12-22	Notice of Entry of Order Re Partial MSJ Nos. 1-6 and MIL to Exclude Expert Testimony	XX	JA4921-JA4927

Date	Description	Vol.#	Page Nos.
2017-10-04	First Amended Order Setting Civil Jury Trial, Pre-Trial Conference, and Calendar Call	XX	JA4928-JA4931
2017-10-11	Individual Defendants' Motion for Evidentiary Hearing Re Cotter Jr.'s Adequacy as Derivative Plaintiff	XX	JA4932-JA4974
2017-10-17	Gould's Joinder to Motion for Evidentiary Hearing re Cotter Jr.'s Adequacy as Derivative Plaintiff	XX	JA4975-JA4977
2017-10-18	RDI's Joinder to Motion for Evidentiary Hearing re Cotter Jr.'s Adequacy as Derivative Plaintiff	XX	JA4978-JA4980
2017-11-09	Individual Defendants' Supplement to Partial MSJ Nos. 1, 2, 3, 5, and 6	XX	JA4981-JA5024
2017-11-21	RDI's Joinder to Individual Defendants' Supplement to Partial MSJ Nos. 1, 2, 3, 5 & 6	XX	JA5025-JA5027
2017-11-27	Transcript of 11-20-17 Hearing on Motion for Evidentiary Hearing re Cotter Jr.'s Adequacy & Motion to Seal	XX	JA5028-JA5047
2017-11-28	Individual Defendants' Answer to Cotter Jr.'s Second Amended Complaint	XX, XXI	JA5048-JA5077
2017-12-01	Gould's Request For Hearing on Previously-Filed MSJ	XXI	JA5078-JA5093
2017-12-01	Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 1 & 2 & Gould MSJ	XXI	JA5094-JA5107
2017-12-01	Declaration of Levin ISO Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 1 & 2 & Gould MSJ	XXI	JA5108-JA5118

Date	Description	Vol.#	Page Nos.
2017-12-01	Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 2 & 5 & Gould MSJ	XXI	JA5119-JA5134
2017-12-01	Declaration of Levin ISO Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 2 & 5 & Gould MSJ	XXI	JA5135-JA5252
2017-12-01	Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 2 & 6 & Gould MSJ	XXI	JA5253-JA5264
2017-12-01	Declaration of Levin ISO Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 2 & 6 & Gould MSJ	XXI	JA5265-JA5299
2017-12-01	Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 2 & 3 & Gould MSJ	XXI, XXII	JA5300-JA5320
2017-12-01	Declaration of Levin ISO Cotter Jr.'s Supplemental Opposition to So-Called MSJ Nos. 2 & 3 & Gould MSJ	XXII	JA5321-JA5509
2017-12-04	Individual Defendants' Reply ISO Renewed Partial MSJ Nos. 1 & 2	XXII	JA5510-JA5537
2017-12-04	Gould's Supplemental Reply ISO of MSJ	XXII	JA5538-JA5554
2017-12-05	Declaration of Bannett ISO Gould's Supplemental Reply ISO MSJ	XXII, XXIII	JA5555-JA5685
2017-12-08	Joint Pre-Trial Memorandum	XXIII	JA5686-JA5717
2017-12-11	Transcript of 12-11-2017 Hearing on [Partial] MSJs, MILs, and Pre-Trial Conference	XXIII	JA5718-JA5792
2017-12-19	Cotter Jr.'s Motion for Reconsideration of Ruling on Partial MSJ Nos. 1, 2 & 3 and Gould's MSJ on OST ("Motion for Reconsideration")	XXIII, XXIV	JA5793-JA5909

Date	Description	Vol.#	Page Nos.
2017-12-26	Individual Defendants' Opposition to Cotter Jr.'s Motion For Reconsideration	XXIV	JA5910-JA5981
2017-12-27	Gould's Opposition to Cotter Jr.'s Motion for Reconsideration	XXIV	JA5982-JA5986
2017-12-27	Declaration of Bannett ISO Gould's Opposition to Cotter Jr.'s Motion for Reconsideration	XXIV, XXV	JA5987-JA6064
2017-12-28	Order Re Individual Defendants' Partial MSJs, Gould's MSJ, and MILs	XXV	JA6065-JA6071
2017-12-28	Cotter Jr.'s Motion to Stay on OST	XXV	JA6072-JA6080
2017-12-29	Notice of Entry of Order Re Individual Defendants' Partial MSJs, Gould's MSJ, and MIL	XXV	JA6081-JA6091
2017-12-29	Cotter Jr.'s Motion for Rule 54(b) Certification and Stay on OST	XXV	JA6092-JA6106
2017-12-29	Transcript of 12-28-17 Hearing on Motion for Reconsideration and Motion for Stay	XXV	JA6107-JA6131
2018-01-02	Individual Defendants' Opposition to Cotter Jr.'s Motion for Rule 54(b) Certification and Stay	XXV	JA6132-JA6139
2018-01-03	RDI's Joinder to Individual Defendants' Opposition to Cotter Jr.'s Motion for Rule 54(b) Certification and Stay	XXV	JA6140-JA6152
2018-01-03	RDI's Errata to Joinder to Individual Defendants' Opposition to Motion for Rule 54(b) Certification and Stay	XXV	JA6153-JA6161
2018-01-03	RDI's Motion to Dismiss for Failure to Show Demand Futility	XXV	JA6162-JA6170
2018-01-03	Cotter Jr.'s Reply ISO Motion for Rule 54(b) Certification and Stay	XXV	JA6171-JS6178

Date	Description	Vol.#	Page Nos.
2018-01-04	Order Granting Cotter Jr.'s Motion for Rule 54(b) Certification	XXV	JA6179-JA6181
2018-01-04	Notice of Entry of Order Granting Cotter Jr.'s Motion for Rule 54(b) Certification	XXV	JA6182-JA6188
2018-01-04	Order Denying Cotter Jr.'s Motion for Reconsideration and Stay	XXV	JA6189-JA6191
2018-01-04	Adams and Cotter sisters' Motion for Judgment as a Matter of Law	XXV	JA6192-JA6224 (FILED UNDER SEAL
			JA6224A-F)
2018-01-05	Cotter Jr.'s Opposition to RDI's Motion to Dismiss for Failure to Show Demand Futility	XXV	JA6225-JA6228
2018-01-05	Cotter Jr.'s Opposition to Defendants' Motion for Judgment as a Matter of Law	XXV	JA6229-JA6238
2018-01-05	Declaration of Krum ISO Cotter Jr.'s Opposition to Motion for Judgment as a Matter of Law	XXV	JA6239-JA6244
2018-01-05	Transcript of 1-4-18 Hearing on Cotter Jr.'s Motion for Rule 54(b) Certification	XXV	JA6245-JA6263
2018-01-08	Transcript of Hearing on Demand Futility Motion and Motion for Judgment	XXV	JA6264-JA6280
2018-01-10	Transcript of Proceedings of 01-8- 18 Jury Trial–Day 1	XXV	JA6281-JA6294
2018-02-01	Cotter Jr.'s Notice of Appeal	XXV	JA6295-JA6297
2018-04-18	Cotter Jr.'s Motion to Compel (Gould)	XXV, XXVI	JA6298-JA6431

Date	Description	Vol.#	Page Nos.
2018-04-23	Cotter Jr.'s Motion for Omnibus Relief on OST	XXVI, XXVII	JA6432-JA6561 (FILED UNDER SEAL JA6350A; JA6513A-C)
2018-04-24	Gould's Opposition to Cotter Jr.'s Motion to Compel	XXVII	JA6562-JA6568
2018-04-24	Gould's Declaration ISO Opposition to Motion to Compel	XXVII	JA6569-JA6571
2018-04-24	Bannett's Declaration ISO Gould's Opposition to Motion to Compel	XXVII	JA6572-JA6581
2018-04-27	Cotter Jr.'s Reply ISO Motion to Compel (Gould)	XXVII	JA6582-JA6599
2018-04-27	RDI's Opposition to Cotter's Motion for Omnibus Relief	XXVII	JA6600-JA6698
2018-05-03	Transcript of 4-30-18 Hearing on Motions to Compel & Seal	XXVII	JA6699-JA6723
2018-05-04	Second Amended Order Setting Jury Trial, Pre-trial Conference, and Calendar Call	XXVII	JA6724-JA6726
2018-05-07	Transcript of 5-2-18 Hearing on Evidentiary Hearing	XXVII, XXVIII	JA6727-JA6815
2018-05-11	Cotter Jr.'s Opposition to RDI's Motion for Leave to File Motion	XXVIII	JA6816-JA6937
2018-05-15	Adams and Cotter sisters' Motion to Compel Production of Docs re Expert Fee Payments on OST	XXVIII, XXIX	JA6938-JA7078
2018-05-18	Cotter Jr.'s Opposition to Motion to Compel Production of Docs re Expert Fee Payments	XXIX	JA7079-JA7087
2018-05-18	Adams and Cotter sisters' Pre- Trial Memo	XXIX	JA7088-JA7135
2018-05-18	Cotter Jr.'s Pre-Trial Memo	XXIX	JA7136-JA7157

Date	Description	Vol.#	Page Nos.
2018-05-24	Transcript of 05-21-18 Hearing on Adams and Cotter sisters' Motion to Compel	XXIX	JA7158-JA7172
2018-06-01	Adams and Cotter sisters' Motion for Summary Judgment ("Ratification MSJ")	XXIX	JA7173-JA7221
2018-06-08	Cotter Jr.'s Motion to Compel on OST	XXIX, XXX, XXXI	JA7222-JA7568
2018-06-12	Cotter Jr.'s Motion for Relief Based on Noncompliance with Court's May 2, 2018 Rulings on OST ("Motion for Relief")	XXXI	JA7569-JA7607
2018-06-13	Cotter Jr.'s Opposition to Ratification MSJ	XXXI	JA7608-JA7797
2018-06-13	Cotter Jr.'s Opposition to RDI's Demand Futility Motion	XXXI, XXXII	JA7798-JA7840
2018-06-15	Adams and Cotter sisters' Reply ISO of Ratification MSJ	XXXII	JA7841-JA7874
2018-06-18	RDI's Combined Opposition to Cotter Jr.'s Motion to Compel & Motion for Relief	XXXII	JA7875-JA7927
2018-06-18	Adams and Cotter sisters' Joinder to RDI's Combined Opposition to Cotter Jr.'s Motion to Compel & Motion for Relief	XXXII, XXXIII	JA7928-JA8295
2018-06-18	Gould's Joinder to RDI's Combined Opposition to Cotter Jr.'s Motion to Compel & Motion for Relief	XXXIII	JA8296-JA8301
2018-06-18	Cotter Jr.'s Reply ISO Motion for Relief Re: 05-02-18 Rulings	XXXIII, XXXIV	JA8302-JA8342
2018-06-20	Transcript of 06-19-18 Omnibus Hearing on discovery motions and Ratification MSJ	XXXIV	JA8343-JA8394

Date	Description	Vol.#	Page Nos.
2018-07-12	Order Granting In Part Cotter Jr.'s Motion to Compel (Gould) & Motion for Relief	XXXIV	JA8395-JA8397
2018-07-12	Order Granting in Part Cotter Jr.'s Motion for Omnibus Relief & Motion to Compel	XXXIV	JA8398-JA8400
2018-08-14	Findings of Fact and Conclusions of Law and Judgment	XXXIV	JA8401-JA8411
2018-08-16	Notice of Entry of Findings of Fact and Conclusions of Law and Judgment	XXXIV	JA8412-JA8425
2018-08-24	Memorandum of Costs submitted by RDI for itself & the director defendants	XXXIV	JA8426-JA8446
2018-08-24	RDI's Appendix of Exhibits to Memorandum of Costs	XXXIV, XXXV, XXXVI	JA8447-JA8906
2018-09-05	Notice of Entry of SAO Re Process for Filing Motion for Attorney's Fees	XXXVI	JA8907-JA8914
2018-09-05	Cotter Jr.'s Motion to Retax Costs	XXXVI	JA8915-JA9018
2018-09-07	RDI's Motion for Attorneys' Fees	XXXVI, XXXVII	JA9019-JA9101
2018-09-12	RDI's Motion for Judgment in Its Favor	XXXVII	JA9102-JA9107
2018-09-13	Cotter Jr.'s Notice of Appeal	XXXVII	JA9108-JA9110
2018-09-14	RDI's Opposition to Cotter Jr.'s Motion to Retax Costs	XXXVII	JA9111-JA9219
2018-09-14	RDI's Appendix ISO Opposition to Motion to Retax ("Appendix") Part 1	XXXVII, XXXVIII, XXXIX	JA9220-JA9592
2018-09-14	RDI's Appendix, Part 2	XXXIX, XL, XLI	JA9593- JA10063
2018-09-14	RDI's Appendix, Part 3	XLI, XLII, XLIII	JA10064- JA10801

Date	Description	Vol.#	Page Nos.
2018-09-14	RDI's Appendix, Part 4	XLIII,	JA10802-
		XLIV	JA10898
2018-09-14	RDI's Appendix Part 5	XLIV,	JA10899-
		XLV	JA11270
2018-09-14	RDI's Appendix, Part 6	XLV,	JA11271-
		XLVI	JA11475
2018-09-14	RDI's Appendix, Part 7	XLVI,	
		XLVII,	JA11476-
		XLVIII,	JA12496
		XLIX, L	
2018-09-14	RDI's Appendix, Part 8	L, LI, LII	JA12497-
		L, Ll, Lll	JA12893
2018-09-14	Suggestion of Death of Gould	LII,	JA12894-
	Upon the Record	ын,	JA12896
2018-09-24	Cotter Jr.'s Reply to RDI's Opp'n to	LII	JA12897-
	Motion to Retax Costs	LII	JA12921
2018-09-24	Cotter Jr.'s Appendix of Exhibits		IA 12022
	ISO Reply to RDI's Opposition to	LII, LIII	JA12922-
	Motion to Retax Costs		JA13112
2018-10-01	Cotter Jr.'s Opposition to RDI's	LIII	JA13113-
	Motion for Judgment in its Favor	LIII	JA13125
2018-10-02	Transcript of 10-01-18 Hearing on	LIII	JA13126-
	Cotter Jr.'s Motion to Retax Costs	LIII	JA13150
2018-11-02	Cotter Jr.'s Letter to Court	LIII	JA13151-
	Objecting to Proposed Order	LIII	JA13156
2018-11-02	Cotter Jr.'s Errata to Letter to		JA13157-
	Court Objecting to Proposed	LIII	JA13157- JA13162
	Order		JA13102
2018-11-06	Order Granting in Part Motion to		JA13163-
	Retax Costs & Entering Judgment	LIII	JA13165- JA13167
	for Costs ("Cost Judgment")		JA1310/
2018-11-06	Notice of Entry of Order of Cost	LIII	JA13168-
	Judgment		JA13174
2018-11-16	Order Denying RDI's Motion for	LIII	JA13175-
	Attorneys' Fees	L111	JA13178

Date	Description	Vol.#	Page Nos.
2018-11-06	Order Denying RDI's Motion for Judgment in Its Favor	LIII	JA13179- JA13182
2018-11-20	Notice of Entry of Order Denying RDI's Motion for Attorneys' Fees	LIII	JA13183- JA13190
2018-11-20	Notice of Entry of Order Denying RDI's Motion for Judgment in Its Favor	LIII	JA13191- JA13198
2018-11-26	Cotter Jr.'s Motion for Reconsideration & Amendment of Cost Judgment, for Limited Stay of Execution on OST	LIII	JA13199- JA13207
2018-11-30	RDI's Opposition to Cotter Jr.'s Motion for Reconsideration and Response to Motion for Limited Stay of Execution on OST	LIII	JA13208- JA13212
2018-11-30	Adams and Cotter sisters' Joinder to RDI's Opposition to Cotter Jr.'s Motion for Reconsideration and Response to Motion for Limited Stay of Execution	LIII	JA13213- JA13215
2018-12-06	Order Re Cotter Jr.'s Motion for Reconsideration & Amendment of Judgment for Costs and for Limited Stay	LIII	JA13216- JA13219
2018-12-06	Cotter Jr.'s Notice of Appeal from Cost Judgment	LIII	JA13220- JA13222
2018-12-07	Notice of Entry of Order Re Cotter Jr.'s Motion for Reconsideration & Amendment of Cost Judgment and for Limited Stay	LIII	JA13223- JA13229
2018-12-14	Cotter Jr.'s Notice of Posting Cost Bond on Appeal	LIII	JA13230- JA13232

Date	Description	Vol. #	Page Nos.
2018-06-18	Adams and Cotter sisters' Joinder to RDI's Combined Opposition to Cotter Jr.'s Motion to Compel & Motion for Relief	XXXII, XXXIII	JA7928- JA8295
2018-11-30	Adams and Cotter sisters' Joinder to RDI's Opposition to Cotter Jr.'s Motion for Reconsideration and Response to Motion for Limited Stay of Execution	LIII	JA13213- JA13215
2018-01-04	Adams and Cotter sisters' Motion for Judgment as a Matter of Law	XXV	JA6192- JA6224 (FILED UNDER SEAL JA6224A-F)
2018-06-01	Adams and Cotter sisters' Motion for Summary Judgment ("Ratification MSJ")	XXIX	JA7173- JA7221
2018-05-15	Adams and Cotter sisters' Motion to Compel Production of Docs re Expert Fee Payments on OST	XXVIII, XXIX	JA6938- JA7078
2018-05-18	Adams and Cotter sisters' Pre- Trial Memo	XXIX	JA7088- JA7135
2018-06-15	Adams and Cotter sisters' Reply ISO of Ratification MSJ	XXXII	JA7841- JA7874
2015-06-18	Amended AOS - Douglas McEachern	I	JA32-JA33
2015-06-18	Amended AOS - Edward Kane	I	JA34-JA35
2015-06-18	Amended AOS - Ellen Cotter	I	JA36-JA37
2015-06-18	Amended AOS - Guy Adams	I	JA38-JA39
2015-06-18	Amended AOS - Margaret Cotter	I	JA40-JA41
2015-06-18	Amended AOS - RDI	I	JA42-JA43
2015-06-18	Amended AOS – Timothy Storey	I	JA44-JA45
2015-06-18	Amended AOS – William Gould	I	JA46-JA47

Date	Description	Vol. #	Page Nos.
2018-04-24	Bannett's Declaration ISO Gould's Opposition to Motion to Compel	XXVII	JA6572- JA6581
2016-04-05	Codding and Wrotniak's Answer to T2 Plaintiffs' First Amended Complaint	II	JA439- JA462
2015-06-12	Complaint	I	JA1-JA31
2016-10-17	Cotter Jr.'s Appendix of Exhibits ISO Opposition to Gould's MSJ	XVIII	JA4458- JA4517
2016-10-17	Cotter Jr.'s Appendix of Exhibits ISO Opposition to Individual Defendants' Partial MSJ No. 1	XVII, XVIII	JA4143- JA4311 (FILED UNDER SEAL JA4151A-C)
2016-10-17	Cotter Jr.'s Appendix of Exhibits ISO Opposition to Individual Defendants' Partial MSJ No. 2	XVIII	JA4312- JA4457
2018-09-24	Cotter Jr.'s Appendix of Exhibits ISO Reply to RDI's Opposition to Motion to Retax Costs	LII, LIII	JA12922- JA13112
2018-11-02	Cotter Jr.'s Errata to Letter to Court Objecting to Proposed Order	LIII	JA13157- JA13162
2018-11-02	Cotter Jr.'s Letter to Court Objecting to Proposed Order	LIII	JA13151- JA13156
2018-04-23	Cotter Jr.'s Motion for Omnibus Relief on OST	XXVI, XXVII	JA6432- JA6561 (FILED UNDER SEAL
2016-09-23	Cotter Jr.'s Motion for Partial	XIV, XV	JA6350A; JA6513A-C) JA3337-
	Summary Judgment ("MPSJ")		JA3697

Date	Description	Vol. #	Page Nos.
2018-11-26	Cotter Jr.'s Motion for Reconsideration & Amendment of Cost Judgment, for Limited Stay of Execution on OST	LIII	JA13199- JA13207
2017-12-19	Cotter Jr.'s Motion for Reconsideration of Ruling on Partial MSJ Nos. 1, 2 & 3 and Gould's MSJ on OST ("Motion for Reconsideration")	XXIII, XXIV	JA5793- JA5909
2018-06-12	Cotter Jr.'s Motion for Relief Based on Noncompliance with Court's May 2, 2018 Rulings on OST ("Motion for Relief")	XXXI	JA7569- JA7607
2017-12-29	Cotter Jr.'s Motion for Rule 54(b) Certification and Stay on OST	XXV	JA6092- JA6106
2018-04-18	Cotter Jr.'s Motion to Compel (Gould)	XXV, XXVI	JA6298- JA6431
2018-06-08	Cotter Jr.'s Motion to Compel on OST	XXIX, XXX, XXXI	JA7222- JA7568
2018-09-05	Cotter Jr.'s Motion to Retax Costs	XXXVI	JA8915- JA9018
2017-12-28	Cotter Jr.'s Motion to Stay on OST	XXV	JA6072- JA6080
2018-02-01	Cotter Jr.'s Notice of Appeal	XXV	JA6295- JA6297
2018-09-13	Cotter Jr.'s Notice of Appeal	XXXVII	JA9108- JA9110
2018-12-06	Cotter Jr.'s Notice of Appeal from Cost Judgment	LIII	JA13220- JA13222
2018-12-14	Cotter Jr.'s Notice of Posting Cost Bond on Appeal	LIII	JA13230- JA13232
2018-01-05	Cotter Jr.'s Opposition to Defendants' Motion for Judgment as a Matter of Law	XXV	JA6229- JA6238

Date	Description	Vol. #	Page Nos.
2016-10-13	Cotter Jr.'s Opposition to Gould's MSJ	XVI	JA4015- JA4051
2018-05-18	Cotter Jr.'s Opposition to Motion to Compel Production of Docs re Expert Fee Payments	XXIX	JA7079- JA7087
2016-10-13	Cotter Jr.'s Opposition to Partial MSJ No. 1	XVI, XVII	JA4052- JA4083
2018-06-13	Cotter Jr.'s Opposition to Ratification MSJ	XXXI	JA7608- JA7797
2018-06-13	Cotter Jr.'s Opposition to RDI's Demand Futility Motion	XXXI, XXXII	JA7798- JA7840
2018-10-01	Cotter Jr.'s Opposition to RDI's Motion for Judgment in its Favor	LIII	JA13113- JA13125
2018-05-11	Cotter Jr.'s Opposition to RDI's Motion for Leave to File Motion	XXVIII	JA6816- JA6937
2018-01-05	Cotter Jr.'s Opposition to RDI's Motion to Dismiss for Failure to Show Demand Futility	XXV	JA6225- JA6228
2018-05-18	Cotter Jr.'s Pre-Trial Memo	XXIX	JA7136- JA7157
2018-06-18	Cotter Jr.'s Reply ISO Motion for Relief Re: 05-02-18 Rulings	XXXIII, XXXIV	JA8302- JA8342
2018-01-03	Cotter Jr.'s Reply ISO Motion for Rule 54(b) Certification and Stay	XXV	JA6171- JS6178
2018-04-27	Cotter Jr.'s Reply ISO Motion to Compel (Gould)	XXVII	JA6582- JA6599
2018-09-24	Cotter Jr.'s Reply to RDI's Opp'n to Motion to Retax Costs	LII	JA12897- JA12921
2016-09-02	Cotter Jr.'s Second Amended Verified Complaint	III	JA519- JA575
2017-12-01	Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 1 & 2 & Gould MSJ	XXI	JA5094- JA5107

Date	Description	Vol. #	Page Nos.
2017-12-01	Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 2 & 3 & Gould MSJ	XXI, XXII	JA5300- JA5320
2017-12-01	Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 2 & 5 & Gould MSJ	XXI	JA5119- JA5134
2017-12-01	Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 2 & 6 & Gould MSJ	XXI	JA5253- JA5264
2016-10-13	Cotter, Jr.'s Opposition to Partial MSJ No. 2	XVII	JA4084- JA4111
2016-10-13	Cotter, Jr.'s Opposition to Partial MSJ No. 6	XVII	JA4112- JA4142
2017-12-27	Declaration of Bannett ISO Gould's Opposition to Cotter Jr.'s Motion for Reconsideration	XXIV, XXV	JA5987- JA6064
2016-10-21	Declaration of Bannett ISO Gould's Reply ISO MSJ	XIX	JA4636- JA4677
2017-12-05	Declaration of Bannett ISO Gould's Supplemental Reply ISO MSJ	XXII, XXIII	JA5555- JA5685
2018-01-05	Declaration of Krum ISO Cotter Jr.'s Opposition to Motion for Judgment as a Matter of Law	XXV	JA6239- JA6244
2017-12-01	Declaration of Levin ISO Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 1 & 2 & Gould MSJ	XXI	JA5108- JA5118
2017-12-01	Declaration of Levin ISO Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 2 & 5 & Gould MSJ	XXI	JA5135- JA5252
2017-12-01	Declaration of Levin ISO Cotter Jr.'s Supplemental Opposition to Partial MSJ Nos. 2 & 6 & Gould MSJ	XXI	JA5265- JA5299

Date	Description	Vol.#	Page Nos.
2017-12-01	Declaration of Levin ISO Cotter Jr.'s Supplemental Opposition to So-Called MSJ Nos. 2 & 3 & Gould MSJ	XXII	JA5321- JA5509
2016-09-23	Defendant William Gould ("Gould")'s MSJ	III, IV, V, VI	JA576- JA1400
2018-08-14	Findings of Fact and Conclusions of Law and Judgment	XXXIV	JA8401- JA8411
2017-10-04	First Amended Order Setting Civil Jury Trial, Pre-Trial Conference, and Calendar Call	XX	JA4928- JA4931
2015-10-22	First Amended Verified Complaint	II	JA263- JA312
2018-04-24	Gould's Declaration ISO Opposition to Motion to Compel	XXVII	JA6569- JA6571
2017-10-17	Gould's Joinder to Motion for Evidentiary Hearing re Cotter Jr.'s Adequacy as Derivative Plaintiff	XX	JA4975- JA4977
2018-06-18	Gould's Joinder to RDI's Combined Opposition to Cotter Jr.'s Motion to Compel & Motion for Relief	XXXIII	JA8296- JA8301
2017-12-27	Gould's Opposition to Cotter Jr.'s Motion for Reconsideration	XXIV	JA5982- JA5986
2018-04-24	Gould's Opposition to Cotter Jr.'s Motion to Compel	XXVII	JA6562- JA6568
2016-10-21	Gould's Reply ISO MSJ	XIX	JA4610- JA4635
2017-12-01	Gould's Request For Hearing on Previously-Filed MSJ	XXI	JA5078- JA5093
2017-12-04	Gould's Supplemental Reply ISO of MSJ	XXII	JA5538- JA5554
2017-11-28	Individual Defendants' Answer to Cotter Jr.'s Second Amended Complaint	XX, XXI	JA5048- JA5077

Date	Description	Vol.#	Page Nos.
2016-03-14	Individual Defendants' Answer to Cotter's First Amended Complaint	II	JA375- JA396
2017-10-11	Individual Defendants' Motion for Evidentiary Hearing Re Cotter Jr.'s Adequacy as Derivative Plaintiff	XX	JA4932- JA4974
2016-09-23	Individual Defendants' Motion for Partial Summary Judgment (No. 1) Re: Plaintiff's Termination and Reinstatement Claims ("Partial MSJ No. 1)	VI, VII, VIII, IX	JA1486- JA2216 (FILED UNDER SEAL JA2136A-D)
2016-09-23	Individual Defendants' Motion for Partial Summary Judgment (No. 2) Re: The Issue of Director Independence ("Partial MSJ No. 2")	IX, X	JA2217- JA2489 (FILED UNDER SEAL JA2489A- HH)
2016-09-23	Individual Defendants' Motion for Partial Summary Judgment (No. 3) On Plaintiff's Claims Related to the Purported Unsolicited Offer ("Partial MSJ No. 3")	X, XI	JA2490- JA2583
2016-09-23	Individual Defendants' Motion for Partial Summary Judgment (No. 4) On Plaintiff's Claims Related to the Executive Committee ("Partial MSJ No. 4")	XI	JA2584- JA2689
2016-09-23	Individual Defendants' Motion for Partial Summary Judgment (No. 5) On Plaintiff's Claims Related to the Appointment of Ellen Cotter as CEO ("Partial MSJ No. 5")	XI, XII	JA2690- JA2860

Date	Description	Vol. #	Page Nos.
2016-09-23	Individual Defendants' Motion for Partial Summary Judgment (No. 6) Re Plaintiff's Claims Re Estate's Option Exercise, Appointment of Margaret Cotter, Compensation Packages of Ellen Cotter and Margaret Cotter, and related claims Additional Compensation to Margaret Cotter and Guy Adams ("Partial MSJ No. 6")	XII, XIII, XIV	JA2861- JA3336
2015-09-03	Individual Defendants' Motion to Dismiss Complaint	I	JA149- JA237
2016-10-26	Individual Defendants' Objections to Declaration of Cotter, Jr. Submitted in Opposition to Partial MSJs	XIX	JA4725- JA4735
2017-12-26	Individual Defendants' Opposition to Cotter Jr.'s Motion For Reconsideration	XXIV	JA5910- JA5981
2018-01-02	Individual Defendants' Opposition to Cotter Jr.'s Motion for Rule 54(b) Certification and Stay	XXV	JA6132- JA6139
2016-10-13	Individual Defendants' Opposition to Cotter Jr.'s MPSJ	XVI	JA3815- JA3920
2016-10-21	Individual Defendants' Reply ISO of Partial MSJ No. 1	XVIII	JA4518- JA4549
2016-10-21	Individual Defendants' Reply ISO Partial MSJ No. 2	XVIII, XIX	JA4550- JA4567
2016-10-21	Individual Defendants' Reply ISO Partial MSJ Nos. 3, 4, 5, and 6	XIX	JA4678- JA4724
2017-12-04	Individual Defendants' Reply ISO Renewed Partial MSJ Nos. 1 & 2	XXII	JA5510- JA5537
2017-11-09	Individual Defendants' Supplement to Partial MSJ Nos. 1, 2, 3, 5, and 6	XX	JA4981- JA5024

Date	Description	Vol. #	Page Nos.
2017-12-08	Joint Pre-Trial Memorandum	XXIII	JA5686- JA5717
2018-08-24	Memorandum of Costs submitted by RDI for itself & the director defendants	XXXIV	JA8426- JA8446
2016-09-23	MIL to Exclude Expert Testimony of Steele, Duarte-Silva, Spitz, Nagy, & Finnerty	VI	JA1401- JA1485
2015-08-10	Motion to Dismiss Complaint	I	JA48-JA104
2018-08-16	Notice of Entry of Findings of Fact and Conclusions of Law and Judgment	XXXIV	JA8412- JA8425
2018-11-20	Notice of Entry of Order Denying RDI's Motion for Attorneys' Fees	LIII	JA13183- JA13190
2018-11-20	Notice of Entry of Order Denying RDI's Motion for Judgment in Its Favor	LIII	JA13191- JA13198
2018-01-04	Notice of Entry of Order Granting Cotter Jr.'s Motion for Rule 54(b) Certification	XXV	JA6182- JA6188
2018-11-06	Notice of Entry of Order of Cost Judgment	LIII	JA13168- JA13174
2018-12-07	Notice of Entry of Order Re Cotter Jr.'s Motion for Reconsideration & Amendment of Cost Judgment and for Limited Stay	LIII	JA13223- JA13229
2017-12-29	Notice of Entry of Order Re Individual Defendants' Partial MSJs, Gould's MSJ, and MIL	XXV	JA6081- JA6091
2016-12-22	Notice of Entry of Order Re Partial MSJ Nos. 1-6 and MIL to Exclude Expert Testimony	XX	JA4921- JA4927
2018-09-05	Notice of Entry of SAO Re Process for Filing Motion for Attorney's Fees	XXXVI	JA8907- JA8914

Date	Description	Vol.#	Page Nos.
2018-01-04	Order Denying Cotter Jr.'s Motion for Reconsideration and Stay	XXV	JA6189- JA6191
2018-11-16	Order Denying RDI's Motion for Attorneys' Fees	LIII	JA13175- JA13178
2018-11-06	Order Denying RDI's Motion for Judgment in Its Favor	LIII	JA13179- JA13182
2015-10-12	Order Denying RDI's Motion to Compel Arbitration	II	JA257- JA259
2018-01-04	Order Granting Cotter Jr.'s Motion for Rule 54(b) Certification	XXV	JA6179- JA6181
2016-10-03	Order Granting Cotter Jr.'s Motion to Compel Production of Documents & Communications Re the Advice of Counsel Defense	XV	JA3698- JA3700
2018-07-12	Order Granting in Part Cotter Jr.'s Motion for Omnibus Relief & Motion to Compel	XXXIV	JA8398- JA8400
2018-07-12	Order Granting In Part Cotter Jr.'s Motion to Compel (Gould) & Motion for Relief	XXXIV	JA8395- JA8397
2018-11-06	Order Granting in Part Motion to Retax Costs & Entering Judgment for Costs ("Cost Judgment")	LIII	JA13163- JA13167
2018-12-06	Order Re Cotter Jr.'s Motion for Reconsideration & Amendment of Judgment for Costs and for Limited Stay	LIII	JA13216- JA13219
2016-10-03	Order Re Cotter Jr.'s Motion to Permit Certain Discovery re Recent "Offer"	XV	JA3701- JA3703
2016-12-21	Order Re Individual Defendants' Partial MSJ Nos. 1–6 and MIL to Exclude Expert Testimony	XX	JA4917- JA4920

Date	Description	Vol. #	Page Nos.
2017-12-28	Order Re Individual Defendants' Partial MSJs, Gould's MSJ, and MILs	XXV	JA6065- JA6071
2015-10-19	Order Re Motion to Dismiss Complaint	II	JA260- JA262
2016-12-20	RDI's Answer to Cotter Jr.'s Second Amended Complaint	XX	JA4891- JA4916
2016-03-29	RDI's Answer to Cotter, Jr.'s First Amended Complaint	II	JA397- JA418
2016-03-29	RDI's Answer to T2 Plaintiffs' First Amended Complaint	II	JA419- JA438
2018-08-24	RDI's Appendix of Exhibits to Memorandum of Costs	XXXIV, XXXV, XXXVI	JA8447- JA8906
2018-09-14	RDI's Appendix ISO Opposition to Motion to Retax ("Appendix") Part 1	XXXVII, XXXVIII , XXXIX	JA9220- JA9592
2018-09-14	RDI's Appendix, Part 2	XXXIX, XL, XLI	JA9593- JA10063
2018-09-14	RDI's Appendix, Part 3	XLI, XLII, XLIII	JA10064- JA10801
2018-09-14	RDI's Appendix, Part 4	XLIII, XLIV	JA10802- JA10898
2018-09-14	RDI's Appendix Part 5	XLIV, XLV	JA10899- JA11270
2018-09-14	RDI's Appendix, Part 6	XLV, XLVI	JA11271- JA11475
2018-09-14	RDI's Appendix, Part 7	XLVI, XLVII, XLVIII, XLIX, L	JA11476- JA12496
2018-09-14	RDI's Appendix, Part 8	L, LI, LII	JA12497- JA12893

Date	Description	Vol.#	Page Nos.
2018-06-18	RDI's Combined Opposition to Cotter Jr.'s Motion to Compel & Motion for Relief	XXXII	JA7875- JA7927
2019-10-21	RDI's Consolidated Reply ISO Individual Defendants' Partial MSJ Nos. 3, 4, 5 & 6	XIX	JA4589- JA4603
2018-01-03	RDI's Errata to Joinder to Individual Defendants' Opposition to Motion for Rule 54(b) Certification and Stay	XXV	JA6153- JA6161
2016-10-13	RDI's Joinder to Individual Defendants' Opposition to Cotter Jr.'s MPSJ	XVI	JA3921- JA4014
2018-01-03	RDI's Joinder to Individual Defendants' Opposition to Cotter Jr.'s Motion for Rule 54(b) Certification and Stay	XXV	JA6140- JA6152
2016-10-03	RDI's Joinder to Individual Defendants' Partial-MSJ No. 1	XV	JA3707- JA3717
2016-10-03	RDI's Joinder to Individual Defendants' Partial MSJ No. 2	XV	JA3718- JA3739
2016-10-03	RDI's Joinder to Individual Defendants' Partial MSJ No. 3	XV	JA3740- JA3746
2016-10-03	RDI's Joinder to Individual Defendants' Partial MSJ No. 4	XV	JA3747- JA3799
2016-10-03	RDI's Joinder to Individual Defendants' Partial MSJ No. 5	XV	JA3800- JA3805
2016-10-03	RDI's Joinder to Individual Defendants' Partial MSJ No. 6	XV, XVI	JA3806- JA3814
2017-11-21	RDI's Joinder to Individual Defendants' Supplement to Partial MSJ Nos. 1, 2, 3, 5 & 6	XX	JA5025- JA5027
2016-10-03	RDI's Joinder to MIL to Exclude Expert Testimony	XV	JA3704- JA3706

Date	Description	Vol. #	Page Nos.
2017-10-18	RDI's Joinder to Motion for Evidentiary Hearing re Cotter Jr.'s Adequacy as Derivative Plaintiff	XX	JA4978- JA4980
2018-09-07	RDI's Motion for Attorneys' Fees	XXXVI, XXXVII	JA9019- JA9101
2018-09-12	RDI's Motion for Judgment in Its Favor	XXXVII	JA9102- JA9107
2015-08-31	RDI's Motion to Compel Arbitration	I	JA127- JA148
2018-01-03	RDI's Motion to Dismiss for Failure to Show Demand Futility	XXV	JA6162- JA6170
2018-11-30	RDI's Opposition to Cotter Jr.'s Motion for Reconsideration and Response to Motion for Limited Stay of Execution on OST	LIII	JA13208- JA13212
2018-09-14	RDI's Opposition to Cotter Jr.'s Motion to Retax Costs	XXXVII	JA9111- JA9219
2018-04-27	RDI's Opposition to Cotter's Motion for Omnibus Relief	XXVII	JA6600- JA6698
2016-10-21	RDI's Reply ISO Gould's MSJ	XIX	JA4604- JA4609
2016-10-21	RDI's Reply ISO Individual Defendants' Partial MSJ No. 1	XIX	JA4568- JA4577
2016-10-21	RDI's Reply ISO Individual Defendants' Partial MSJ No. 2	XIX	JA4578- JA4588
2015-08-20	Reading International, Inc. ("RDI")'s Joinder to Margaret Cotter, Ellen Cotter, Douglas McEachern, Guy Adams, & Edward Kane ("Individual Defendants") Motion to Dismiss Complaint	I	JA105- JA108
2015-11-10	Scheduling Order and Order Setting Civil Jury Trial, Pre-Trial Conference and Calendar Call	II	JA313- JA316

Date	Description	Vol. #	Page Nos.
2018-05-04	Second Amended Order Setting Jury Trial, Pre-trial Conference, and Calendar Call	XXVII	JA6724- JA6726
2016-06-21	Stipulation and Order to Amend Deadlines in Scheduling Order	II	JA463- JA468
2018-09-14	Suggestion of Death of Gould Upon the Record	LII,	JA12894- JA12896
2016-02-12	T2 Plaintiffs' First Amended Complaint	II	JA317- JA355
2015-08-28	T2 Plaintiffs' Verified Shareholder Derivative Complaint	I	JA109- JA126
2015-10-06	Transcript of 9-10-15 Hearing on Defendants' Motion to Dismiss & Plaintiff Cotter Jr. ("Cotter Jr.")'s Motion for Preliminary Injunction	I, II	JA238- JA256
2016-02-23	Transcript of 2-18-16 Hearing on Motion to Compel & Motion to File Document Under Seal	II	JA356- JA374
2016-06-23	Transcript of 6-21-16 Hearing on Defendants' Motion to Compel & Motion to Disqualify T2 Plaintiffs	II	JA469- JA493
2016-08-11	Transcript of 8-9-16 Hearing on Cotter Jr.'s Motion for Partial Summary Judgment, Motion to Compel & Motion to Amend	II, III	JA494- JA518
2016-11-01	Transcript of 10-27-16 Hearing on Motions	XIX, XX	JA4736- JA4890
2017-11-27	Transcript of 11-20-17 Hearing on Motion for Evidentiary Hearing re Cotter Jr.'s Adequacy & Motion to Seal	XX	JA5028- JA5047
2017-12-11	Transcript of 12-11-2017 Hearing on [Partial] MSJs, MILs, and Pre-Trial Conference	XXIII	JA5718- JA5792

Date	Description	Vol. #	Page Nos.
2017-12-29	Transcript of 12-28-17 Hearing on Motion for Reconsideration and Motion for Stay	XXV	JA6107- JA6131
2018-01-05	Transcript of 1-4-18 Hearing on Cotter Jr.'s Motion for Rule 54(b) Certification	XXV	JA6245- JA6263
2018-01-08	Transcript of Hearing on Demand Futility Motion and Motion for Judgment	XXV	JA6264- JA6280
2018-01-10	Transcript of Proceedings of 01-8- 18 Jury Trial–Day 1	XXV	JA6281- JA6294
2018-05-03	Transcript of 4-30-18 Hearing on Motions to Compel & Seal	XXVII	JA6699- JA6723
2018-05-07	Transcript of 5-2-18 Hearing on Evidentiary Hearing	XXVII, XXVIII	JA6727- JA6815
2018-05-24	Transcript of 05-21-18 Hearing on Adams and Cotter sisters' Motion to Compel	XXIX	JA7158- JA7172
2018-06-20	Transcript of 06-19-18 Omnibus Hearing on discovery motions and Ratification MSJ	XXXIV	JA8343- JA8394
2018-10-02	Transcript of 10-01-18 Hearing on Cotter Jr.'s Motion to Retax Costs	LIII	JA13126- JA13150

### **CERTIFICATE OF SERVICE**

I certify that I am an employee of MORRIS LAW GROUP; I am familiar with the firm's practice of collection and processing documents for mailing; that, in accordance therewith, I caused the following document to be e-served via the Supreme Court's electronic service process. I hereby certify that on the 28th day of August, 2019, a true and correct copy of the foregoing JOINT APPENDIX TO OPENING BRIEFS FOR CASE NOS.

77648 & 76981, was served by the following method(s):

☑ Supreme Court's EFlex Electronic Filing System:

Stan Johnson Cohen-Johnson, LLC 255 East Warm Springs Road, Ste. 110 Las Vegas, Nevada 89119

Christopher Tayback Marshall Searcy Quinn Emanuel Urquhart & Sullivan LLP 865 South Figueroa Street, 10th Floor Los Angeles, CA

Attorneys for Respondents Edward Kane, Douglas McEachern, Judy Codding, and Michael Wrotniak

Mark Ferrario Kara Hendricks Tami Cowden Greenberg Traurig, LLP 10845 Griffith Peak Drive Suite 600 Las Vegas, Nevada 89135

Attorneys for Nominal Defendant Reading International, Inc.

Donald A. Lattin Carolyn K. Renner Maupin, Cox & LeGoy 4785 Caughlin Parkway Reno, Nevada 89519

Ekwan E. Rhow Shoshana E. Bannett Bird, Marella, Boxer, Wolpert, Nessim, Drooks, Lincenberg & Rhow, P.C. 1875 Century Park East, 23rd Fl. Los Angeles, CA 90067-2561

Attorneys for Respondent William Gould

Judge Elizabeth Gonzalez Eighth Judicial District court of Clark County, Nevada Regional Justice Center 200 Lewis Avenue Las Vegas, Nevada 89101

By: <u>/s/ Gabriela Mercado</u>

At any time after an Event of Default and without prejudice to any other remedies, Westpac NZ may immediately by notice to the Borrower terminate the Loan with the consequences set out in clause 2.2.

### 11. change in circumstances

- 11.1 Increased Costs
  | If as a result of:
  | westpace NZ complying with any law; or
  | any change in or introduction of any law or change in the interpretation or application or application of any law by any Governmental Agency or court which:
  | any change in or introduction of any law or change in the interpretation or application or a

### any of the following occur:

- c) the cost to Westpac NZ of making, funding or maintaining the Loan is increased; or
  d) the moneys payable to Westpac NZ or the effective return to Westpac NZ under or in connection with this Agreement is reduced; or
  Westpac NZ makes any syment or foregoes any interest or other return or, or calculated by reference to, any sum received or receivable by it under any Bank Document; or
  Westpac NZ is unable to obtain the rate of return on capital (including any notional return on capital calculated on a risk adjusted basis) which it would have received at the date of this Agreement;

g) Westpac NZ will use its best efforts promptly to notify the Borrower in writing of those events, provided that failure to do so will not affect Westpac NZ's rights under this clause; and the Borrower must pay on demand, from time to time, for the account of Westpac NZ, the amount certified by an Officer which will compensate Westpac NZ for its increased cost, reduction, payment or foregone interest or other return.

11.2 Survival of Obligations
The obligations of the Borrower under clause 11.1 will survive termination of the Loan and payment or repayment of all Outstanding Moneys.

11.3 Re-negotiation of Loan
If an additional amount is required to be paid under clause 11.1 then, without limiting the Borrower's obligations under this Agreement, Westpac NZ will, at the request of the Borrower given within 30 days of notice from Westpac NZ, consult with the Borrower with a view to mengodiating the terms of the Loan in order to mitigate or avoid any additional amounts payable under clause 11.1 provided that nothing in this clause will require Westpac NZ to act to its detriment.

11.4 Illegality
If as a result of:
a) the introduction or amendment of any law; or
b) any other circumstance affecting any interbank market or any relevant foreign exchange market generally;

It is unlawfut, impracticable or impossible for Westpac NZ to make, fund or allow to remain outstanding the Loan or to comply with its obligations or exercise its rights under this Agreement, then Westpac NZ will as soon as practicable give written notice to the Borrower and Westpac NZ will be entitled, to terminate the Loan, with effect from the date Westpac NZ specifies and with the consequences set out in clause 2.2.

### 12. indemnity

12.1 General Indemnity
The Borrower agrees to indemnify Westpac NZ, every Officer and every employee of Westpac NZ upon demand from against any loss, cost, expense, charge, damage, claim or liability which Westpac NZ, an Officer or an employee of Westpac NZ may suffer or incur as a direct or indirect consequence of:

an Event of Default or Potential Event of Default; an Event of Default or Potential Event of Default; the exercise, contemplated exercise or attempted exercise of any Power or the failure to exercise any Power; the receipt of any amount to be paid under any Bank Document a)

d) the Loan requested under clause 3 not being made for any reason (excluding default by Westpac NZ) on the date notified by the Borrower to Westpac NZ as the drawd

- 12.2 Examples

  Without institution, the indemnity in clause 12.1 will extend to:
  any losses, costs, penalties and expenses incurred by reason of the liquidation or re-engloyment of deposits or other funds acquired or contracted for by Westpac NZ (including loss of margin); and
  b) any losses, costs, penalties and expenses which may be incurred by Westpac NZ in:
  1) terminating any options or forward rate agreements or interest or currency swap contracts entered into in connection with the Loan; or
  ii) in entering into any new contracts which it deems appropriate to protect the return it would otherwise have expected under this Agreement.

### 12.3 Currency Indemnity

3 Currency meemany

If or any response;

Westpac NZ receives or annount under a Bank Document in a currency other than the currency in which it should have been paid and, after Westpac NZ has converted that other currency to the correct currency there is not enough to pay off the full amount then due under the Bank Document; or

Westpac NZ obtains any judgment or court order against the Borrower in a currency other than the currency in which the Outstanding Moneys are due, and Westpac NZ incurs any loss from the conversion of any amount actually received by it from that other currency to the correct currency.

Then, as a separate and independent indemnity obligation, the Borrower must pay Westpac NZ the full amount of any shortfall or of any such loss incurred by Westpac NZ. b)

12.4 Unconditional indemnities
The indemnities in this clause 12 are unconditional and irrevocable and will survive termination of the Loan and payment of all Outstanding Moneys and the release of any Security and will not be discharged or impaired by any act, omission, matter or thing which might discharge them but for this provision.

### 13. Set off and combination

13.1 Set Off
If the Borrower has any money in any account with Westpac NZ, then Westpac NZ can use it to pay amounts the Borrower owes under this Agreement but need not do so. If the Borrower is in default, Westpac NZ can use money which has not yet matured due and convert money in the account of the Borrower in foreign currencies. To the maximum extent allowed by law, the Borrower gives up any right to set off any amounts Westpac NZ owes it against the Outstanding Moneys.

omangent amounts
any lime an amount is contingently due from the Borrower or an amount due is not quantified, Westpac NZ may retain and withhold repayment of any money in any account of the Borrower and the payment of interest or other moneys pending that amount coming due and/or being quantified and may set off the maximum liability which may at any time be or become owing to Westpac NZ by the Borrower and in each case without prior notice or demand.

13.3 Combination
Subject to any applicable Bank Document, where the Borrower has two or more accounts with Westpac NZ:
Westpac NZ may at any time combine any two or more of those accounts. It may do so without notice and whether or not it has allowed a set-off for a calculation of interest between any of those accounts;
Westpac NZ may at any time combine any two or more of those accounts even where one or more of the combined accounts are in different currencies and may effect currency exchanges appropriate to implement that combination; and if Westpac NZ combines two or more accounts, it may decline to pay cheques and it may otherwise act as if the combined accounts had always been one account.

13.4 Contractual Rights
Westpac NZ's rights under this clause are contractual rights affecting the terms upon which a credit balance is held and the creation of those rights does not constitute the creation of a security interest in respect of that credit balance.

13.5 Deposits with Westpac NZ
Any moneys which, pursuant to a Bank Document, are deposited at any time by the Borrower with Westpac NZ (or withheld by Westpac NZ from a payment to the Borrower and retained on deposit with it) must, unless otherwise provided, be held on the following basis:

a) each deposit and all rights of the Borrower relating to it must be incapable of assignment by the Borrower or of being the subject of a security interest except:

ii) In favour of Westpac NZ:
iii) with the prior written consent of Westpac NZ: and

b) the Borrower must have no right to withdraw any moneys from a deposit until all obligations of the Borrower under the Bank Documents (present and future, direct and contingent) have been performed and compiled with, except:

**JA3060** 

- as expressly permitted by the terms of any Bank Documents under which that deposit was made;
   for the purpose of complying with its obligations under the Bank Documents; or
   with the prior written consent of Westpac NZ.

13.6 Interest

Each amount deposited with Westpac NZ under clause 13.5 will (unless otherwise agreed) bear interest calculated by reference to successive deposit periods of a duration agreed by Westpac NZ and the Borrower (or, in the absence of agreement, as Westpac NZ and the Borrower (or, in the absence of agreement, as Westpac NZ certifies as applicable to deposits of similar size and maturity placed with it by customers). As long as no Event of Default or Potential Event of Default has occurred (in which case interest is to be added to the deposit) and subject to clause 13.1, that interest is to be paid to the Borrower as it may direct.

To the extent permitted by law, neither Westpac NZ nor any Officer or Attorney will be liable in respect of any conduct, (including but not limited to the agreement to provide the Loan), omission, delay, negligence or breach of duty in the exercise or failure to exercise a Power or for any loss (including consequential loss) which results. However, Westpac NZ and any Officer or Attorney will be liable where liability arises from its own fraud or willul misconduct.

The Borrower confirms that:
It has not entered in any Bank Document in reliance on, or as a result of any conduct of any kind of or on behalf of Westpac NZ or any Related Company of Westpac NZ (including any advice, warranty, representation or undertaking); and neither Westpac NZ nor any Related Company of Westpac NZ is obliged to do anything (including disclose anything or give advice), except as expressly set out in the Bank Documents or in writing signed by or on behalf of Westpac NZ or any Related Company of Westpac NZ.

16.1 Westpac NZ may assign
Westpac NZ can transfer this Agreement and all or part of the Outstanding Moneys to someone else. If it does, this Agreement and any transferred Bank Document will apply to the transferee as if it was Westpac NZ. To the maximum extent allowed by law, any transfer will be free of any set off, equity or cross claim which the Borrower would have had against Westpac NZ or the transferee of any Bank Document but for this clause.

16.2 Borrower may not assign
The Borrower may not assign or otherwise transfer or grant any Security Interest over any of its rights and obligations under this Agreement or any other Bank document.

16.3 disclosure of information
Westpac NZ may disclose to a Related Company, potential assignee or transferee of its rights or obligations under any Bank Document or any other relevant party any information about the Borrower as Westpac NZ considers appropriate. However, before disclosure of any other relevant party any information about the Borrower as Westpac NZ considers appropriate. However, before disclosure of any other relevant party any information about the Borrower as Westpac NZ considers appropriate. However, before disclosure of the information about the Borrower of its infendior; and advise the Borrower of its intention; and

a) require an undertaking protecting the confidentiality of that information from any potential assignee or transferee or other relevant party.

### 17. counterpart execution

This Agreement may be executed in two or more counterparts, all of which will be deemed to constitute the same instrument. Westpac NZ may accept as an original a facsimile copy of this Agreement executed by the Borrower and a facsimile copy, when taken with a counterpart executed by Westpac NZ, will be deemed to be one original copy of this Agreement.

### 18. anti-money laundering

The Borrower agrees to provide all information to Westpac NZ which Westpac NZ requires in order to manage its anti-money-taundering and countering terrorism financing obligations, to manage its economic and trade sanctions risks or to comply with any laws, niles or regulations in New Zealand or any other country. The Borrower agrees that Westpac NZ may refuse to establish a business relationship with the Borrower, may be required to delay, defer, stop or refuse to process any transaction, or may consist or refuse to process any transaction without notice, if the Borrower files to provide this information to Westpac NZ in the manner and timeframe specified by Westpac NZ. The Borrower agrees that Westpac NZ may delay, defer, stop or refuse to process any transaction without incurring any liability if Westpac NZ knows or suspects that:

a) the transaction will not not require the process any transaction will or may treat any laws or regulations in New Zealand or any other country; or indirectly, to any person (natural, corporate or governmental) that is sanctioned, under economic and trade sanctions imposed by any country.

Unless the Borrower has disclosed to Westpac NZ that the Borrower is acting in a trustee capacity or on behalf of another party, the Borrower warrants that the Borrower is acting solely on the Borrower's own behalf in entering into this Agreement

For each transaction conducted under this Agreement, the Borrower represents and warrants to Westpac NZ that, to best of the Borrower's knowledge, information and belief at the time the transaction takes place, the processing of that transaction by Westpac NZ in accordance with the Borrower's instructions will not breach any taxes or regulations in New Zealand or any other country relevant to the transaction.

19.1 Definitions The following definitions apply, unless the context requires otherwise:

Agreement means this agreement and any variations or additions to it agreed in writing between the parties;

Attorney means a person appointed as attorney under any Bank Document;

Availability Period means the period beginning on the date that Westpac NZ confirms to the Borrower that the condition precedent in this Agreement has been satisfied and expiring on 30 June 2015;

Bank Document means a document or agreement:
to which Westpac NZ: and any one or more of the Borrower and/or any Guarantor are or become parties or purport to be or become parties; or
under which obligations arise or are intended to arise from any one or more of the Borrower and/or any Guarantor to, or for the benefit of, Westpac NZ;
(in each case whether or not the parties are involved or it arises as a result of an assignment or transfer). It includes this Agreement and any Security;

Banking Day means any day (other than a Saturday or Sunday) on which registered banks are open for business of the nature required by this Agreement in Auckland and, if on that day a transfer of funds is to be made under this Agreement, the city to which the funds are to be transferred;

Commencement Date means the date the Loan is first drawn down;

<u>Default Rate</u> means the aggregate of Westpac NZ's Indicator Lending Rate from time to time and 7% per annum;

Development means the development to be carried out on the land at 200 Wakefield Street. Wellington:

Dollar and S\_means the lawful currency of New Zealand;

Event of Default means any of the events described as such in the General Security Agreement, notwithstanding that the General Security Agreement may be wholly or partially satisfied, released, discharged, avoided, replaced, lost or destroyed from time to

First Tranche means \$35,000,000;

Floation Rate, means, for each Floating Rate Period, the aggregate of the Margin and Westpac NZ's 90 day bank bill bid rate (rounded upwards to the nearest first decimal place) for bills of a comparable amount to the amount of the Loan at the relevant time on the first day of that Floating Rate Period.

Floating Rate Period means a period of 3 calendar months provided that:

the first Floating Rate Period will begin on the Commencement Date;

: 11 :

- b) each subsequent Floating Rate Period will begin at the end of the previous one;
   any Floating Rate Period which would otherwise end on a day which is not a Banking Day will be extended to the next Banking Day;
   if any Floating Rate Period would otherwise extend beyond the Termination Date it will end on the Termination Date;

General Security Agreement, means the general security agreement entered into by the Borrower and Westpac NZ and dated on or about the date of this Agreement;

Governmental Agency means any government or any governmental, seni-governmental or judicial entity or authority including any local government, statutory or self-regulatory organisation established, approved or authorised under law, and any stock exchange, in any case having jurisdiction in relation to the affairs of any party to a Bank Document or to whose control or jurisdiction any party to a Bank Document has consented;

Group means Reading New Zealand Limited and its subsidiaries;

Guarantee means any guarantee, indemnity, letter of credit, logally binding letter of confort or suretyship. It includes any obligation or inevocable offer to be responsible for a debt (as defined below) or for the insolvency or financial condition of another person. It also includes any other obligation or inevocable offer to pay a debt or to purchase a dest, to provide funds for the payment or discharge of a debt (whether by the advance of money, the purchase of or subscription for shares, stock or other interests issued by another person, the purchase of assets or services, or discharge, or to indensity against the consequences of dealth in the payment of a debt. For the purposes of this definition debt includes any obligation or indettedness of another person, present or future, actual, prospective or contingent and any capital or premium on shares, stock or other interests issued by another person;

<u>Guarantor</u> means any person who creates or enters into a Guarantee in support of any Outstanding Moneys;

Interest Calculation Date means the date falling on the day in each calendar month which numerically corresponds to the Commencement Date provided that where there is no numerically corresponding day in any calendar month then it will be the first day in the next calendar month.

Interest Payment Date means, unless Westpac NZ shall determine otherwise, the date falling on the day after each interest Calculation Date provided that: where the interest Calculation Date would otherwise fail on a non-Banking Day that the Interest Payment Date shall be the Banking Day after the next Banking Day; and where the Interest Calculation Date would otherwise fail on a Banking Day which immediately precedes a non-Banking Day then the Interest Payment Date shall be the next Banking Day;

Land means any land mortgaged or charged from time to time by any member of the Charging Group to Westpac NZ, together with all buildings and improvements on such land;

Letter of Offer means the letter of offer dated 13 April 2015 from Westpac NZ to the Borrower in relation to the Loan;

Loan\_means the sum of \$50,000,000 (being the aggregate of the First Tranche and the Second Tranche);

Margin means, subject to clause 6.2, 1,75% per annum:

Officer includes each employee of Westpac NZ whose title includes the word Manager or occupying an office whose title includes the word Manager and any person (who need not be an employee) authorised by Westpac NZ;

Quistanding Moneys means, at any time, the Loan outstanding and all other moneys payable, including contingently payable, by the Borrower under this Agreement including accrued interest (including default interest), fees, costs and other expenses whether or not those sums are then due and owing;

Potential Event of Default means any event which, with the giving of notice, lapse of time or satisfaction of any condition or happening of any event would constitute an Event of Default;

Bower means a power, right, authority, discretion or remedy which is conferred on Westpac NZ, an Officer or an Attorney by a Bank Document or by law in relation to a Bank Document;

Quantity Surveyor means any quantity surveyor approved in writing by Westpac NZ;

Related Company has the meaning given to that term in the Companies Act 1993 but on the basis that Subsidiary has the meaning given to it in this Agreement and that body corporate includes any entity;

Security means the General Security Agreement and any other security or undertaking provided by the Borrower or any of the Guarantors or procured by the Borrower to be provided to, or for the benefit of, Westpac NZ from time to time in respect of the Borrower's obligations under this Agreement;

Subsidiary has the meaning given to it in the Financial Reporting Act 1993;

Tax includes any tax, levy, impost, deduction, charge, rate, duty or withholding which is levied or imposed by a Governmental Agency and is required by law to be paid and any related interest, penalty, charge, fee or other amount;

Termination Date means 31 March 2018.

- k) I)
- Termination Date means 31 March 2018.

  19.2 General
  Headings are for convenience only. They do not affect interpretation. The following rules apply unless the context requires otherwise:
  Singular includes the plant and the converse;
  b) A general reduction as general gen
- o)

**JA3064** 

## Loan Agreement

THIS LOAN AGREEMENT (the "Agreement") is unide as of the 26th day of June, 2014, between Santauder Bank, N.A., with a place of business at 195. Montague. Street, Brooklyn, New York 11201 (bereinsfer referred to as "Lender") and Sutron Bill Properties, LLC, a Nevada limited liability company, with a principal place of business at 6160. Center Drive, Suite 903, Los Angeles, California 90045, Attention Andrzej Matyczytski (heroinafter referred to as "Borrower").

## WITNESSETH:

WHEREAS, Lender is the holder of that certain Amended and Restated Mortgage, Assignment of Lenzes and Rents, Security Agreement and Fishure Filing dated as of Jone 28, 2012, made by the Borrower, in favor of Jender, as Lender, in the principal amount of \$15,000,000.050, and recorded on July 20, 2012 in the Register's Office of the City and State of New York as CRFN 201260228512 (regetter with all extensions, renewals, modifications, substitutions and amendments thereof the "First Mortgage" which secures that certain Security Agreement and Restated Promissory Note in the original prancipal amount of \$15,000,000.00 made by Borrower to the order of Lender (togetter with all extensions, renewals, modifications, substitutions and amendments thereof, the "First Note"; the First Note, the First Mortgage and any of the Isan documents which secure and evidence the indebtedness pursuant to the First Note shall hereinafter be referred to as the "First Mortgage Loan Documents");

WHEREAS, the Borrower desires to borrow and the Lender desires to lend to the Borrower, for Borrower's var., so more particularly set forth berein (the "Lose") in connection with its ownership of the real property located at 1901-1907 Third Avenue, New York, New York, 19022 (the "Property"), the principal authors of Six Million and 061102 Dollars (US \$6,000,080,000, subject and persuant to the terms of this Agreement and of that certain Promissory Note in like amount of even date herewith executed and delivered by the Borrower (the "Note");

WHEREAS, the Note will be secured by, surong other things., a Mortgage, Assignment of Leeses and Rents, Security Agreement and Fixture Filing by and between Borrower and Lender (together with all extensions, renewels, modifications, substitutions and amendaments thereof, the "Security Instrument") which upon recording will encounter the Property.

Capitalized terms used but not otherwise defined herein shall have the meanings ascribed thereto in the Note.

NOW, THEREFORE, the parties hereto agree as follows: Article L. THE ADVANCE

Section 1.01 Advances.

The Lender agrees, upon the terms, and subject to the conditions beroof, to make a conclusion showed the aggregate principal and conditions beroof, to make a conclusion showed on fully 1, 2016 (the "Maturity Date"), interest on the Advance shall accrue and to payable in accordance with the terms of the Note.

## Section 1.02 Manner of Borrowing.

When the Borrower desires to obtain the Advance, it shall give the Lender at least thirty

When the Hormover desired to obtain the Actioner, it sholl give the Lender at least thirty (30) days prior written notice (a "Notice to Borrow Flands shall state that it is irreveasible and specify the proposed date of borrowing and the amount thereof and shall constitute the Borrower's affirmation that all representations and warmanics contained herein are true and correct and that the Advance shall be subject to all of the Conditions (as bereined and set forthin this Agreement). The Advance shall be in a whole to marbler and in the uninfirmal amount of \$200,000,00. The Notice to Borrow Flands to be given personant to this Section shall be accepted only from the personant and the personant to be presented and the state of the state of the Section shall be accepted only from these personant obtained in the state of the Section shall be accepted only from the personant software state of all parties authorized to execute Notices to Borrow Flands.

The failure of the Lender to make the Advance on the date set forth in the Notice to Borrower fund; shall not subject the Lender to any damages whatsoever provided the Advance is made within a reasonable time after the later to occur of the date set forth in the Notice to Borrower Funds or the date all Conditions to the Advance are satisfied; it being expressly acknowledged by the Borrower that such condition is deemed a material inducement to the Lender for entering into the arrangement manifested by this Losa Agreement and the Note.

## Arricle B. REPRESENTATIONS AND COVENANTS

## Section 2.01 Representations.

The Borrower represents and warrants to the Lander that on the date hereof:

(a) The Borrower has the power and sortbority to execute, deliver and perform this Agreement, and each of the other documents executed in connection therewith 1 to own its properties and to entry on its business as now conducted.

(b) The execution, delivery and performance of the Nate and this Agreement (i) have been day anthorized by all requisite action of the Borrower (ii) to Borrower (iii) to Borrower (iiii) do not violate the Borrower's Certificate of Formation or its Operating Agreement; any order of any court or other agency, or any agreement to which the Borrower is a party or by which the Borrower is bound; and (iv) will not be in conflict with, result in a breach of or constitute (with the notice or lapse of time or both) a definit under any such agreement;

- (e) There are no actions, suits or proceedings before or by any federal state, municipal or other governmental department, commission board, barena, agency or usurumentality pending against the Borrower which if determined adversely to the Borrower, would have a material adverse effect on the business, properties, operations or conditions.
- (d) No Event of Default into occurred under this Agreement and to default has occurred under any of the other Loan Documents beyond the expiration of any applicable notice, grace or cure period;
- (e) The Borrower makes no claim that the terms of the Note, including without limitation the interest rate thereon, is usurious nor that there exists any offset, deduction or defense with respect to the Borrower's obligations under the Lora Documents;
- (f) The Property is free and clear of any tiens, charges or encumbrances other than the First Mortgage:
- (g) To Borrower's knowledge, there are no outstanding judgments against the Borrower which have not been paid;
- (i) The recet recent Financial Statements heretofore delivered to the Lender are complete and correct and since the date thereof there has not occurred any material adverse change in the financial condition or operations of the Borrower, Guarantee or the Property from that shown on said Financial Statements;
- (i) Borrower has no knowledge of any impediments to the full and complete performance by the Borrower hereunder or under any of the Loan Documents;
- G) Neither the Borrower nor any person or entity acting or purporting to sot on the Borrower's behalf has dealt with any broker in connection with the making of this ioan, except as set forth in the, commitment letter of Santander Baeia, N.A. addressed to Borrower and dated as of June 11, 2014, as ansended by letter agreement, if any. Borrower hereby indemnifies Lander, and agrees to hold Lender harmless, from and against all loss, liability, damage and expense, including reasonable attorneys' fees and expenses, suffered or incurred by Lender by reason of a breach of this representation. This provision will survive the closing and the repayment of the Note;
- (i) The Borrower is a limited liability company duly organized and validly existing under the laws of the State of Nevada and duly authorized to do business in the State of New York pursuant to the laws of the State of New York; and
- (1) That the funds received by the Borrower from the Advance requested hereunder and pursuant to the Note will be expended exclusively in connection with the acquisition of floor area rights and/or air rights related to induce development of the Property as some will be more particularly described in the Security Instrument.

Section 2.02 Cartain Covenants.

The Borrower covenants and agrees that so long as this Agreement shall remain in effect, Borrower shall;

- Pay all sums one and owing order the Note pursuant to its terms;
- (b) Do or cause to be done all things necessary to preserve and keep in full force and effect its existence under the laws of its state of formation;
- (c) Give protopt notice to the Leader of (i) any proceedings of which the Borrower has notice instituted by or against the Borrower, and (ii) any other action, event or condition of any nature which in either case the consagement of the Borrower reasonably believes could have, lead to or result in a touterful adverse effect upon the business, assets or function of the Borrower.
- (d) Refrain from mortgaging problems, granting or permitting may security interest, lien or encumbrance of may nature in may amount to exist with respect to may of the Borrower's property including without limitation the Property, except where such security interest, lien or encumbrance in for the benefit of the Lender or has otherwise been approved by the Lender;
- (a) Provide to Lender off of the deliverables to and when required presumn to Section 2.04 herein (collectively, the "Financial Statements");
- (f) Perform all of the Borrower's obligations ender the First Moragege encembering the Property including without limitation, payment of all sums due thereunder, in a timely reasoner. Upon the maturity (by acceleration or otherwise, or upon prepayment thereof) of the First Moragege held by Lender (or its assignes) covering the Property, or upon prepayment thereof, all amounts due hereunder shall simultaneously become due and payable;
- (g) Not insur any additional indebedness except, in the ordinary course of business, with customery time payment arrangements with vendors and suppliess; and
- (b) Pay all sums that may be necessary to be paid in order to enforce the Note and to embree and/or to record the Security Instrument and any agreement or any other documentation executed and delivered in connection with this Agreement, whether such sums be in the nature of recording fees, nontaging the or any other expense in connection with such recording.

## Section 2.03 Negative Piedge Covenants.

The Borrower pledges, covenants and agrees that so long as this Agreement shall remain in effect it shall not, without the prior written consent of the Lender, do any of the following:

(a) Except as otherwise permitted in the Security Instrument and the First Mongage, sell, transfer or otherwise convey, either voluntarily or involuntarily, all or any portion of the Property or any uncerest or extate therein;

- (b) Grant or suffer to exist any mortgage, pledge, lien, secured interest, hypotheoation or other encumbrances upon the Property including any personal properly owned by the Borrower now or hereafter placed in or attached to and necessarily used in connection with the Property except as maybe approved by Lender; or
- (c) Except for the Master Lease, enter into any leasing arrangement of any kind in respect of all or substantially all of the Property;
- (d) Reduce the rent payable by the tenant pursuant to the Master Lenze or enter into any amendments or modifications of the Master Lease;
- (e) Suffer or permit any mechanics' or other statutory tien which is filed against the Property to remain undischarged or not bonded for a period exceeding sorty (60) days beyond the filing date thereof; or
- (f) Grain or suffer to exist any independence (secored or insecured), other than indebtedness owing to Lender, or grain or suffer to exist any Lien on or with respect to any deposit accounts (other than any Lien in favor of the Lender), whether new owned or hereafter consisted by the Borrower, or pledge; assign or transfer any tights to any deposit accounts, except as may be approved in writing by Lender. "Lien" means any mortgage, pledge, hypothecasion, assignment, deposit accounts consistency or other), or preference, pricedy of other security agreement or preference attemperature of any kind or nature whateverer

#### Section 2.04 Books and Records, 1

- (a) Borrower shall keep accurate books and records of account in accordance with generally accepted accounting principles in which full, true and correct entries shall be promptly made with respect to Borrower; the Property and the operation thereof, and will permit all such books and records (including without immittion all contracts, statements, mences, bills and claims for labor, materials and services supplied for the construction, repair or operation to Borrower of the Improvements) to be inspected or nadited and copies made by Leader and its representatives during normal business hours and at any other reasonable times on at least for eight; 48) hours advance oncine. Borrower represents their its oblief executive office is as to find in the introductory principles of the Agreement and that all books and records pertaining to the Property or at its chief executive office. Borrower will familish, or cause to be franished, to Lender on or before ninety (90) calendar days following the end of each estendar year the following items, each certified by Borrower as being true and correct, in such forms and in such densit or Lender or or statewise true.

  (i) a written statement (rent roll) dated as of the last day of each such extendar year dentifying each of the Leasen by the term, space occupied, restal required to be paid (archiding percentage rents and tenant sates), security deposit paid, any rental concessions, all rent escalations, any rents paid none (1) month in advance, any apecial provisions or indecencents granted to tenants, any

2.04 stree as Section 3.8 of the Mortgage

taxes, maintenance and other common charges paid by tenants, all vacancies and identifying any defaults or payment delinquencies thereunder;

- (ii) year-to-date operating statements prepared for each calendar quarter during each such reporting period detailing the total revenues received, total expenses incurred, total cost of all capital improvements, total debt service and total cash flow; and
- (iii) a current personal financial statement of each Guarantor, if applicable, in a form satisfactory to Lender.
- (b) Within ninety (90) calendar days following the end of each calendar year, Borrower shall furnish a statement of the financial affairs and condition of the Borrower and the Property including a statement of profit and loss for the Property in such format and in such detail as Lender or its servicer may reasonably request, and setting forth the financial condition and the income and expenses for the Property for the immediately preceding calendar year prepared and audited by an independent certified public accountant. Borrower shall deliver to Lender copies of all income tax returns, requests for extension and other similar items contemporaneously with its delivery of same to the Internal Revenue Service.
- (c) Borrower will permit representatives appointed by Lender, including independent accountants, agents, attorneys, appraisers and any other persons, to visit and inspect on at least twenty-four hours advance notice during its normal business hours and at any other reasonable times any of the Property or Borrower's chief executive office and to make photographs thereof, and to write down and record any information such representatives obtain, and shall permit Lender or its representatives to investigate and verify the accuracy of the information furnished to Lender under or in connection with this Agreement or any of the Loan Documents and to discuss all such matters with its officers, employees and representatives. Borrower will furnish to Lender at Borrower's expense all evidence which Lender may from time to time reasonably request as to the accuracy and validity of or compliance with all representations and warranties made by Borrower in the Loan Documents and satisfaction of all conditions contained therein. Any inspection or audit of the Property or the books and records of Borrower, or the procuring of documents and financial and other information, by or on behalf of Lender, shall be at Borrower's expense and shall be for Lender's protection only, and shall not constitute any assumption of responsibility or liability by Lender to Borrower or anyone else with regard to the condition, construction, maintenance or operation of the Property, nor Lender's approval of any certification given to Lender nor relieve Borrower of any of Borrower's obligations. Lender may divulge to any Investor (as hereinafter defined) all such information, and furnish to such Investor copies of any such reports, financial statements, certificates, and documents obtained under any provision of this Agreement, or related agreements and documents, provided that such information shall be provided on a confidential basis.

(d) Without limiting Lender's other rights and remedies under this Agreement and the other Loan Documents, in the event that any statement or document required to be delivered to Lender pursuant to this Section 2.04 is not delivered within thirty (30) days of the date when the same is due "Delinquent Reports") upon fifteen (15) days notice to Borrower, the

Interest Rate (as defined in the Note) shall be increased by adding one quarter percent (0.25%) per annum (i.e. 25 basis points) to the Interest Rate until such time as all Delinquent Reports, in form and substance reasonably satisfactory to Lender, have been delivered to Lender.

## Article III. CONDITIONS TO ADVANCE

Section 3.01 Conditions

The obligation of the Lender to make the Advance is subject to the satisfaction of all of the following conditions precedent (the "Conditions") (each of which is deemed material and none of which may be waived except by an instrument executed by the Lender):

- (i) The Lender shall have received:
- (1) the irrevocable Notice to Borrow Funds;

(2) pertified copy of the resolutions or consent adopted by the members of Borrower, in form and substance reasonably satisfactory to Lender authorizing the execution, delivery and performance of the Security Instrument and all of the other related Loan Documents required at the time of the Advance, including, but not limited to the Notice to Borrow Funds;

- (3) copies of the most recent year end Financial Statements;
- (4) a certificate from Borrower that all the representations and warranties set forth herein shall be true and correct on and as of such time with the same effect as though such representations and warranties have been made on and as of such time, except to the extent such representations and warranties expressly relate to an earlier date; and
- (5) a certificate of Borrower, stating no proceedings exist affecting Borrower or the Property that could have a Material Adverse Effect on Borrower, the Loan, the First Mortgage or the Property, and (B) a Certificate from the Guarantors, to the effect that no proceedings exist affecting any Guarantor which could have a material adverse effect on such Guarantor, in each case, which have not been disclosed to Lender in writing.
- (ii) Borrower shall be in compliance with all the terms and provisions set forth herein on its part to be observed or performed and no "Event of Default" (as hereinafter defined), nor any event which upon notice or lapse of time or both would constitute an Event of Default, shall have occurred and be continuing either at the time of the Advance or as a result of such borrowing, including but not limited to compliance with all financial reporting requirements set forth herein
  - (iii) Borrower shall be in compliance with all the terms and provisions set forth in the First Mortgage Loan Documents on its part to be observed or performed and no

- "Event of Default" (as defined in the First Mortgage Loan Documents), nor any event which upon notice or lapse of time or both would constitute an Event of Default pursuant to the First Mortgage Loan Documents, shall have occurred and be continuing either at the time of the Advance or as a result of such borrowing, including but not limited to compliance with all financial reporting requirements set forth therein;
- (iv) Lender's computation that the Borrower shall maintain a Debt Yield of no less than 11% during the term of the Loan. In the event that Lender shall at any time determine that the Debt Yield is less than 14%, Borrower shall within thirty (30) days of notice and demand by Lender, either reduce the Loan amount (by the payment of principal) or pledge such additional collateral as may be acceptable to Lender in order to maintain the required Debt Yield, Borrower's failure to either reduce the Loan balance as necessary or satisfy Lender's demand for additional collateral acceptable to it within thirty (30) days of notice having been given by Lender, shall be considered an Event of Default hereunder. "Debt Yield" shall mean, as of any date, the quotient (expressed as a percentage) obtained by dividing (i) the net operating income for the twelve (12) month period ending with the most recently completed calendar month by (b) the outstanding principal balance of the Note and the First Note as of such date. Lender's calculation of the Debt Yield, and all component calculations, shall be conclusive and binding on Borrower absent manifest error;
- (v) Immediately prior to the Advance, Lender shall cause title company to file and record the Security Instrument substantially in form attached hereto as Exhibit A and related certificates for the Property in the appropriate recording office in New York County. The Borrower shall (i) pay any and all expenses of the Lender incurred in connection with the foregoing matters, including, without limitation, the costs of any title insurance, recording fees, unvey, environmental reports and the reasonable fees and expenses of the Lender's counsel and (ii) deliver any documents reasonably requested by the Lender or the title company in connection with effectuating the foregoing matters;
- (vi) Lender shall have received a NY ALTA (1992) loan policy of title insurance (the "Loan Title Policy") in the amount of the Advance, substantially in the form attached hereto as Exhibit B issued by the title company to Lender, insuring the Security Instrument as a valid and subsisting second mortgage lien on the Property subject only to standard and customary permitted exceptions, including the lien of the First Mortgage. The Loan Title Policy shall contain such affirmative insurance and endorsements as Lender shall reasonably require;
- (vii) opinion(s) of counsel as to the authority, due execution, and enforceability of the Security Instrument and related Loan Documents in form substantially similar to the opinion given in connection with this Agreement or otherwise reasonably acceptable to the Bank;
- (viii) a certificate from Borrower and other evidence satisfactory to Lender in its reasonable discretion, including but not limited to, an agreement of purchase and sale in connection with the acquisition of floor area rights and/or air rights and/or other

development or zoning rights that will benefit the Property as same will be more particularly described in the Security Instrument;

- (ix) payment to Lender of a standby fee equal to one quarter of one percent (0.25%) of the Unused Portion on the first anniversary of the Loan and every anniversary thereafter until the Loan is paid in full, provided, however, Borrower shall pay Lender a pro rata portion of such standby fee if the Loan is paid in full prior to any given one year anniversary of the Loan. The Unused Portion shall mean the difference, if any, between the maximum aggregate principal amount that may be advanced under the Note and the Advance; and
  - (x) payment of all of Lender's expenses in connection with the review and Borrower's compliance with the Conditions including but not limited to Lender's reasonable attorneys' fees and expenses.

## Article IV. DEFAULT

## Section 4.01 Events of Default.

Each of the following shall constitute an "Event of Default" under this Agreement:

- (a) if any representation or warranty of Borrower or of its members, general partners, principals, affiliates, agents or employees, or of any Guarantor made herein or in the Environmental Indemnity or in any Loan Document, in any guaranty, or in any certificate, report, financial statement or other instrument or document furnished to Lender shall have been false or misleading in any material respect when made;
- (b) The failure to make any payment of principal or interest under the Note on or before the tenth (10 : h) calendar day after the day on which the same is due (without regard to any applicable cure and/or notice period) or if the entire Debt is not paid on or before the Maturity Date, along with applicable prepayment premiums, if any
- (c) if Borrower, or its general partner, member or manager, if applicable, violates or does not comply with any of the provisions of Section 3.3, Section 4.3, or Article 8 hereof, or Article 19 of the Note, or fails to deliver any of the reports required by Section
- (d) if Borrower or any Guarantor shall make an assignment for the benefit of creditors or if Borrower or any Guarantor shall admit in writing its inability to pay, or Borrower's or any Guarantor's failure to pay its debts as they become due;
- (e) if (i) Borrower or any member or manager of Borrower, or any Guarantor shall commence any case, proceeding or other action (A) under any existing or future law of any jurisdiction, domestic or foreign, relating to bankruptcy, indovency, reorganization, conservatorship or relief of debtors, seeking to have an order for relief entered with respect to it, or seeking to adjudicate it a bankrupt or insolvent, or seeking reorganization, arrangement, adjustment, winding-up, liquidation, dissolution, composition or other relief with respect to it or its debts, or (B) seeking appointment of a receiver, trustee, custodian, conservator or other

similar official for it or for all or any substantial part of its assets, or Borrower or any member or manager of Borrower, or any Guarantor shall make a general assignment for the benefit of its creditors; or (ii) there shall be commenced against Borrower or any member or manager of Borrower, or any Guarantor any case, proceeding or other action of a nature referred to in clause (i) above which (A) results in the entry of an order for relief or any such adjudication or appointment or (B) remains undismissed, undischarged or unbonded for a period of ninety (90) calendar days; or (iii) there shall be commenced against Borrower or any member or manager of Borrower or any Guarantor any case, proceeding or other action seeking issuance of a warrant of attachment, execution, distraint or similar process against all or any substantial part of its assets which results in the entry of any order for any such relief which shall not have been vacated, discharged, or stayed or bonded pending appeal within ninety (90) calendar days from the entry thereof; or (iv) Borrower or any member or manager of Borrower, or any Guarantor shall take any action in furtherance of, or indicating its consent to, approval of, or acquiescence in, any of the acts set forth in clause (i), (ii) or (iii) above; or (v) Borrower or any member or manager of Borrower, or any Guarantor shall generally not, or shall be unable to, or shall admit in writing its inability to, pay its debts as they become due;

- (f) The occurrence of any materially adverse change in the financial condition of the Borrower or any Guarantor (for purposes of this Agreement, a "materially adverse change" shall mean one which in the determination of the Lender that, in its sole judgment, impairs the ability of the Borrower or any Guarantor with respect to (i) maintaining and/or operating the Property, (ii) paying all real estate and, related taxes and charges and (iii) meeting all of its financial obligations to the Lender and all other creditors);
- (g) The occurrence of any default beyond the expiration of any applicable notice, grace or cure period under (i) any mortgage(s) now or hereafter covering the Property, giving rise to a right to accelerate payment thereof regardless of whether such mortgage(s) is/are held by Lender or a third party or (ii) any other agreement (loan or otherwise) between Lender and Borrower now existing or hereafter made;
- (h) The failure to perform any of the other covenants, terms or agreements on the part of the Borrower to be performed hereunder or under any of the other Loan Documents beyond any applicable notice and cure period;
- (i) if any federal tax lien is filed against Borrower, any general partner, member or manager of Borrower, any Guarantor or the Property and same is not discharged of record within forty-five (45) calendar days after Borrower becomes aware of such filing;; or
- U) Any "Event of Default" (as defined in the First Mortgage Loan Documents) or any event which upon notice or lapse of time or both would constitute an Event of Default occurs pursu to the First Mortgage Loan Documents.

(a) Upon the occurrence of an Event of Default, the Lender, in its sole and absolute discretion, may (1) temporarily suspend the extension of credit and refuse to honor any request for an Advance, (2) demand the prepayment of the Advance, (3) pursue any other rights

or remedies available to the Lender under this Agreement or the other Loan Documents, (4) terminate this Agreement and any obligation hereunder to make the Advance by declaring the loan to be immediately due and payable and/or exercising such of the other remedies provided for in the Loan Documents as the Lender may elect. The costs and expenses of Lender incurred in carrying out any or all of the above and enforcing the terms of this Agreement or the Note, including reasonable attorneys' fees, shall be at Borrower's expense.

- (b) Without limiting any remedy otherwise available to the Lender, the Borrower shall pay a late charge, to the extent permitted by law, of five (\$.05) cents per each dollar (\$1.00) of each payment received and accepted by the Lender more than ten (10) days after the date it is due, to cover the extra expense involved in handling such delinquent payment.
- (c) Anything to the contrary herein notwithstanding, Lender shall give written notice to Borrower with respect to any non-monetary defaults hereunder and Borrower shall have a period of thirty (30) days from the date of such notice to cure the default, if Borrower or any Guarantor, as the case may be, shall continue to be in default under any other term, covenant or condition of this Agreement or any Loan Documents for thirty (30) calendar days after notice from Lender; provided that if such default cannot reasonably be cured within such thirty (30) calendar day period and Borrower (or such Guarantor as the case may be) shall have commenced to cure such default within such thirty (30) calendar day period and therefore diligently and expeditionally proceeds to cure the same, such thirty (30) calendar day period shall be extended for so long as it shall require Borrower (or such Guarantor as the case may be) in the exercise of due diligence to cure such default, it being agreed that no such extension shall be for a period in excess of sixty (60) calendar days after the notice from Lender referred to above.

(d) If the Borrower fails to observe or perform any of the covenants or agreements on the part of the Borrower to be performed hereunder, then the Lender may, but shall not be obligated to, perform the same and all necessary and all reasonable coats incurred by the Lender in performing the Borrower's covenants and agreements, including reasonable coansel fees, shall be repaid by the Borrower upon demand, together with interest thereon at the default rate under the Note.

#### Section 4.03 No Waiver.

(a) Any failure of the Lender to exercise its option to declare the loan immediately due and payable, or any forbearance by the Lender before or after any exercise of such option, or any forbearance to exercise any other remedy of the Lender, or any withdrawal or abandonment of the Lender of any of its rights in any one circumstance, shall not be construed as a waiver of any option, power, remedy or right of the Lender hereunder except to the extent, if any, the action of the Lender constitutes an express waiver with respect to such one circumstance. The rights and remedies of the Lender expressed and contained in this Agreement and in the Loan Documents are cumulative and none of them shall be deemed to be exclusive of any other or of any right or remedy the Lender may not or hereafter have in law or in equity. The election of any one or more remedies shall not be deemed to be an election of remedies under any statute, rule, regulation or other law.

(b) The obligations of the Borrower (and the rights and remedies of the Lender against the Borrower) hereunder shall in no way be modified, abrogated, terminated or adversely affected by (i) any forbearance by the Lender in collecting any sums due, or (ii) the granting of any extension of time to perform any obligation hereunder or (iii) any impairment of the collateral, if any, which may now or hereafter be assigned or delivered to Lender to secure payment of the amounts due under the Note or hereunder, by reason of any act, failure to act or negligence of the Lender.

#### Article V. MISCELLANEOUS

Section 5.01 Notices.

All notices to be given hereunder shall be delivered by hand, or sent to the party to be notified via certified mail, return receipt requested or sent by a nationally recognized overnight courier which provides evidence of receipt and shall be deemed given when delivered by hand or one (1) day after delivery to such nationally recognized overnight courier or three (3) days after being posted with the United States Postal Service addressed to the parties as follows:

Santander Bank, N.A. 195 Montague Street Brooklyn, New York 11201 Attn: Elizabeth Bae

Windels Marx Lane & Mittendorf, LLP With a copy to:

156 West 56th Street New York, New York 10019 Attention: Michele Arbeeny, Esq.

Sutton Hill Properties, LLC 6100 Center Drive If to the Borrower at:

Suite 900 Los Angeles, California 90045 Attention: Andrzej Matyczynski

to:

Marcus Rosenberg & Diamond LLP 488 Madison Avenue
New York, New York 10022 Attention: Jeffrey M. Diamond, Esq.

Section 5.02 Successors and Assigns.

The terms Borrower and Lender, shall include the named Borrower and the named under and their respective legal representatives, successors and any permitted assigns.

Section 5.03 Severability.

If any one or more of the provisions contained in this Agreement or in any of the Loan Documents shall for any reason be held to be invalid, illegal or unenforceable in any respect, such invalidity, illegality or enforceability shall not affect any other provision of this Agreement or of the Note or any Loan Document.

Section 5.04 Expenses of Lender.

The Borrower shall pay all out-of-pocket expenses, including but not limited to, reasonable counsel fees incurred by the Lender in connection with the preparation execution and delivery of this Agreement and the enforcement or amendment of any of its rights or provisions hereunder.

Section 5.05 Indemnity.

The Borrower shall indemnify and hold harmless the Lender from and against any and all liabilities, obligations, losses, damages, penalties, claims, actions, suits, proceedings, judgments, costs, expenses and disbursements, including but not limited to, counsel fees in any way relating to or arising out of the failure of the Borrower to perform in full its obligations under this Agreement or under any of the Loan Documents.

Section 5.06 Applicable Law.

This Agreement shall be governed by and construed in accordance with the laws of the State of New York without regard to conflict of laws principles.

Section 5.07 Jurisdiction.

ANY ACTION ORPROCEEDING IN CONNECTION WITH THIS AGREEMENT MAY BE BROUGHT IN A COURT OF RECORD OF THE STATE OF NEW YORK, COUNTY OF NEW YORK, THE PARTIES HEREBY CONSENTING TO THE JURISDICTION THEREOF.

Section 5.08 Waiver of Certain Defenses.

IN ANY ACTION OR PROCEEDING IN CONNECTION WITH THIS AGREEMENT, OR ANY OTHER LOAN DOCUMENT, THE BORROWER WAIVES ANY CLAIM THAT NEW YORK IS AN INCONVENIENT FORUM AND FURTHER WAIVES THE RIGHT TO INTERPOSE ANY DEFENSE BASED UPON ANY STATUTE OF LIMITATIONS OR ANY CLAIM OF LACHES AND ANY SET-OFF OR COUNTERCLAIM OF ANY NATURE OR DESCRIPTION EXCEPT FOR ANY COUNTERCLAIMS DEEMED COMPULSORY UNDER APPLICABLE COURT RULES OR STATUTES.

Section 5.09 Waiver of Jury Trial and Waiver of Certain Damages.

IN ANY ACTION OR PROCEEDING IN CONNECTION WITH THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, THE LENDER AND BORROWER MUTUALLY WAIVE, TO THE FULLEST EXTENT PERMITTED BY LAW, TRIAL BY JURY AND BORROWER HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY CLAIM FOR CONSEQUENTIAL, PUNITIVE OR SPECIAL DAMAGES.

## Section 5.10 Joint and Several Liability.

If this Agreement is executed by more than one person or entity, all representations, warranties, obligations and covenants made by the Borrower hereunder shall be deemed to have been made by each of such persons and entities and the obligations and duties of such parties hereunder shall be deemed to be joint and several in all respects.

#### Section 5.11 Origination Fee.

This Agreement shall not become effective unless Lender has received, (a) payment of Borrower to Lender of the origination fee to Lender and (b) payment by Borrower of legal fees and expenses incurred by Lender to the Lender's counsel.

#### Section 5.12 Counterparts.

This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original, but all of which, when taken together, shall constitute one and the same instrument and shall become effective when copies hereof, when taken together, bear the signatures of each of the parties hereto and it shall not be necessary in making proof of this instrument to produce or account for more than one of such fully executed counterparts.

#### Section 5.13 Satisfaction of Note

At such time as the First Mortgage or any other mortgage(s) covering the Property, whether held by Lender or another lending institution, becomes due and payable, whether upon the maturity thereof or by acceleration or otherwise, or upon prepayment thereof, the Borrower shall also be required to immediately repay to Lender all amounts outstanding under the Note and this Agreement and if the Advance is not outstanding at such time, no advance shall be available hereunder.

### Section 5.14 Authorization.

The execution and delivery of this Agreement and the Note referred to herein have been duly authorized by the members of Borrower.

### Section 5.15 Cooperation

In the event Borrower determines that it would be beneficial to have the proceeds of the Loan in the form of multiple advances, Lender agrees to reasonably cooperate with Borrower with respect to the modification of the Loan Documents to provide for multiple advances subject to Borrowers payment of all costs and expenses in connection with any such modification.

## Separation and Release Agreement

This Separation and Release Agreement (this "Agreement") dated as of May 30, 2014, will be effective as of the lapse of the Revocation Period set forth in Section 9 hereof (the "Effective Date"), and is made and entered into by and between Andrzej Matyczynski ("Executive" or "you") and Reading International, Inc., a Nevada corporation ("Reading" or the "Company").

### RECITALS:

WHEREAS, Executive has been employed by Reading and certain affiliates as its Chief Financial Officer ("CFO"), Treasurer and Corporate Secretary pursuant to the terms of an employment agreement dated October 28, 1999 (the "Employment Agreement");

WHEREAS, you and the Company desire an appropriate strategy for you to retire and amicably end your career with the Company;

NOW, THEREFORE, in consideration of the mutual covenants hereinafter set forth, the parties hereto agree as follows:

- 1. Resignation as Chief Financial Officer. Effective immediately upon request of the Company's Chief Executive Officer ("CEO") (the "Resignation Date"), you will resign as the Company's Chief Financial Officer, Treasurer and Corporate Secretary and from any and all other positions that you hold as an officer, director and/or manager of the Company and its various direct and indirect subsidiaries. Your status as a corporate officer, director, manager or any fiduciary position (including as a fiduciary of any employee benefit plans sponsored by the Company or any affiliate) with the Company and all affiliates will end on the applicable Resignation Date and you hereby agree to submit your written resignation from any such offices and positions upon request on or after the Resignation Date, effective as of the Resignation Date. You acknowledge and agree that your Resignation Date may differ from office to office and from position to position.
- 2. Transition Period Services. Commencing on the date of this Agreement and continuing through the Retirement Date (defined below), Executive Agrees to continue to serve as the Company's CFO and Principal Financial Officer for Securities Exchange Act reporting purposes until a successor CFO and/or Principal Financial Officer is appointed and to provide such advice and transition assistance to the Company as the CEO of the Company may reasonably designate from time to time, including to continue to perform the duties of the CFO and Principal Financial Officer during the Transition Ferried on an "at will" basis (hereinafter, the "Services"). Executive hereby agrees to execute as CFO and Principal Financial Officer all appropriate filings to be made by the Company under the Securities Exchange Act during the period prior to the effective date of the appointment of a successor CFO. The period between the date of this Agreement and the Retirement Date is hereinafter referred to as the "Transition Period, Executive's duties will be substantially the same as present, but may be limited from time to time by the CEO. During the Transition Period, ovu agree to spend sufficient time on Company matters to perform the duties of the CFO and

Principal Financial Officer until a new CFO is appointed and to facilitate an orderly transition of responsibilities to the new CFO and, after a new CFO is appointed, to continue to make yourself available as needed, up to a full time basis, to attend to Company matters as requested from time to time. Unless otherwise specified by the CEO, services will be performed at the Company's executive headquarters building, in space provided by the Company, will provide for parking during the Transition Period. It is anticipated that, unless otherwise agreed, Executive will be available at the Company's executive headquarters building Monday through Thursday of each week. On each Friday during the Transition Period, Executive may provide Services remotely. In providing the Services during the Transition Period, you will continue to be an employee of the Company but will not have the authority to speak on behalf of or bind the Company except as authorized by the CEO of the Company or as required to discharge your duties in connection with providing the Services and performing the duties of the CFO and Principal Financial Officer. All compensation under this Agreement, including any compensation attributable to the Services, will be subject to tax withholding by the Company. The Company may terminate this Agreement timediately and without prior notice if Executive refuses to or is unable to perform the Services or is in breach of any material provision of this Agreement. Termination due to Executive's "Retirement Date" in Principal Financial Officer. It is referred to herein as "Termination for Cause"). Executive's "Retirement Date "will be the last day Services are provided during the period starting on the date of this Agreement and ending on September 1, 2014, or such earlier termination date.

3. Restrictive Covenants. Due to Executive's position with and relationship to the Company and its affiliates, Executive has had, and/or shall have, access to confidential or proprietary data or information of the Company and/or any affiliates of Company. This confidential information includes, but is not limited to, the Company's human resources and Executive-related information, strategic business plans, budgets, financial performance and financial statements, operational information, business and employment contracts, compensation information, and other information that the Company treats as confidential or proprietary. Executive agrees he will not disclose or use the Company's confidential or proprietary information. Executive further agrees that the terms of any and all Company policies regarding trade secrets, confidential or proprietary information will continue to apply to him after the Retirement Date. Executive understands that the Company may seek from a court of competent jurisdiction an injunction to prohibit such disclosure. For so long as Executive is employed by the Company, and for a period of one year after the Retirement Date, Executive shall not directly or indirectly, either alone or in concert with others, solicit or entice any employee of the Company or any of its affiliates, which independent contractor of the Company or any of its affiliates, which independent contractor is primarily engaged in business related to the Company or any of its affiliates or any individual whose employment or business relations with the Company and/or any of its affiliates ceased at least six months prior to Executive's solicitation or enticement of the individual. The provisions of this Section 3 and the Company policies that relate to trade secrets, confidential and proprietary information and non-solicitation of employees will survive the termination of your employment and are incorporated in this Section 3 and the Company policies that relate to trade secrets, confidential and proprietary information and

the provisions of the Restrictive Covenants and the provisions of this Agreement. In the event of any violation by you of the Restrictive Covenants or the provisions of this Agreement, no further payments will be made under Section 4.1 and no additional vesting will occur under Section 4.2. Your right to any unpaid payments under Section 4.1 and any unvested equity awards under Section 4.2 will be forfeited.

4. Compensation. In exchange for your Services, the restrictive covenant obligations described above, the Release provided below and satisfaction of all contractual obligations under the terms of the Employment Agreement, you will receive the following compensation and benefit treatment:

(a) Base Salary: Accrued Obligations. As consideration for your agreement to provide Services under Section 2 and your agreement to the Restrictive Covenants under Section 3, you will continue to receive your current base salary and employee benefits during the Transition Period. You will continue to paid through the end of the Transition Period or earlier termination of the Agreement. On the Retirement Date, you will receive payment in respect of your accrued and unused vacation, accrued but unpaid base salary through the Retirement Date and reimbursement of unreimbursed business expenses for which substantiation has been submitted in accordance with the Company's policies and procedures (collectively, the "Accrued Obligations"). You acknowledge and agree that as of the date of this Agreement, you have 200 hours of accrued and unused vacation. The amount of accrued and unused vacation will be adjusted for additional accruals and vacation time taken during the Transition Period. You acknowledge and agree that you will consult with the CEO regarding scheduling any vacation time during the Transition Period to accommodate the Company's requirements and to not unduly interfere with the performance of the Services.

(b) COBRA. You will be offered the opportunity to receive continuation coverage for yourself and your eligible dependents under the Company's medical and dental plans pursuant to the provisions of the Consolidated Onnibus Budget Reconciliation Act of 1985, as amended ("COBRA") following the Retirement Date, provided you timely elect and pay for such coverage.

(c) Severance Pay. No later than the eighth (8 ") day following the Retirement Date, provided that there has not been a Termination for Cause and that the executed Reaffirmation Agreement described in Section 7.6 has been delivered to the Company without revocation, you will be entitled to a lump sum payment of severance in an amount equal to \$244,500.00, which is comprised of a "leaving bonus" equivalent to six months of your annual base salary in the amount of \$154,500, an additional "leaving bonus" in the amount of \$50,000 and an additional payment of \$40,000 (collectively, the "Severance Pay will be subject to applicable tax withholding and will be further reduced by \$33,000, representing an offset for repayment of the outstanding loan (which loan was grandfathered under Section 13(k) of the Securities Exchange Act as amended by the Sarbanes-Oxley Act of 2002) to Executive from the Company (the "Loan Offset"). Executive acknowledges and agrees that payment of the Severance Pay, reduced by the Loan Offset, satisfies any and all "leaving

bonus" obligations under the Employment Agreement and is in excess of the amount Executive otherwise would be entitled to,

- (d) Deferred Compensation Plan . Reference is made to that certain Non-Qualified Deferred Compensation Plan between the Company and the Executive dated as of October 19, 2012 (the "Deferred Compensation Plan"). Provided that there has not been a termination "For Cause" as defined in the Deferred Compensation Plan, Executive will continue to be eligible to receive the benefits of the Deferred Compensation Plan according to its terms. If, in the determination of the Board, Executive has satisfactorily performed the Services and all of Executive's other obligations under this Agreement, and provided that there has been no Termination for Cause under this Agreement and that the executed Reaffirmation Agreement described in Section 7.6 has been delivered to the Company without revocation, the Board of Directors will authorize an allocation under the Deferred Compensation Plan for the 2014 plan year and will make a corresponding contribution to the grantor trust in the amount of \$50,000 (representing a pro-rata amount of the annual allocation and contribution for 2014 through the Retirement Date). The timing and the amount of the distribution under the Deferred Compensation Plan of Executive's vested benefit will be determined according to the terms of the Deferred Compensation Plan.
- 4.2 Equity Awards . All unvested stock options, restricted stock or other equity compensation granted to Executive prior to the Resignation Date will continue to vest through the Retirement Date.
- 4.3 O ther Compensation Matters. Notwithstanding anything to the contrary contained in this Agreement (including the Release set forth in Section 7 hereof), you hereby acknowledge that, in connection with your Resignation and Retirement when you cease to be an employee of the Company, you will not be entitled to receive from the Company or an affiliate (i) any additional severance pay or benefits except as provided in Section 4.1 and Section 4.2, or (ii) any retiree termination welfare benefits (other than health care continuation coverage that you may be entitled to elect pursuant to Section 4980B of the Code), in each case including, but not limited to any severance pay or benefits pursuant to the Employment Agreement. Your participation in all Company perquisites will cease as of the Retirement Date.
- 5. Nondisparagement. The Executive agrees to refrain from making any false or misleading statements or comments about the Company and any of its respective affiliates, their officers, directors, personnel, or any of their products and services. The Executive agrees to refrain from making any disparaging remarks to any person (other than comments to Executive's immediate family members or advisers that are made on a confidential basis and are not repeated or published by such persons) about the Company and any of its respective affiliates, their officers, directors, their respective personnel, and their respective products and services; except to the extent otherwise required by applicable law.

- 6. Cooperation. In order to ensure a smooth transition from Executive's employment with Company, Executive agrees to provide reasonable assistance to and cooperation with Company following the Retirement Date in connection with any Company matters for which Executive had knowledge or responsibility while employed by Company. If Company is involved in any legal action or investigation after Executive's Retirement Date relating to events which occurred during Executive's employment (including unitout limitation, the Malulani Investments Limited, Starn O'Toole, the Blumenfeld Enterprises Inc. and the Urquhart matters) Executive with the Company to the fullest extent reasonably possible (taking into consideration Executive's schedule and other commitments) in the preparation, prosecution, or defense of the Company's case, including, but not limited to, required travel, appearances and testimony, the execution of affidavits or documents or providing information requested by the Company. Company will reimbursts Executive for reasonable pre-approved out-of-pochet expenses and reasonable pre-approved compensation (if Executive is no longer receiving Base Salary or severance payments), based on an hourly rate of \$200 per hour, for time related to such assistance.
- 7. Release. You hereby acknowledge that the Company's obligations under Section 4 hereof are in excess of any payments or benefits to which you are entitled under law, contract or otherwise and are contingent upon your timely execution of, and failure to revoke this Agreement, including the release of claims set forth in this Section 7 (the "Release"). In the event that you do not timely execute the Agreement or if you timely revoke the Agreement as described below, the Company will have no under this Agreement. For purposes of this Section 7, "Released Parties" include the Company and its affiliated companies and their officers, directors, shareholders, employees, agents, representatives, plans, trusts, administrators, fiduciaries, insurance companies, successors, and assigns.
- 7.1 You, on behalf of yourself and your personal and legal representatives, heirs, executors, successors and assigns, hereby acknowledge full and complete satisfaction of, and fully and forever waive, release, and discharge the Released Parties from any and all claims, causes of action, demands, liabilities, damages, obligations, and debts (collectively referenced as "Claims"), of every kind and nature, whether known or unknown, suspected or unsuspected, that you hold as of the date you sign this Agreement, or at any time previously held against any Released Party, arising out of any matter whatsoever (with the exception of breach of this Agreement). This release specifically includes, but is not limited to, any and all Claims:
- (a) Arising out of or in any way related to your employment with or separation of employment from the Company, or any contract or agreement between you and the Company or the termination thereof;
- (b) Arising out of or in any way related to any treatment of Executive by any of the Released Parties, which shall include, without limitation, any treatment or decisions with respect to hiring, placement, promotion, discipline, work hours, demotion, transfer, termination, compensation, performance review, or training; (iv) any statements or alleged statements by the Company or any of the Released Parties regarding Executive, whether oral or in writing; (v) any damages or injury that Executive may have suffered, including without limitation, emotional or physical injury, compensatory damages, or lost wages; or (vi) employment discrimination, which shall include, without limitation, any individual or class

claims of discrimination on the basis of age, disability, sex, race, religion, national origin, citizenship status, marital status, sexual preference, or any other basis whatsoever.

- (c) Arising under or based on the Equal Pay Act of 1963 (EPA); Title VII of the Civil Rights Act of 1964, as amended (Title VII); Section 1981 of the Civil Rights Act of 1866 (42 U.S.C. §1981); the Civil Rights Act of 1991 (42 U.S.C. §1981a); the Americans with Disabilities Act of 1990, as amended (ADA); the Family and Medical Leave Act of 1993, as amended (FMLA); the Genetic Information Nondiscrimination Act of 2008 (GINA); the National Labor Relations Act (NLRA); the Worker Adjustment and Retaining Notification Act of 1988 (WARN); the Uniform Services Employment and Retaining Notification Act of 1974; the Occupational Safety and Health Act (OSIIA); the Employee Retirement Income Security Act of 1974 (RRISA) (except claims for vested benefits, if any, to which you are legally entitled); the False Claims Act; Title VIII of the Corporate and Criminal Fraud and Accountability Act, as amended (18 U.S.C. §1514A) (Sarbanes-Oxley Act); the federal Whistleblower Protection Act and any state whistleblower protection statute(s); the California Fair Employment and Housing Act or any other federal, state or local law relating to employment or any other fair employment protection of any state, in all cases arising out of or relating to your employment by Reading or investment in Reading or your services as an officer or employee of Reading or its subsidiaries, or otherwise relating to the termination of such employment or services.
- (d) Arising under or based on any other federal, state, county or local law, statute, ordinance, decision, order, policy or regulation prohibiting employment discrimination; providing for the payment of wages or benefits (including overtime and workers' compensation); or otherwise creating rights or claims for employees, including, but not limited to, any and all claims alleging breach of public policy; the implied obligation of good faith and fair dealing; or any express, implied, oral or written contract, handbook, manual, policy statement or employment practice, including, but not limited to, the Employment Agreement; or alleging misrepresentation; (belan stander; interference with contractual relations; intentional or negligent infliction of emotional distress; invasion of privacy; assault; battery; fraud; negligence; harassment; retaliation; or wrongful discharge; and
- (e) Arising under or based on the Age Discrimination in Employment Act of 1967 ("ADEA"), as amended by the Older Workers Benefit Protection Act ("OWBPA"), and alleging a violation thereof by any Released Party, at any time prior to the date you sign this Agreement.
- 7.2 You agree that, except as set forth in this Agreement, you are not entitled to any payment or benefits from any of the Released Parties, including, but not limited to, any payments or benefits under any plan, program or agreement with any Released Party, including, but not limited to, the Employment Agreement.
- 7.3 You agree that, this Agreement extinguishes all claims and charges that you could have raised against any of the Released Parties, whether known to you or not. You expressly waive all rights and benefits under Section 1542 of the California Civil Code and any

similar law of any state or territory of the United States. Section 1542 of the California Civil Code provides as follows:

- "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor."
- $7.4\ You\ hereby\ represent\ that\ you\ know\ of\ no\ claim\ that\ you\ have\ that\ has\ not\ been\ released\ by\ this\ Section\ 7\ .$
- 7.5 Nothing contained in this Release will (i) release any claim that cannot be waived under applicable law, (ii) release your rights to any benefits under any employee welfare benefit plan of the Company, the 401(k) Plan or with respect to the right to elect health care continuation under COBRA, (ii) release any entitlement to or with respect to indemnification which you may have pursuant to agreement, the Company's bylaws, any policy of insurance maintained by the Company or otherwise under law, or (iv) be construed to release your rights under this Agreement or restrict you in any manner from bringing appropriate proceedings to enforce this Agreement. You acknowledge that your execution of this Agreement terminates any claims you previously held to any and all compensation and employee benefits, other than those specifically identified in this Agreement.
- 7.6 By signing this Agreement, you represent that you have not commenced or Joined in any claim, charge, action or proceeding whatsoever against any of the Released Parties arising out of or out of the matters set forth in this Section 7. You further represent that you will not be entitled to any personal recovery in any action or proceeding that may be commenced on your behalf arising out of the matters released hereby. As a condition to continued receipt of payments under Section there the Retirement Date, you hereby agree to realfirm the release terms of this Section 7 applicable to the period between execution of this Agreement and the Retirement Date by executing a reaffirmation agreement (the "Reaffirmation Agreement") substantially in the form attached hereto as Schedule A.

#### 8 General Provisions

- 8.1 Severability. It is the desire and intent of the parties that the provisions of this Agreement will be enforced to the fullest extent permissible. In the event that any one or more of the provisions of this Agreement will be held to be invalid, illegal or unenforceable, the validity, legality and enforceability of the remainder of this Agreement will remain valid and enforceable and continue in full force and effect to the fullest extent consistent with law. Moreover, if any one or more of the provisions contained in this Agreement is held to be excessively broad as to duration, scope, activity or subject, such provisions will be construed by limiting and reducing them so as to be enforceable to the maximum extent compatible with applicable law.
  - 8.2 No Admission. By entering into this Agreement, the parties do not admit to, and expressly deny, any wrongdoing.

- 8.3 Return of Property. You agree to return to the Company, on or prior to the Retirement Date, all files, records, documents, reports, computers and other property of the Company in your possession or control, including, but not limited to, any documents or other materials containing Confidential Information, and you further agree that you will not keep, transfer or use any copies or excerpts of the foregoing items. Executive will be permitted to copy and remove any electronic files on the computer or cell phone that contain his personal information (but not any Confidential Information or proprietary Company information or data), including contact information.
- 8.4 Notices. Any and all notices, requests, demands and other communications provided for by this Agreement will be in writing and will be effective when delivered in person, consigned to a reputable national or international courier service (including Federal Express), and addressed to you at your last known address on the books of the Company or, in the case of the Company, at the Company's principal place of business, attention of the President of the Company, or to such other address as either party may specify by notice to the other actually received.
- 8.5 Successors and Assigns. This Agreement is personal to you and, without the prior written consent of the Company, will not be assignable by you otherwise than by will or the laws of descent and distribution. This Agreement will inure to the benefit of and be enforceable by your legal representatives. This Agreement will inure to the benefit of and be binding upon the Company and its successors and assigns.
- 8.6 Governing Law; Captions; Amendment. This Agreement will be governed by, and construed in accordance with, the laws of the State of California, without reference to principles of conflict of laws. The captions of this Agreement are not part of the provisions hereof and will have no force or effect. This Agreement may not be amended or modified except by a written agreement executed by the parties hereto or their respective successors and legal representatives.
- 8.7 Code Section 409A Compliance. The Company and you each hereby affirm that it is their mutual view that the provision of payments and benefits described or referenced herein are either exempt from or intended to be in compliance with the requirements of Section 409A of the Code and the Treasury regulations relating thereto ("Section 409A") and that each party's tax reporting will be completed in a manner consistent with such view. The Company and you each agree that upon the Retirement Date, you will experience a "separation from service" for purposes of Section 409A. Any payments that qualify for the "short-term deferral" exception or another exception under Section 409A will be paid under the applicable exception. For purposes of the limitations on nonqualified deferred compensation under Section 409A of the Code, each payment of compensation. Notwithstanding anything contained herein to the contrary, to the extent required in order to avoid accelerated taxation and/or tax penalties under Section 409A, fee amounts in Section 4.1 that constitute nonqualified deferred compensation and would otherwise be payable pursuant to this Agreement on account of separation from service during the six-month period immediately following the Retirement Date will instead be paid on the first business day after the date that is six months following the Retirement Date (or death, if earlier). Notwithstanding anything to the contrary in this Agreement, all reimbursements and in-

kind benefits provided under this Agreement will be made or provided in accordance with the requirements of Section 409A of the Code, including, where applicable, the requirement that (x) the amount of expenses eligible for reimbursement, or in kind benefits provided, during a calendar year may not affect the expenses eligible for reimbursement, or in kind benefits to be provided, in any other calendar year (y) the reimbursement of an eligible expense will be made no later than the last day of the calendar year following the year in which the expense is incurred; and (z) the right to reimbursement or in kind benefits is not subject to liquidation or exchange for another benefits. Neither the Company nor its affiliates will be liable in any manner for any federal, state or local income or excise taxes (including but not limited to any taxes under Sections 409A of the Code), or penalties or interest with respect thereto, as a result of the payment of any compensation or benefits hereunder or the inclusion of any such compensation or benefits for the value thereof in your income. You acknowledge and agree that the Company will not be responsible for any additional taxes or penalties resulting from the application of Section 409A.

- 8.8 Withholding. Notwithstanding any other provision of this Agreement, the Company may withhold from amounts payable under this Agreement all amounts that are required or authorized to be withheld, including, but not limited to, federal, state, local and foreign taxes to be withheld by applicable laws or regulations.
- 8.9 Preparation of Agreement. This Agreement will be interpreted in accordance with the plain meaning of its terms and not strictly for or against any of the parties hereto. Regardless of which party initially drafted this Agreement, it will not be construed against any one party, and will be construed and enforced as a mutually-prepared document.
- 8.10 Entire Agreement. This Agreement constitutes the entire agreement between you and the Company with respect to the subjects addressed herein, and together with the Restrictive Covenants that survive, and the Reaffirmation Agreement supersede all prior agreements, understandings and representations, written or oral, with respect to those subjects, including, but not limited to the, Employment Agreement. Without limiting the generality of the foregoing, you acknowledge that the Employment Agreement will be terminated upon the effectiveness of this Agreement.
- 8.11 Legal Fees. The Company will reimburse you for the legal fees, incurred by you in connection with the negotiation and execution of this Agreement, up to a maximum of \$5,000. Such reimbursement will be made by the Company within twenty business days of your submission to the Company of an invoice or invoices from counsel, which submission will be made no later than June 15, 2014.
  - 8.12 Counterparts. This Agreement may be executed in counterparts, each of which will be deemed an original, and which together will be deemed to be one and the same instrument.
- 9. Consultation with Attorney; Voluntary Agreement. You understand and agree that you have the right and have been given the opportunity to review this Agreement and, specifically, the Release set forth in Section 7 above, with an attorney of your choice. You also understand and agree that you are under no obligation to consent to the Release. You

acknowledge that you have read this Agreement and the Release and understand their terms and that you enter into this Agreement freely, voluntarily, and without coercion. You acknowledge that you have been given at least twenty-one (21) days during which to review and consider the provisions of this Agreement and, specifically, the Release set forth in Section 7 above, although you may sign and return it sooner if you so desire. You further acknowledge that you have been advised by the Company that you have the right to revoke this Agreement for a period of seven (7) days after signing it (the "Revocation Period"). You acknowledge and agree that, if you wish to revoke this Agreement, you must do so in a writing, signed by you and received by the Company to the attention of James J. Cotter, Jr., President, no later than 5:00 p.m. Pacific Time on the seventh (7th) day of the Revocation Period. If no such revocation occurs, the General Release and this Agreement will become effective on the eighth (8th) day following your execution of this Agreement. You further acknowledge and agree that, in the event that you revoke this Agreement, it will have no force or effect.

10. Other Representations. You agree to execute such documents and take such actions as may be necessary or desirable to further effectuate the foregoing. Executive, by his initials set forth below, acknowledges and agrees that he was given a copy of this Agreement on the 20th day of May, 2014, to review and consider execution of the terms and conditions contained herein.

/S/ AJM Executive 's Initials

[Signature page follows]

## READ CAREFULLY BEFORE SIGNING

THIS SEPARATION AND RELEASE AGREEMENT INCLUDES A RELEASE OF ALL KNOWN AND UNKNOWN CLAIMS AND A WAIVER OF YOUR RIGHTS UNDER THE AGE DISCRIMINATION IN EMPLOYMENT ACT AS WELL AS OTHER FEDERAL, STATE AND LOCAL LAWS PROTECTING EMPLOYEE RIGHTS. IF YOU SIGN THIS AGREEMENT, YOU ARE WAIVING ALL OF YOUR RIGHTS TO ASSERT ANY CLAIMS UNDER THESE LAWS. PLEASE READ THIS AGREEMENT CAREFULLY AND SEEK THE ADVICE OF AN ATTORNEY REGARDING THE LEGAL EFFECT OF SIGNING THIS AGREEMENT.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement on the day and year written opposite their signature.

"Executive"	
Date: May 30, 2014 Andrzej Matyczynski, an individual	/s/ Andrzej Matyczynski
"Compan y" Reading International, Inc.	
Date: May 30, 2014	By: /s/ James J. Cotter  James J. Cotter , President

Aedrocj Matyczyoski EXHIBIT 19.29

Ro: Amendment of Separation and Release Agreement

Dosc Mr. Modyczynski,

Noticember to made to that certain Separatres and Kalenan Agraement dated so of Mary 30, 2014 (the Separatres Agraement). This latter is to confirm the following:

- 1. The last scotteness of Section 2 is hereby deleted in its entirety and replaced with the following: "Executive's "Refirement Dake" will be the last day Services are provided during the period starting on the dake of this Agreement and ending on Decumber 1, 2014, or such carlies terminosistes date."
- 2. The third centered of Sertion 4.1(d) is modified to detect the reference to \$50,900 and to replace it with the figure \$68,750.
- 3. Except as specifically provided above, the Separation Agricolated requires in hill force and effect and two put two put two provided or specifically provided above, the Separation Agricolated reacting in hill force and effect and two put two provided or specifically provided above, the Separation Agricolated reacting in hill force and effect and two put two provided above.

If this contestly states our agreement, please sign and return to us. Thank you for your cogning time and assistance.

Very truly yours,

/s/ James J. Cotter

James J. Cotter, President

ACCEPTED AND AGREED as of this 6 th day of August, 2014

/s/ Andrzej Matyczynski

Andrzej Matyczynski

## SECOND AMENDMENT

TO

## SEPARATION AND RELEASE AGREEMENT

This second amendment (the "Second Amendment") is entered into as of this 26 ° day of November, 2014, by and between Reading International, Inc. (the "Company") and Andrzej Matyczynski ("Executive") and amends that certain Separation and Release Agreement dated as of May 30, 2014 (the "Initial SRA"), as previously amended by that certain first amendment of Separation and Release Agreement dated effective August 6, 2014 (the "First Amendment" and collectively with the Initial SRA, the "First Amended SRA").

Any defined terms not specifically defined in this Second Amendment have the same meanings as set forth in the First Amended SRA. Except as specifically amended by this Second Amendment, the First Amended SRA remains in full force and effect and has not been modified or amended in any respect, and none of the respective rights or obligations of the parties thereunder have been waived or released.

- 1. Retirement Date: Section 1 of the First Amendment is hereby deleted in its entirety and replaced with the following: "Executive's "Retirement Date" will be the last day Services are provided during the period starting on the date of this Agreement and ending on June 1, 2015, or such earlier termination date."
- 2. Deferred Compensation: Section 2 of the First Amendment is hereby deleted in its entirety and replaced with the following: "The third sentence of Section 4.1(d) is amended (i) to replace the amount of \$50,000 and the immediately following parenthetical with the amount of \$75,000, and (ii) to add the following to the end thereof, "and will authorize an allocation under the Deferred Compensation P lan for the 2015 plan year and will make a corresponding contribution to the grantor trust in the amount of \$31,200 (representing a pro-rata mount of the annual allocation and contribution for 2015 the Retirement Days."
- 3. Notwithstanding any other provision of the First Amended SRA, it is agreed that Executive may do a majority of his work from his home in Atlanta, Georgia. Executive will only be required to be at the Company's Los Angeles offices for a total of ten (10) weeks (Monday through Thursday). It is anticipated that the parties will work together to sehedule such time in full week blocks, one week in one month, two weeks in the next month, one week in the next month, etc. The Company will promptly reimburse Executive for all reasonable travel expenses (against reasonable decounteration), including airfare, hotel and car rental (such travel expenses not to exceed \$1,200.00 per weekly trip).

  In Witness Whereof this Second Amendment is executed and delivered effective of the date first set forth above.

Reading International, Inc.

By: /s/ James J. Cotter		
James J. Cotter, Jr.		
Chief Executive Officer		
Executive:		
/s/ Andrzej Matyczynski		
Andrzei Matyczynski		

#### THIRD AMENDMENT

-

### SEPARATION AND RELEASE AGREEMENT

This T hird A mendment to Separation Agreement (th is "Third A mendment") is entered into effective as of May 1, 2015, by and between Reading International, Inc. (the "Company") and Andrzej Matyezynski ("Executive") and amends that certain Separation and Release Agreement dated as of May 30, 2014 (the "Initial SRA"), as previously amended by that certain F int A mendment of Separation and Release Agreement dated effective August 6, 2014 (the "First Amendment") and that certain Second Amendment to Separation and Release Agreement dated effective November 26, 2014 (collectively with the Initial SRA and the First Amendment, the "Amendment, the "Amen

Any defined terms not specifically defined in this Third Amendment have the same meanings as set forth in the Amended SRA. Except as specifically amended by this Third Amendment, the Amended SRA remains in full force and effect and has not been modified or amended in any respect, its terms and conditions are incorporated herein by reference, and none of the respective rights or obligations of the parties thereunder has been waived or released.

- 1. Retirement Date: Section 1 of the First Amendment is hereby deleted in its entirety and replaced with the following: "Executive's "Retirement Date" will be the last day Services are provided during the period starting on the date of this Agreement and ending on Agril 15, 2016, or such earlier termination date (such period, the "Term"). After May 11, 2015, or which date Executive shall cease to serve as the Company's Chief Financial Officer and Treasurer, Executive shall essee to serve as the Company Services as the Company may from time to the request to the additional financial advisory services as the Company may from time to the request. It is understood that these services will be provided as a full time employee of the Company; such services being referred as the "Services" for purposes of this Third Amendment and superseding the prior definition of "Services" as used in the Amendment SRA
- Deferred Compe nisition: Section 2 of the Second Amendment is hereby deleted in its entirety and replaced with the following: "T he third sentence of Section 4.1(d) is amended to change the reference to the "2014 plan year" to the "2015 plan year," to change the reference to "550,000" to 21,875," and thereafter to change the reference to "50,000" to 21,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to change the reference to "510,000" to 18,875," and thereafter to 18,875, and the reference to 18,875, a
- 3. Venue: Notwithstanding any other provision of the Amended SRA, it is agreed that during the Term, Exec utive may do a majority of his work from his home in Atlanta, Georgia. Executive will only be required to be at the Company's Los Angeles offices for a total of seventeen (1 7 ) weeks (Monday through Thursday) between the date hereof and the Retirement Date. It is anticipated that the parties will work together to schedule such time in full week blocks, one week in one month, two weeks in the next month, one week in the next
- 4. Salary: Section 4.1 of the Original SRA is hereby amended to provide that, effective January 1, 2015 and for the balance of the Term, Executive shall be paid as compensation for the performance of his duties

hereunder and under the Amended SRA, at a rate of \$312,000 per annum, in accordance with the Company's standard payment practices, prorated for the period January 1, 2016 through and including April 15, 2016.

- 5. Stock Option Vesting: Of the previously granted unvested Class A Stock options, 12,500 shall continue to vest on August 22, 2015. The balance of Executive 's Class A Stock options previously granted, totaling 12,500, shall vest on April 15, 2016, rather than August 22, 2016. As to any Class A Stock options vested as of the Retirement Date, the Executive shall have eighteen (18) months from his Retirement Date to exercise same, rather than three (3) months.
- 6. Release: Executive hereby affirms in all respects his Release of the Released Parties set forth in the Initial SRA as though made on the date hereof, as well as all other terms and conditions of the Amended SRA, including, without limitation, Sections 3, 5 and 6.

  In Witness Whereof this Third Amendment is executed and delivered effective of the date first set forth above.

Reading International, Inc.

By: /s/ James J. Cotter

James J. Cotter, Jr.

Chief Executive Officer

Executive:

/s/ Andrzej Matyczynski

Andrzej Matycazynski

# Amended and Restated Compensatory Arrangements for Executive and Management Employees

## Executive Compensation

From late January to late February 2016, the Compensation and Stock Options Committee ("Compensation Committee") of Reading International, Inc., ("Reading," "Registrant" or the "Company") met five separate times with Willis Towers Watson, an international compensation consulting firm, the Chief Executive Officer, and legal councel. As part of its engagement Willis Towers Watson reviewed the Company's compensation paid to executive and management officers by position, in light of each person's daties and responsibilities. Willis Towers Watson then compared the top executive and management survey Report and the 2015 Mercer MBD Executive Compensation Survey, in each case, identified by officer of the officer. The peer group utilized by Willis Towers Watson included the officer of the officer. The peer group utilized by Willis Towers Watson included the officer of the officer.

Areadia Realty Trust Associated Estates Realty Corp. Carmike Cinemas Inc. Cedar Realty Trust Inc. Charter Hall Group EPR Properties Vicinity Centres IMAX Corporation

Inland Real Estate Corp. Kite Realty Group Trust Marcus Corporation. Pennsylvania Real Estate Investment Trust Ramco-Gershenson Properties Trust Urstadt Biddle Properties Inc. Village Roadshow Ltd.

Willis Towers Watson selected the above peer group because (i) the companies included US and Australian based companies reflecting the Company's geographic operations and/or (ii) the companies were comparable to the Company based on revenue.

The executive pay assessment prepared by Willis Towers Watson measured the executive and management compensation paid by the Company against compensation paid by the peer group companies and the companies listed in the two surveys based on the 25 °, 50 ° and 75 ° percentile of such peer group and surveyed companies. The 50 ° percentile was the median compensation paid by such peer group and surveyed companies to executives performing similar responsibilities and duties.

In its report to the Compensation Committee, Willis Towers Watson noted that for Company executive officers

- Base salaries in the aggregate were generally in the competitive zone of the market (1% below the market 50 \*percentile), with certain notable exceptions on a position by position review;
- Total cash compensation (base salary and cash bonus) in the aggregate was 26% below the 50 <sup>a</sup>percentile; and

Total compensation (base salary, each bonus and long term incentive awards) in the aggregate was 40% below the 50 \*percentile.

The Compensation Committee, recommended, and the Board subsequently adopted, a compensation philosophy for the Company's management team members to:

- Attract and retain talented and dedicated management team members;
  Provide overall compensation that is competitive in its industry;
  Correlate annual cash incertives to the achievement of its business and financial objectives; and
  Provide management team members with appropriate long-term incentives aligned with stockholder value.

As part of the compensation philosophy the Company's compensation focus will be to (1) drive the Company's strategic plan on growth, (2) align officer and management performance with the interests of the Company's stockholders, and (3) encourage tion of officers and management team member

In furtherance of the compensation policy and as a result of the extensive deliberations, including consideration of the Willis Towers Watson recommendations, the Compensation Committee adopted an executive and management

- A base salary comparable with job description and industry standard;

  A short term incentive or cash bosus plan based on a combination of factors including individual performance against corporate goals as well as overall corporate and division performance, with a target bosus to be denominable with specific goals weightings and pay-out ranges); or

  A long term incentive or equity awards in line with job description, performance, and industry standards.

The Compensation Committee's intention is that the compensation structure approved for 2016 will remain in place indefinitely. However, it will review performance and results after the first year and thereafter and evaluate from time to time whether enhancements, changes or other compensation structures are in the Company's and it stockholders best interests.

Reflecting the new approach, the Compensation Committee established (i) annual base salaries at levels that it believed (based heavily on the data provided by Willis Towers Watson) are generally competitive with executives in our peer group and in other companable publicly-held companies as described in the executive pay assessment prepared by Willis Towers Watson, and (ii) short term incentives in the form of discretionary annual cash bossues based on the achievement of identified goals and benchmarks. Long-term incentives in the form of employee actors (expicious and restricted stock units will be used as a retention tool and as a means to further align an executive's long-term interests with those of the Company's stockholders, with the ultimate objective of affording our executives an appropriate incentive to help drive increases in stockholder value.

The Compensation Committee will evaluate both executive performance and compensation to maintain the Company's ability to attract and retain highly-qualified executives in key positions and to assure that compensation provided to executives remains competitive when compared to the compensation paid to similarly situated executives of companies with whom we compete for executive talent or that we consider comparable to our company.

## Role of Chief Executive Officer in Compensation Decisions

In connection with the implementation of the new compensation structure, the Compensation Committee conducted the thorough review of executive compensation discussed above. The Conconsidered with great weight the recommendations of the Chief Executive Officer as to compensation for executive and management team members other than for the Chief Executive Officer

In connection with consideration of 2015 performance beauses for members of management, the Chief Executive Officer pepared and submitted recommendations for each of the executive and management team members, other than her own. In considering these recommendations, the Compensation Committee had the benefit of its extensive deliberations as well as the data provided by Willis Towers Watson. In executive session, the Compensation Committee approved a 2015 performance bonus for the Chief Executive Officer. At the Compensation Committee's February 17, 2016 meeting, it approved recommendations to the Board of Directors (the "Board") for its February 18, 2016 meeting, at which time the Board approved the same. The Board approval certain officers including the five officers set forth below. In addition, our Chief Executive Officer decommendations for other management team members but the Compensation Committee and Board agreed that such positions were within the scope of the Chief Executive Officer's authority and did not require the Compensation Committee or Board approval.

The Compensation Committee expects to perform an annual review of executive compensation, generally in the first quarter of the year following the year in review, with a presentation by the Chief Executive Officer regarding each element of the executive compensation arrangements. At the Compensation Committee's direction, the Chief

Executive Officer prepared an executive compensation review for each executive officer (other than the Chief Executive Officer), as well as the full executive team, which included recommendations for:

- 2016 Base Salary
- . A proposed year-end short-term incentive in the form of a target cash bonus based on the achievement of certain objectives; and
- A long-term incentive in the form of stock options and restricted stock units for the year under review.

As part of the compensation review, the Chief Executive Officer may also recommend other changes to an executive's compensation arrangements such as a change in the executive's responsibilities or a change in title. The Compensation Committee will evaluate the Chief Executive Officer's recommendations and, in its discretion, may accept or reject the recommendations, subject to the terms of any written employment agreements.

The Compensation Committee met in executive session without our Chief Executive Officer to consider the Chief Executive Officer's compensation, including base salary, cash bonus and equity award, if any. Prior to such executive sessions, the Compensation Committee interviewed the Chief Executive Officer's contributing to the Chief Executive Officer's compensation. With the exception of these executive sessions of the Compensation Committee, as a rule, our Chief Executive Officer participated in all deliberations of the Compensation Committee relating to executive compensation. However, the Compensation Committee will ask the Chief Executive Officer to be excused for certain deliberations with respect to the compensation recommended for Margaret Cotter, the sister of the Chief Executive Officer.

In conjunction with the year-end annual compensation review, or as soon as practicable after the year-end, our Chief Executive Officer will recommend to the Compensation Committee the Company objectives and other criteria to be utilized for purposes of determining cash bonuses for certain senior executive officers. The Compensation Committee, in its discretion, may revise the Chief Executive Officer's recommendations. At the end of the year, the Compensation Committee, in consultation with the Chief Executive Officer, will review each performance goal and determine the ventrat to which the officer achieved auch goals, the resultability performance goals, the Compensation Committee expects to consider whether the goals could possibly result in an incentive for any executives to take unwarranted risks in our Company's business and intend to seek to avoid creating any such incentives.

#### Base Salarie

The Compensation Committee reviewed the executive pay assessment prepared by Willis Towers Watson and other factors and engaged in extensive deliberation and then recommended the following 2016 base salaries (the 2015 base salaries are shown for comparison purposes) for the following officers; the Board approved the recommendations of the Compensation Committee on March 10, 2016: the President and Chief Executive Officer, Chief Financial Officer and the persons identified and Named Executive Officers in the Company's proxy statement dated November 10, 2015 other than our prior Chief Executive Officers James J. Cotter, Jr:

Name	Tide	2016 Base Salary (4)	2015 Base Salary (4)
Ellen Cotter (1)	President and Chief Executive Officer	\$450,000	\$402,000
Devasis Ghose (2)	Chief Financial Officer	400,000	400,000
Andrzej Matyczynski (3)	EVP Global Operations	336,000	312,000
Robert F. Smerling	President, US Cinemas	375,000	350,000
Wayne Smith	Managing Director, ANZ	A\$370,000	A\$365,360

(1) Ellen M. Cotter was appointed Interim President on June 12, 2015 and President and Chief Executive Officer on January 8, 2016.

- (2) Devasis Ghose was appointed Chief Financial Officer on May 11, 2015. Mr. Ghose is the only executive officer that is a party to an employment agreement.

  (3) Andrzej Matyczynski was the Company's Chief Financial Officer until May 11, 2015 and thereafter he acted as corporate advisor to the Company. He was appointed EVP-Global Operations on March 10, 2016.

  (4) All dollars are in US dollars except the salary for Wayne Smith is reported in Australian dollars.

### Short Term Incentives

The Short Term Incentives authorized by the Compensation Committee and the Board provides the Company's executive officers and other management team members, who are selected to participate, with an opportunity to earn an annual cash bonus based upon the achievement of certain Company financial goals, division goals and individual goals, established by the Company's Chief Executive Officer and approved by the Compensation Committee and the Board of Directors (in future years, under the Compensation Committee Charter approved by the Board on March 10, 2016, the Compensation Committee will have full authority to approve these matters). Specifically, a participant in the short term incentive plan will be advised of his or her annual potential target bonus expressed as a percentage of the participant's base salary and by dollar amount. The participant will be eligible for a short term incentive bonus once the participant achieves goals identified at the beginning of the year for a threshold target, the potential target or potential marget bonus opportunity. The bonus will vary depending upon the achievements unde by the individual participant, the corporate division and the Company. Corporate goals will include levels of earnings before interest, depreciation, taxes and amortization ("non-GAAP Operating Income") and property development milestoned post in lineable levels of division cands those and amortization attends and some statements of the company. Each of the corporate, division and individual goals will include unique performance goals specific to the individual's position in the Company. Each of the corporate, division and individual goals will include unique performance goals specific to the individual's position in the Company. Each of the corporate, division and individual goals will include unique performance goals specific to the individual's position in the Company. Each of the corporate, division and individual goals will include will appear to the property development milestone.

For 2016, executive officers will have an annual bonus opportunity expressed and determined as a percent of their base salary. This approach also was a recommendation of the Willis Towers Watson report to the Compensation Committee and provided points of reference for our Compensation Committee to compare short-term incentive opportunities for our executive and management team to those in peer and competitor companies.

Ms. Ellen Cotter, President and Chief Executive Officer, has a potential target bosus opportunity of 95% of Base Salary, or \$427,500 at target based upon Ms. Cotter's achievement of her performance goals and the Company's achievement of corporate goals as discussed above. Of that potential target bosus opportunity, a threshold bosus of \$213,750 may be achieved based upon Ms. Cotter's achievement of her performance goals and the Company's achievement of certain of the corporate goals as discussed above, and a potential maximum target of \$541,250 is based on achieving performance goals approved by the Chairman of the Compensation Committee. Ms. Cotter's aggregate annual bosus opportunity can range from \$0 to \$541,250. Mr. Devasia Ghose, Chief Financial Officer, has a potential larged bosus opportunity or a range from \$0 to \$200,000 (the maximum potential target, which is based on achievement of his performance goals and the Company's achievement of corporate goals and service and the company's achievement of potential target based upon the company's achievement of potential target in additional performance goals are met by Mr. Ghose's aggregate annual bosus opportunity can range from \$0 to \$200,000 (the maximum potential target if additional performance goals are met by Mr. Smelling's aggregate annual bosus opportunity of 30% of base gave, or 168,000 at target, which is based on achievement of his performance goals, the Company's achievement of corporate goals are met by Mr. Smelling's aggregate annual bosus opportunity of 30% of base gave, or 18,000 at target, which is based on achievement of his performance goals, the Company's achievement of corporate goals are met by Mr. Smelling. Mr. Waynes Smith, Managing Director, ANZ, has a target bosus opportunity of 30% of base gave, or \$112,500 at target, which is based on achievement of his performance goals, the Company's achievement of corporate goals are met by Mr. Smelling. Mr. Smelling's aggregate annual bosus opportunity of 30% of base gave, or \$112,500 at targe

The positions of other management team members have target bonus opportunities ranging from 20% to 30% of Base Salary based on achievement of certain goals. The highest level of achievement a participant may receive is the maximum potential target which is set at 150% of the participant's target bonus amount.

## Long-Term Incentives

Long-Term incentives will utilize the equity-based plan under the Company's 2010 Incentive Stock Plan, as amended (the "2010 Plan"). For 2016, executive and management team participants will receive awards in the following forms: 50% time-based restricted stock units and 50% non-statutory stock options. The grants of restricted stock units and options will vest ratably over a four (4) year period with 1/4 °vesting on each anniversary date of the grant date.

On March 10, 2016 the following grants were made:

Name	Tide	Dollar Amount of Restricted Stock Units (1)	Dollar Amount of Non- Statutory Stock Options (1)
Ellen Cotter	President and Chief Executive Officer	\$150,000	\$150,000
Devasis Ghose (2)	Chief Financial Officer	0	0
Andrzej Matyczynski	EVP Global Operations	37,500	37,500
Robert F. Smerling	President, US Cinemas	50,000	50,000
Wayne Smith	Managing Director, ANZ	27,000	27,000

- (1) The number of shares of stock to be issued will be calculated using the Black Scholes pricing model as of the date of grant of the award.
- (2) Mr. Devasis Ghose was awarded 100,000 non-statutory stock options vesting over a 4 year period on Mr. Ghose's commencement of employment on May 11, 2015.
- All long-term incentive awards will be subject to other terms and conditions set forth in the 2010 Plan and award grant.

## TERMIDATION AGREEMENT AND RELEASE

This Tendentian Agreement and Release (this "Agreement") is made as at Mouth 1-0, 2015, by and between Liberty Throsers . LLC ("UNLC") and ONL . LLC ("UNLC") food on the internal control of the Throser Monogeneous Agreement Agreement Agreement Agreement Agreement (a t defined interior).

WHEVELA II. reference is notice to the correla Theorie Management Agromment (the "Theorie Management Agromment"), deced effective as of Jensory 1, 2002, by and between Library Theories, field. (LTT") and Offi.

WHEREAS, LTI was interpretely merged into LTLLC.

WHEREA II. purment to the Theoret Mongoment Agreement, Old provides Management Services to theorets wanted by LTLUI and its Operating Subsidiories.

WHEREAS, LYELC and OBI desire to terminote the Thomas Monogeneous Agreement on of Nismin 1 0, 2016 ("Terminotion Dun").

WHEREAS . LITLE miknowledges and agrees that Offt has field and so disfectivity performed its calligations under sed pursuant to that Thesian Management Agreement.

WHEREAS, OUT referencingers and agrees that references that the Final Engagement (defined hereafter) LTLA C has believed a satisfactorily performed its obligations under and personant in the Theater Assungation Agreement

NOW, THEREFORE, to occasion result in the motion primities sent traverages contained berein. The parties bereto agree as follows:

- 1. Effective to 16 the Transcontine Date. the Thereor Management Agreement is brown, terransposed. On and other the Transcontine Energy. LTLLC shall been no forther obligations to OBI under the Thereor Management Agreement, inter these is indemnify obligations as faith in Section 5.1 of the Thereor Management Agreement, and OBI shall been to insufer riviligations. In LTLLC under the Thereor Management Agreement, and of all shall been to insufer riviligations. In LTLLC under the Thereor Management Agreement Agreement Agreement to There is the Agreement of the There is the There
- 2. CBS waives to right to receive incentive feet parameter to frection 4.1(d) of the Theater Management Agreement behaved after the Tennitreton Date.
- 2. Other thus payments of the Frank Phaneus by LTLL' to Off one computes provided in Section 1, above, with respect to the survival of exitain indexently obligations, OBI these berefer renoveably searced incoming, horself, in the part of the Frank Phaneus by LTLL' and its parent 1, abovingly, failly, finally and completely furner releases and discingues LTLL' and its parent 2, abovinging, division, surfaces, constant, officer, officer, apprecia, constant, officer, officer, officer, apprecia, constant, officer, officer, officer, apprecia, constant, officer, officer

thing, set are follows to not inhomover occurring at any time on at principal the date of this Agreement that related to the Theodor Management Agreement, OBI invovable 1 y coverance that it will are, discretly at indirectly at information the date of this Agreement and discharged seasons to the Societies 1. As a number of childrenian that release does not release any children this discretization and the release above not release any children that discretization that disagreed control to the Management Agreement and the control of the Societies 1. As a number of childrenian that release does not release any children that discretization that Management Agreement and the control of the societies of the societies of the societies of the Management Agreement and the control of the Management Agreement and the societies of the Management and the Management and the Societies of the Management Agreement and the Societies of the Managem

4.T.T.L.C date bestely increased by the control of the control of

5 OBS and LTLLC vision-rings and agree that the Theater Management Agreement extends to their respective obligations to each vision respect to the subject trades the Theater Management Agreement — and LTLLC but made will payment — and it is their respective obligations to each vision respective of the firm the First F. Payment.

6. This Agreements tail to governed by and a received in inconductive with the large of the State of Nevada without giving effect to the condition in town principles thereof.

7. This a Agreements may be ensounded in multiple concentrative, each of which should be decimed an engaged decreased, that all is which logisther a half ercus blice on each the a were hastranism.

 $f(N) = W(T(N)) \otimes S(W(H) \otimes S(G)), \quad \text{the parties a scalars that Termination Agreement with the date first and forth when he$ 

OBEM ANAGEMENT, ELC

ny: <u>(1811-172-)</u> Note: Horner Court

Name Magazet C Tabi Massan

LOBERTY THEATERS, LLC

\$1.



Systems Continues stratume Foots

National Australia Bank Limited Reading Entertainment Australia Pty Ltd Each Guarantor



## Contents

- 1 Definitions 1
- 2 Consideration 1
- 3 Conditions precedent 2
- 4 Variation of Facility Agreement 2
- 5 Acknowledgments 2
- 6 Warranties and representations 2

  - 6.1 General 2 6.2 Survival of warranties 3
- 7 General 3

  - 7.1 Construction 3
    7.2 Costs and Expenses 4
    7.3 Counterparts 4
    7.4 Entire understanding 4
    7.5 Governing law and jurisdiction 5

### Schedule 1 - Guarantors 6

Execution 8

Annexure A - Restated Document 17

Corre	Chambare	Westgarth

Date:

### **Parties**

National Australia Bank Limited ABN 12 004 044 937 of Pier 3 Level 4, 800 Bourke Street, Docklands, Victoria 3008 ( Bank )

Reading Entertainment Australia Pty Ltd ACN 070 893 908 of 98 York Street, South Melbourne, Victoria 3205 ( Borrower)

Each person listed in schedule 1 (each a Guarantor)

### Agreed terms

### Definitions

In this document words and expressions which are defined or given a specific meaning in the Facility Agreement but which are not defined or given a specific meaning in this document have the same meaning as in the Facility Agreement. Otherwise, terms have the following meanings.

Facility Agreement
The Facility Agreement atted 24 June 2011 between the Bank, the Borrower and the Guarantors, as amended, varied or amended and restated from time to time, including on 14 June 2013 and 27 June 2014.

PPSA Personal Property Securities Act (2009) (Cth).

Restated Facility Agreement
The Facility Agreement as varied and restated by this document.

Variation Date The date on which Bank notifies the Borrower that the conditions precedent set out in clause 3 are satisfied.

### Consideration

Each party has entered into this document in consideration of each other party agreeing to make the acknowledgements contained in this document and to amend the Facility Agreement in accordance with this document and acknowledges receipt of that consideration.

### 3 Conditions precedent

The amendments to the Facility Agreement are subject to the conditions precedent that:

- (a) the Bank has received in form and substance satisfactory to the Bank:
  - (i) an original copy of this document, duly executed by the Borrower and each Guarantor,
  - (ii) a non-refundable restructure fee of \$30,000; and
  - (iii) anything which the Bank has reasonably requested that a party provide to it in relation to the Facility Agreement; and
- (b) no Event of Default or Potential Event of Default subsists.

### 4 Variation of Facility Agreement

- (a) On and from the Variation Date the Facility Agreement is varied and restated in the form of annexure A
- (b) The Borrower and each Guarantor agree to be bound by the Restated Facility Agreement on and from the Variation Date.

### 5 Acknowledgments

The Borrower and each Guarantor.

- (a) acknowledge that nothing in this document or any related financing change statement under the PPS Act releases, terminates or otherwise affects any liabilities of the Borrower to the Bank, or affects its liability under any Transaction Document;
- (b) acknowledge that the Bank has agreed to execute this document at the request of the Borrower and each Guarantor;
- (c) agree to the variation to the Facility Agreement and agrees that each Collateral Security extends to and secures the Borrower's obligations to the Bank under the Transaction Documents, including the Restated Facility Agreement; and
- (d) acknowledge that their obligations (as borrower, guarantor, indemnifier, or otherwise) and the Bank's rights are not affected by anything which might abrogate, prejudice or limit them or the effectiveness of this document, including the failure by any person named as an Original Guarantor to become bound by this document.

### 6 Warranties and representations

#### 6.1 Genera

The Borrower and each Guarantor warrant and represent to the Bank that:

(a) at the time of execution, and at the Variation Date:

- (i) it has capacity unconditionally to execute, deliver and comply with its obligations under this document and the Transaction Documents to which it is a party (including the Restated Facility Agreement);
- (ii) it has taken all necessary action to authorise the unconditional execution and delivery of, and the compliance with, its obligations under this document and the Transaction Documents (including the Restated Facility Agreement):
- (iii) this document and the Transaction Documents to which it is a party (including the Restated Facility Agreement) are its valid and legally binding obligations and are enforceable against it by each other party in accordance with the terms of this document and the Transaction Documents to which it is a party (including the Restated Facility Agreement), subject to principles of equity and rules affecting creditors' rights generally; and
- (iv) its unconditional execution and delivery of this document and compliance with its obligations under this document and the Transaction Documents (including the Restated Facility Agreement) do not contravene:
  - (A) any law or directive from a government entity;
  - (B) its constituent documents;
  - (C) any agreement or instrument to which it is a party; or
  - (D) any obligation of it to any other person.

### 6.2 Survival of warranties

The warranties and representations in clause 6 survive the execution of this document and the variation and restatement of the Facility Agreement.

### 7 General

#### 7.1 Construction

Unless expressed to the contrary, in this document:

- (a) words in the singular include the plural and vice versa;
- (b) any gender includes the other genders;
- (c) if a word or phrase is defined its other grammatical forms have corresponding meanings;
- (d) 'includes' means includes without limitation;
- (e) no rule of construction will apply to a clause to the disadvantage of a party merely because that party put forward the clause or would otherwise benefit from it;
- (f) a reference to:

- (i) a person includes a partnership, joint venture, unincorporated association, corporation and a government or statutory body or authority;
- a person includes the person's legal personal representatives, successors, assigns and persons substituted by novation;
- (iii) any legislation includes subordinate legislation under it and includes that legislation and subordinate legislation as modified or replaced;
- (iv) an obligation includes a warranty or representation and a reference to a failure to comply with an obligation includes a breach of warranty or representation;
- (v) a right includes a benefit, remedy, discretion or power;
- (vi) time is to local time in Melbourne;
- (vii) '\$' or 'dollars' is a reference to Australian currency;
- (viii) this or any other document includes the document as novated, varied or replaced and despite any change in the identity of the parties;
- (ix) writing includes any mode of representing or reproducing words in tangible and permanently visible form, and includes fax transmissions;
- (x) this document includes all schedules and annexures to it; and
- (xi) a clause, schedule or annexure is a reference to a clause, schedule or annexure, as the case may be, of this document; and
- (g) where time is to be calculated by reference to a day or event, that day or the day of that event is excluded.

### 7.2 Costs and Expenses

The Borrower must on demand pay and if paid by the Bank reimburse to the Bank:

- (a) the Bank's reasonable costs and expenses (including legal costs and expenses on a full indemnity basis) relating to the negotiation, preparation, execution, stamping and registration of this document; and
- (b) any duties and registration or other fees (including fines and penalties relating to such duties and fees) which are payable or are assessed by a relevant government body or other person to be payable in relation to this document or any transaction contemplated by it.

#### 7.3 Counterparts

This document may consist of a number of counterparts and, if so, the counterparts taken together constitute one document.

### 7.4 Entire understanding

(a) This document contains the entire understanding between the parties as to the subject matter of this document.

**JA3107** 

- (b) All previous negotiations, understandings, representations, warranties, memoranda or commitments concerning the subject matter of this document are merged in and superseded by this document and are of no effect. No party is liable to any other party in respect of those matters.
- (c) No oral explanation or information provided by any party to another.
  - (i) affects the meaning or interpretation of this document; or
  - (ii) constitutes any collateral agreement, warranty or understanding between any of the parties.

### 7.5 Governing law and jurisdiction

- (a) This document is governed by and is to be construed in accordance with the laws applicable in Victoria.
- (b) Each party irrevocably and unconditionally submits to the non-exclusive jurisdiction of the courts exercising jurisdiction in Victoria and any courts which have jurisdiction to hear appeals from any of those courts and waives any right to object to any proceedings being brought in those courts.

# Schedule 1

### Guarantors

Name	ACN	Particulars for delivery of notices
Reading Entertainment Australia Pty Ltd	070 893 908	Address: 98 York Street, South Melbourne VIC 3205 Australia
		Fax: 03 9685 0999
		Attention: Managing Director, Wayne Smith
		AND TO:
		Reading International Inc.
		Address: 6100 Center Drive, Suite 900 Los Angeles California 90045
		United States of America
		Fax: +1 213 235 2229
		Attention: Chief Financial Officer, Dev Ghose
	ļ ģ	
Australia Country Cinemas Pty Ltd	076 276 349	Same as for Borrower
Australian Equipment Supply Pty Ltd	122 571 420	Same as for Borrower
Burwood Developments Pty Ltd	105 384 905	Same as for Borrower
Epping Cinemas Pty Ltd	073 997 172	Same as for Borrower
Hotel Newmarket Pty Ltd	094 367 969	Same as for Borrower
Newmarket Properties Pty Ltd	105 386 409	Same as for Borrower
Newmarket Properties No. 2 Pty Ltd	109 038 806	Same as for Borrower
Newmarket Properties #3 Pty Ltd	126 697 505	Same as for Borrower
Reading Auburn Pty Ltd	126 697 470	Same as for Borrower
Reading Australia Leasing (E&R) Pty Ltd	107 939 211	Same as for Borrower
Reading Belmont Pty Ltd	126 697 498	Same as for Borrower
Reading Bundaberg 2012 Pty Ltd	122 406 320	Same as for Borrower

Name	ACN	Particulars for delivery of notices
(formerly Reading Moonee Ponds Pty Ltd)		
Reading Charlestown Pty Ltd	123 938 483	Same as for Borrower
Reading Cinemas Pty Ltd	073 808 643	Same as for Borrower
Reading Cinemas Management Pty Ltd	122 406 311	Same as for Borrower
Reading Colac Pty Ltd	108 861 061	Same as for Borrower
Reading Dandenong Pty Ltd	129 018 739	Same as for Borrower
Reading Elizabeth Pty Ltd	114 582 099	Same as for Borrower
Reading Exhibition Pty Ltd	103 529 782	Same as for Borrower
Reading Licences Pty Ltd	089 544 605	Same as for Borrower
Reading Maitland Pty Ltd	126 697 461	Same as for Borrower
Reading Melton Pty Ltd	109 074 517	Same as for Borrower
Reading Properties Pty Ltd	071 195 429	Same as for Borrower
Reading Properties Indooroopilly Pty Ltd as trustee for The Landplan Property Partners Discretionary Trust	121 284 884	Same as for Borrower
Reading Properties Taringa Pty Ltd as trustee for the Reading Property Partners No. 1 Discretionary Trust	128 819 483	Same as for Borrower
Reading Property Holdings Pty Ltd	126 289 772	Same as for Borrower
Reading Rouse Hill Pty Ltd	123 245 885	Same as for Borrower
Reading Sunbury Pty Limited	109 074 571	Same as for Borrower
Rhodes Peninsula Cinema Pty Ltd	120 827 812	Same as for Borrower
Westlakes Cinema Pty Ltd	108 531 308	Same as for Borrower
A.C.N. 143 633 096 Pty Ltd	143 633 096	Same as for Borrower

C	ors Chambers Westgarth
1	Annexure A
-	Restated Facility Agreement
*****	
*********	



System, Metrocore Hallowera Fresh

Annexure A - Restated Facility Agreement
National Australia Bank Limited
Reading Entertainment Australia Group

# Corporate Markets Loan & Bank Guarantee Facility Agreement

**JA3112** 

### Contents

# Interpretation 1

- Interpretation 1

  1.1 Definitions 1

  2 Construction 18

  1.3 Headings 19

  1.4 Corporations Act, GST and Accounting Standards 19

  1.5 Subsisting Events of Default and Potential Events of Default 19

  1.6 Not used 19

  1.7 Inconsistency 19

### 2 Consideration 19

### 3 Conditions precedent 20

- Not used 20
   Conditions precedent to Advances and Drawings 20
   Not used 20

### 4 Facility 20

- Facility 20
  4.1 Nature 20
  4.2 Purpose 20
  4.3 Advances and Drawings 20
  4.4 Funding Notices 21
  4.5 Not used 22
  4.6 Not used 22
  4.7 Not used 22
  4.8 Bank Guarrantee Facilities 22
  4.9 Cancellation 22
  4.10 Market disruption 22
  4.11 Alternative basis of interest or funding 23
  4.12 Pricing Review Events 23
  4.13 Consequences of a Pricing Review 23
  Payments 24
  Payments 24

## Payments 24

- | 1. Not used 24 | Voluntary prepayments 24 | Indemnity in respect of Bank Guarantees 24 | Mandatory prepayments 26 | Repayment 26 | Not used 27 | Not used 28 | Not used 28 | Not used 29 | Not used

#### Interest and fees 27

- 6.1 Pricing Periods 27
  6.2 Payment and rate 27
  6.3 Computation of interest 28
  6.4 Capitalisation of interest 28
  6.5 Merger 28

13.1 Taxable supply 13.2 Adjustment events 13.3 Payments

7	Payn	Payments 29			
	7.1 7.2 7.3	Place, manner and time of payment 29 Gross-up 29 Appropriation 29			
8		esentations and warranties 29			
9	Gene	eral obligations 33			
	9.1	Fees 33			
	9.2	Records 34			
	9.3	Financial Statements and other financial information 34			
	9.4	Other information 35			
	9.5	Other financial undertakings 36			
	9.6	Insurance 39			
	9.7 Financial ratios 39				
	9.8	Environment 40			
	9.9	No default 42			
	9.10	Obligations of Trustees 42			
	9.11	Release for Permitted Disposals 43			
10	Events of Default 43				
	10.1	Nature 43			
	10.2	Effect of Event of Default 46			
	10.3	Cash Cover Account regarding Bank Guarantees 47			
	10.4	Review Events 48			
	10.5	Reviews 48			
11_	Costs and expenses 48				
	11.1	Interpretation	48		
	11.2	Nature	48		
	11.3	Remuneration	49		
12	Indemnities 49				
	12.1	Nature 49			
	12.2	Representatives 50			
	12.3	Currency deficiency 50			
	12.4	Independence and survival 50			
		Accounting for transactions 50			
	12.6	Liability for Regulatory Events 51			
13_	Good	is and Services Tax 51			
	40.4	Toughle comb.	51		

52 52

Corrs Chambers Westgarth

- 18.19 Legal advice 64 18.20 Further assurances 64 18.21 Exclusion of certain provisions 65 18.22 Notice of changes 65

### 19 Notices 65

- 19 Notices 65
  19.1 General 65
  19.2 How to give a communication 65
  19.3 Particulars for delivery of notices 66
  19.4 Communications by post 66
  19.5 Communications by fax 66
  19.6 After hours communications 66
  19.7 Process service 66
  Schedule 1 Transaction Parties 67

Schedule 2 - Facilities 70

Schedule 3 – Collateral Security 71

Schedule 4 - Not used 74

Schedule 5 - Conditions Precedent 75

Schedule 6 - Verification Certificate 79

Schedule 7 - Funding Notice 81

Schedule 8 – Guarantor Accession Deed 82

Schedule 9 - Compliance Certificate 84

Schedule 10 – Interim Compliance Certificate 86

	200	W10301		170.46
Corrs	Cham	oers.	West	carth

#### Date

15th December 2015

### **Parties**

National Australia Bank Limited ABN 12 004 044 937 of Pier 3 Level 4, 800 Bourke Street, Docklands, Victoria 3008 ( Bank )
Reading Entertainment Australia Pty Ltd ACN 070 893 908 of 98 York Street, South Melbourne, Victoria 3205 ( Borrower)
Each person listed in schedule 1 (each an Original Guarantor )

### Agreed terms

- 1 Interpretation
- 1.1 Definitions

In this document:

Accounting Standards means accounting principles and practices consistently applied which are generally accepted in Australia and are consistent with any applicable legislation in each case as in effect on the date of this document, including instruments in force under section 334 of the Corporations Act and provisions of such instruments.

Adjusted EBITDA means, for any period, EBITDA adjusted to exclude:

(a) any non-cash impairment for non-current assets included in the consolidated financial statements of the Reading Entertainment Australia Group during the relevant period; and any net foreign exchange amounts (whether realised or unrealised) included in the consolidated financial statements of the Reading Entertainment Australia Group during the relevant period.

and subject to adjustment in respect of any further extraordinary items with the Bank's written consent.

Advance means the principal amount of an advance made under the Corporate Markets Loan Facility or, where appropriate, requested under the Corporate Markets Loan Facility.

Aggregate Amount means, in relation to a Drawing, the aggregate of the Face Values of all Bank Guarantees comprising that Drawing.

Annual Compliance Certificate means, in relation to a Financial Year, a

certificate substantially in the form of schedule 9

Approved Valuer means a company or firm of duly qualified and licensed real estate valuers acceptable to the Bank in all respects and instructed by (or with the approval of) the Bank. Attorney means any attorney appointed under this document and any sub-attorney appointed by an Attorney.

Attorney means any attorney appointed under this document and any sub- attorney appointed by an Attorney.

Authorisation includes any authorisation, consent, licence, permission, approval or exemption from any Government Body. If a Government Body could prohibit anything being done in connection with any matter or otherwise intervene within a specified time after notice has been given to it or any document lodged or filed with it in connection with the matter, the relevant matter will not be taken to have been Authorised until the specified time limit has expired without the Government Body taking any relevant action.

Authorised Representative means, in relation to any party to this document, a person with the right to act as the agent of that party for the purposes of this document. It includes a director or company secretary of that party (if it is a corporation) and, in the case of the Bank, an employee of the Bank whose title contains the word "manager", "director", "associate" or a similar term and a lawyer for the Bank. It also includes a person appointed by a party as an Authorised Representative of that party whose appointment is notified by the appointor to the other party in a notice which contains the specimen signature of the appointee.

Availability Period means in respect of each Facility, the period beginning on the date on which the conditions precedent are satisfied or waived by the Bank in accordance with the Transaction Documents and ending on the Termination Date.

Available Commitment means in respect of a Facility, the Facility Limit less the Outstanding Accommodation relating to that Facility.

Bank Guarantee means each bank guarantee issued (or deemed to have been issued) in accordance with this document.

Bank Guarantee Facility means the Facility described as such in schedule 2 and granted pursuant to clause 4.1(a)(ii).

Bank Guarantee Margin means, in respect of each Bank Guarantee:

- prior to the 'Variation Date' under the Restatement Deed, 2.35% per annum; and
- on and from the first services fee charge date (to be determined in accordance with clause 9.1(e)) following the 'Variation Date' under the Restatement Deed, 1.90% per

Base Rate means, in relation to a Pricing Period:

(a) the rate (expressed as a percentage yield per annum to maturity) being the arithmetic average (rounded up to the nearest four decimal places) of the buying rates published at or about 10.15 am on the first Business

Day of the Pricing Period on the Reuters Screen under the heading "BBSY" for Bills with a tenor as nearly as possible equal to that Pricing Period; or

- (b) if:
  - (i) the rate is not displayed for a term equivalent to that period; or
  - (ii) the basis of the calculation of the rate is changed after the date of this document so that in the opinion of the Bank it ceases to reflect the cost of providing the Facility,

the Base Rate will be the rate per centum per annum determined by the Bank to be the average of the buying rates quoted to the Bank by at least three Reference Banks at or about that time on that date. The buying rates must be for bills of exchange accepted by a leading Australian bank and which have a term equivalent to the period. If there are no buying rates, the rate will be determined by the Bank having regard to indexes or other bases which the Bank determines to be as near as practicable to the indexes and bases used to determine the rate referred to in paragraph (a).

Beneficiary means in relation to a Bank Guarantee, the person who from time to time is entitled to make a claim for payment under that Bank Guarantee against the Bank.

Bill means a bill of exchange as defined in the Bills of Exchange Act 1909 (but does not include a cheque). It includes a document which, when signed by the persons named as drawer and acceptor in the relevant document, will become such a bill of exchange.

Break Costs means, in relation to any financial accommodation provided or to be provided by the Bank under a Facility, any liability or costs incurred by the Bank by reason of:

- (a) liquidating or re-deploying deposits or other funds acquired or contracted for by or on account of the Borrower or the Bank;
- (b) terminating or reversing any agreement or arrangement (including by entering into new agreements or arrangements to close out or net off existing agreements or arrangements) entered into by or on account of the Borrower or the Bank with a counterparty or an internal department of the Bank responsible for such agreements or arrangements to hedge, fix, swap or limit its effective cost of funding; or
- (c) any loss of any margins in relation to future lending or loss of any fees.

### Burwood Property means:

- (a) the land and improvements known as 78 Middleborough Road, Old Burwood Road, Burwood, Victoria; and
- (b) all present or future agreements, documents or rights relating to the ownership, occupation, use, sale, transfer or disposal of all or any part of the land and improvements described in paragraph (a).

Business Day means a day which is not a Saturday, Sunday or bank or public holiday in Melbourne.

Cash Cover Rate means the rate (expressed as a rate per centum per annum) determined by the Bank (in good faith) to be the interest rate which it would pay on deposits at call for an amount similar to the amount at which the relevant deposit is made.

Calculation Date means 31 March, 30 June, 30 September and 31 December in each year.

Calculation Period means each period of twelve months ending on a Calculation Date.

Change of Control means there is a change (from that prevailing at the date of this document) in the persons who control any of the following in respect of a Transaction Party.

- (a) more than 50% of the votes eligible to be cast in the election of directors or any similar matter; or
- (b) the right to appoint or remove directors (or members of a governing body having functions similar to a board of directors) representing more than 50% of the votes exercisable by the directors (or persons have similar functions); or
- (c) an interest of more than 50% in any category of the profits, distributions or net liquidation proceeds.

### Collateral Security means:

- (a) any Guarantee by which any person Guarantees the Borrower's compliance with its obligations under any of the Transaction Documents;
- any Security which secures the payment of money owing (actually or contingently) from time to time by:
  - any Transaction Party in relation to any of the Transaction Documents; or
  - any person in relation to a Guarantee of any Transaction Party's compliance with its obligations under any of the Transaction Documents; and
- (c) without limiting the generality of paragraphs (a) and (b) each thing listed in schedule 3.

Contaminant means a noxious, harmful or hazardous condition (including an odour, temperature, sound, vibration or radiation) or substance the presence or use of which (having regard, without limitation, to the nature and quantity of the substance and other substances with which it is stored or used) does or may result in the breach of an Environmental Law or the issuing of an order or direction under an Environmental Law.

Corporate Markets Loan Facility means the Facility described as such in schedule 2 and granted pursuant to clause 4.1(a)(i).

**JA3120** 

Corporations Act means the Corporations Act 2001 (Cth).

Current Bank Guarantee means a Bank Guarantee which has not Matured or Expired.

Daily Interest Rate means, for any day, the Interest Rate on that day divided by 365.

Disposal means a sale, lease, transfer or other disposal by any Transaction Party of any interest in:

- (a) any share or stock (whether or not ordinary or preference and whether or not redeemable) or any other instrument convertible or exchangeable into or entitling a person to acquire or subscribe for any share or stock.
- (b) the whole or any part of a business, business unit or line of business; or
- (c) any other asset under a particular transaction or related transactions not in the ordinary course of business of the Reading Entertainment Australia Group taken as a whole. Distribution means:
- (a) in relation to any share capital of a Transaction Party, any dividend, charge, interest, fee, payment or other distribution (whether in cash or in kind) or redemption, repurchase, defeasance, retirement or redemption
- any interest, any redemption or early redemption of any amount of principal or any other payment in respect of any shareholder loan or other subordinated loans made to any Transaction Party; or (b)
- any loan or other financial accommodation made available by a Transaction Party to a person other than another Transaction Party. Drawing means each Bank Guarantee issued or to be issued in accordance with this document under the same Funding Notice.

EBIT means, in relation to any period and without double counting, operating profit (loss) of the Reading Entertainment Australia Group (on a consolidated basis) from ordinary operations before interest, income tax and minority interests, but after deduction of depreciation and amortisation for that period, as determined in accordance with Accounting Standards.

EBITDA means, in relation to any period, EBIT for the Reading Entertainment Australia Group for that period, plus depreciation and amortisation as determined in accordance with Accounting Standards.

Encumbrance means any interest in or right over property and anything which would at any time prevent, restrict or delay the registration of any interest in or dealing with property. It includes a Security Interest.

Environmental Assessment Report means a report in relation to compliance with Environmental Law of the Land and any activities carried out on the Land.

Environmental Law means any legislation, regulations or related codes, standards or policies which relate to environmental and planning matters, including matters concerning land use, development, building works, pollution,

contamination, waste, toxic and hazardous substances, disposal of waste or other substances, human health, conservation of natural or cultural resources, heritage and resource

Environmental Liability means any liability, obligation, expense, penalty or fine arising out of a breach of Environmental Law which could be imposed on any Transaction Party or the Bank in respect of the Land as a result of activities carried on during the ownership, occupation or control of the Land by that Transaction Party, the Bank, any predecessor in title or any previous occupier or controller of the Land.

Event of Default means any event or circumstance described in clause 10.1. Excluded Financial Indebtedness means Financial Indebtedness of the kind referred to in paragraph (a), (c) or (d) of the definition of Permitted Financial Indebtedness.

#### Excluded Property means:

- (a) the Burwood Property;
- (b) the present of future interest of Reading Cinemas Pty Ltd in the joint venture with Champion Pictures Pty Ltd relating to the Elsternwick cinema business or the assets the subject of the joint venture or the relevant joint venture agreement:

  (c) the present or future interest of Reading Exhibition Pty Ltd in the Garden City Cinema joint venture with Village Roadshow Exhibition and Birch Carroll & Coyle or the assets the subject of the joint venture or the relevant joint venture agreement; and
- (d) the present or future interest of Epping Cinemas Pty Ltd in the lease granted by Bevendale Pty Ltd or the property the subject of the lease to the extent that the existence of a charge over that interest or property would cause a breach of the that lease.

Expired means, in relation to a Bank Guarantee, that its Expiry Date has passed whether or not a claim has been made under it by the Beneficiary.

Expiry Date means, in relation to a Bank Guarantee, the date specified in that Bank Guarantee as the latest date by which the Beneficiary may make a claim under it.

Face Value means, in relation to a Bank Guarantee:

- (a) subject to paragraph (b), the amount specified in that Bank Guarantee as the aggregate maximum amount which the Beneficiary may claim under it; or
- (b) if the Beneficiary makes a claim, then between when the Beneficiary makes the first of those claims and the first to occur of the Bank Guarantee Maturing or Expiring, the Face Value of the Bank Guarantee will be the difference between its original face value and the aggregate of all valid claims made under it.

Facility means each of the facilities listed in schedule 2 (and each Facility

may be referred to by the Facility Name listed in schedule 2).

Facility Limit means, in respect of each Facility, the relevant Facility Limit set out in schedule 2, as reduced under this document.

Financial Close means the initial Funding Date.

Financial Indebtedness means any indebtedness or other liability (present or future, actual or contingent) relating to any financial accommodation including indebtedness or other liability:

- (a) for money borrowed or raised;
- (b) relating to the sale or negotiation of any negotiable instrument;
- (c) as lessee under any finance lease, as hirer under any hire purchase agreement or as purchaser under any title retention agreement;
- relating to any preference share or unit categorised as debt under Accounting Standards;
- (e) under any commodity, currency or interest rate swap agreement, forward exchange rate agreement or futures contract (as defined in any statute);
- (f) under any Guarantee relating to any financial accommodation; or
- (g) for any deferred purchase price (other than in the nature of warranty retention amounts) for any asset or service.

Financial Statements means a balance sheet, an income statement, a statement of changes in equity, a cash flow statement, notes comprising a summary of significant accounting policies and other explanatory note; and any directors' declarations, directors' reports and auditor's reports attached to, intended to be read with or required by the Corporations Act to accompany, all or any of those documents.

Financial Year means a period of 12 months ending on 31 December.

Fixed Charges Cover Ratio means, at any date, the ratio of:

- (a) the aggregate amount of:
  - (i) Adjusted EBITDA in respect of the 12 month period ending on that date; and
  - (ii) Total Lease Payments in respect of the 12 month period ending on that date,

to

- (b) the aggregate amount of:
  - (i) Gross Interest Expense paid or payable by the Reading Entertainment Australia Group (whether payable in respect of the Facilities or otherwise) in respect of the 12 month period ending on that date; and
  - (ii) Total Lease Payments in respect of the 12 month period ending on that date.

**JA3123** 

Freehold Property means each freehold property owned by a Transaction Party that is the subject of a real property mortgage referred to in of schedule 3. Funding Date means a date on which:

- (a) an Advance is, or is proposed to be, made; or
- a Bank Guarantee is, or is proposed to be, issued, under this document.

Funding Notice means a notice in accordance with clause 4.4.

Government Body means any person or body exercising an executive, legislative, judicial or other governmental function. It includes any public authority constituted under a law of any country or political sub-division of any country. It also includes any person deriving a power directly or indirectly from any other person or body referred to in this definition.

Gross Interest Expense means, in relation to any period, the aggregate of all interest and amounts in the nature of interest (including commissions, discount fees, acceptance fees, facility fees, the interest element of a finance lease and fees or charges) payable in connection with any Financial Indebtedness of the Reading Entertainment Australia Group (other than Excluded Financial Indebtedness) for that period on a consolidated basis, whether accrued, paid, payable or expensed (including interest expense under each of the Facilities).

#### Guarantee means:

- a guarantee, indemnity, undertaking, letter of credit, Security, acceptance or endorsement of a negotiable instrument or other obligation (actual or contingent) given by any person to secure compliance with an obligation by another person;
- (b) an obligation (actual or contingent) of a person to ensure the solvency of another person or the ability of another person to comply with an obligation, including by the advance of money or the acquisition for valuable consideration of property or services; and
- an option under which a person is obliged on the exercise of the option to buy:
  - (i) any debt or liability owed by another person; or
  - (ii) any property which is subject to a Security Interest.

### Guaranteed Money means all money:

- which now or in the future is owing (actually or contingently) by a Transaction Party to the Bank under or in relation to any of the Transaction Documents;
- (b) which having now or in the future become owing (actually or contingently) by a Transaction Party to the Bank under or in relation to any of the Transaction Documents, ceases to be owing by reason of any law relating to insolvency and remains unpaid by the Transaction Party

and unreleased by the Bank; or

- (c) that now or in the future may become owing (actually or contingently) by a Transaction Party to the Bank under or in relation to any of the Transaction Documents,
- for any reason, whether such money is payable:
- (d) by a Transaction Party alone or jointly or severally with any other person;
- (e) by a Transaction Party in its own right or in any capacity;
- (f) to the Bank in its own right or in any capacity; and
- (g) by a Transaction Party as liquidated or unliquidated damages caused or contributed to by any breach by the Transaction Party of any obligation owed by the Transaction Party (or any other Transaction Party) to the Bank under or in relation to any of the Transaction Documents,

and if any Transaction Document or any obligation of a Transaction Party to the Bank under or in relation to any of the Transaction Documents is void, voidable or otherwise unenforceable by the Bank in accordance with its terms, it includes all money which would have been within this definition if that Transaction Document or obligation was not void, voidable or otherwise unenforceable.

Guarantor means the Original Guarantors and each person that becomes a guarantor under clause 16. If there are more than one, Guarantor means each of them individually and every two or more of them jointly.

Guarantor Accession Deed means a deed substantially in the form of schedule 8 .

Half means each six month period ending on 30 June and 31 December in each year.

Hedging Transaction means a contract, agreement or arrangement (other than in respect of the price of electricity, gas, oil, foreign exchange or any other non-interest rate derivative contract) which is a futures contract or an interest rate hedge, swap, option, swaption, forward rate agreement or any other contract, agreement or arrangement similar to or having in respect of its subject matter a similar effect to any of the preceding.

Indemnity Amount means, in relation to a Bank Guarantee, the amount or, as the case may be, the aggregate of the amounts payable by the Borrower in relation to a Bank Guarantee in accordance with clause 5.3

#### Insolvency means:

- (a) in relation to a corporation, its winding up or dissolution or its administration, provisional liquidation or any administration having a similar effect;
- (b) in relation to an individual, his or her bankruptcy; and
- (c) in relation to a person, any arrangement (including a scheme of arrangement or deed of company arrangement), composition or

compromise with, or assignment for the benefit of, all or any class of that person's creditors or members or a moratorium involving any of them

#### Insolvency Event means any of the following:

- (a) a person is or states that the person is unable to pay from the person's own money all the person's debts as and when they become due and payable;
- (b) a person is taken or must be presumed to be insolvent or unable to pay the person's debts under any applicable legislation;
- (c) an order is made for the winding up or dissolution or an effective resolution is passed for the winding up or dissolution of a corporation;
- (d) an administrator, provisional liquidator, liquidator or person having a similar or analogous function under the laws of any relevant jurisdiction is appointed in relation to a corporation or an effective resolution is passed to appoint any such person and the action is not stayed, withdrawn or dismissed within 10 Business Days;
- (e) a controller is appointed in relation to any property of a corporation;
- (f) a corporation is deregistered under the Corporations Act or notice of its proposed deregistration is given to the corporation;
- a distress, attachment or execution is levied or becomes enforceable against any property of a person;
- (h) a person enters into or takes any action to enter into an arrangement (including a scheme of arrangement or deed of Borrower arrangement), composition or compromise with, or assignment for the benefit of, all or any class of the person's creditors or members or a moratorium involving any of them;
- (i) a petition for the making of a sequestration order against the estate of a person is presented and the petition is not stayed, withdrawn or dismissed within seven days or a person presents a petition against himself or herself;
- (j) a person presents a declaration of intention under section 54A of the
- Bankruptcy Act 1966; or
- (k) anything analogous to or of a similar effect to anything described above under the law of any relevant jurisdiction occurs in relation to a person.

Insurance means insurance which a Transaction Party is obliged to take out or maintain under a Transaction Document.

Interest Rate means, in relation to a Pricing Period for an Advance until it becomes due and owing, an interest rate equal to the aggregate of the Base Rate for that Pricing Period and the Margin.

Interim Compliance Certificate means a certificate in substantially the form set out in schedule 10.

Land means any land owned or occupied by a Transaction Party that forms

part of the Secured Property.

Leasehold Properties means each leasehold property leased by a Transaction Party that is the subject of a mortgage of lease referred to in schedule 3 (including the mortgage of lease described at item 11 of schedule 3).

Leverage Ratio means, as at any date, the ratio of:

- (a) Total Gross Debt outstanding on that date; to
- Adjusted EBITDA in respect of the 12 month period ending on that date. For the purposes of calculating Leverage Ratio on any date occurring before the first anniversary of Financial Close, Leverage Ratio will be based on a pro-forma EBITDA for the 12 month period to that date.

Loan to Value Ratio at any date means the ratio (expressed as a percentage) of:

- (a) the aggregate of the Total Gross Debt outstanding on that date and any Outstanding Accommodation in relation a Current Bank Guarantee as at that date; to
- (b) the market value of the Freehold Properties and Leasehold Properties included in the Secured Property as noted in the most recent Valuation provided to the Bank pursuant to this document and accepted by the Bank.

Management Fees means management and consulting fees payable to Reading International Inc each Financial Year.

Margin means 0.95% per annum.

Material Adverse Effec t means a material adverse effect on:

- (a) the business, operation, property, condition (financial or otherwise) of a Transaction Party or the Reading Entertainment Australia Group taken as a whole;
- the ability of a Transaction Party to perform its obligations under the Transaction Documents; or
- the validity or enforceability of the whole or any material part of any Transaction Document or any rights or remedies of the Bank under the Transaction Documents.

Matured means, in relation to a Bank Guarantee, that the Beneficiary has made a claim and is not entitled to claim any more under the relevant Bank Guarantee. Month means a calendar month.

Outstanding Accommodation means at any time, the aggregate of:

- (a) the aggregate of the unpaid Advances outstanding under the Corporate Markets Loan Facility;
- (b) the Face Values of all Current Bank Guarantees and all Indemnity

**JA3127** 

Amounts in relation to each Bank Guarantee which are due and payable; and

- (c) for the purposes of clauses 5.5, 10 and 18.14 only and for no other purposes, any other amounts which the Borrower owes to the Bank or which the Borrower may owe to the Bank under or in connection with the Facilities and includes:
  - (i) any other amounts which the Borrower owes to the Bank or which the Borrower may owe to the Bank under or in connection with any Hedging Transaction; and
    - ) all interest, costs and fees payable under the Transaction Documents,

whether such amounts are owing actually or contingently and whether such amounts are then due for payment or will or may become due for payment and includes all interest, costs and fees payable under the Transaction Documents.

When used in relation to any Facility, it means the Outstanding Accommodation in relation to Advances or Drawings under that Facility (as applicable).

Overdue Money means money due and payable from time to time under each Transaction Document.

Overdue Rate means at any time, the aggregate of the Interest Rate and a default margin of 4.50% per annum.

Parent Subordination Agreement means the document entitled 'subordination deed' dated on or about the date of this document between the Borrower, Reading International Cinemas LLC and the Bank.

Permitted Disposal means a disposal:

- (a) of assets between the Transaction Parties;
- (b) of the land and improvements known as 70 Station Road, Indooroopilly Queensland and described in certificate of title 11485156, subject to Reading Properties Indooroopilly Pty Ltd seeking a release of the Bank's Security over that property and compliance with clause 5.4(b);
- (c) of the Burwood Property:
- (d) of Reading Cinemas Pty Ltd's interest as lessee in the Elsternwick cinema business, to Champion Pictures Pty Ltd;
- (e) of trading stock or cash made in the ordinary course of business;
- (f) of plant and equipment in exchange for other assets comparable or superior as to type, value and quality;
- (g) of obsolete or redundant assets;
- (h) arising as a result of a Permitted Encumbrance or a Distribution or payment permitted by clause 9.5(f), clause 9.5(j) or clause 5.4(b)(ii)

- (i) of assets that are the subject of a floating charge (or its equivalent) under a Collateral Security, provided the disposal is made in the ordinary course of business; or
- where the aggregate value of the assets disposed of in the 12 month period ending on the date of the relevant disposal (and including the value of the relevant disposal) does not exceed \$500,000. (j)

Permitted Distributions means a distribution of any proceeds of the sale of:

- (a) 40 Hall Street, Moonee Ponds, Victoria; or
- 78 Middleborough Road, Old Burwood Road, Burwood, Victoria, applied towards:
- first, the repayment of parent loans to the Borrower from Reading International Inc (including accrued Management Fees, interest, capitalised interest or principal); and (c)
- (d) after the repayment of all parent loans under paragraph (c), dividends paid to shareholders of the Borrower, provided the Leverage Ratio is less than 3.0 times at the time of the

#### Permitted Encumbrance means:

- an Encumbrance which has been approved by the Bank (including the Security Interests created by any Transaction Document);
- any right of set off or combination arising by operation of law or practice over money deposited with a bank or financial institution in the ordinary course of the business of a Transaction Party;
- (c) an Encumbrance which arises by operation of law in the ordinary course of the business of a Transaction Party provided the debt secured by that Encumbrance is paid when due or contested in good faith by appropriate proceedings;
- (d) every easement, restrictive covenant, caveat or similar restriction over property, right of way, exception, encroachment, reservation, restriction, condition or limitation which arises in the ordinary course of the ordinary business of the relevant Transaction Party and does not either by itself or in the aggregate materially interfere with or impair the operation or use of a property affected thereby, have a Material Adverse Effect or otherwise restrict or prevent the Bank exercising its rights against any Secured Property under the relevant Collateral Security;
- (e) every right reserved to, or vested in, any municipality or governmental or other public authority by the terms of any right, power, franchise, grant, licence or permit to control or regulate any part of the property of a Transaction Party, or to use that property in any manner which does not either by itself or in the aggregate materially interfere with or impair the operation or the use thereof, have a Material Adverse Effect or otherwise restrict or prevent the Bank exercising its rights against any Secured Property under the relevant Collateral Security.

- (f) every Encumbrance incurred or deposits made in the ordinary course of ordinary business to secure the performance of tenders, statutory obligations, surety bonds, bids, leases, government contracts, performance and return of money bonds (provided that such Encumbrances do not restrict or prevent the Bank exercising its rights against any Secured Property under the relevant Collateral Security) or in connection with workers' compensation, unemployment insurance and other types of social security;
- every Encumbrance incurred or deposit made in the ordinary course of the business of a Transaction Party in respect of a leasehold property, the purchase of assets or the use of utilities, provided that:
  - (iii) in relation to an Encumbrance incurred or deposit made in respect of the purchase of assets which secures an aggregate amount greater than \$250,000 the Bank has given prior written consent to the Borrower; and
  - (iv) the recourse of the holder of that Encumbrance is limited to the leasehold interest, the assets purchased or use of utilities and the proceeds of enforcement of the Encumbrance.
- (h) every retention of title arrangement in respect of trading stock acquired or to be acquired by a Transaction Party in the ordinary course of business;
- (i) any easement, caveat or other restriction in relation to a Freehold Property that would be apparent from a title search conducted before the date of this document; and
- (j) an Encumbrance over the Burwood Property granted by a Transaction Party in connection with the sale of the Burwood Property and the development of the Burwood Property by the purchaser (or an affiliate of the purchaser), provided that the recourse of the holder of that Encumbrance is limited to the Burwood Property and the proceeds of enforcement of the Encumbrance.

### Permitted Financial Accommodation means:

- financial accommodation granted by a Transaction Party to another Transaction Party;
- any trade credit extended by a Transaction Party to its customers on normal commercial terms and in the ordinary course of business; or
- (c) any other financial accommodation granted with the prior consent of the Bank.

### Permitted Financial Indebtedness means:

- (a) trade debt incurred in the ordinary course of business of the Transaction Parties;
- (b) Financial Indebtedness incurred under the Transaction Documents;
- (c) Financial Indebtedness owing from one Transaction Party to another

#### Transaction Party:

- (d) any Subordinated Debt;
- (e) a \$225.000 loan from the landlord of the Westlakes Cinema property;
- (f) a \$400,000 loan from the landlord of the Rhodes Cinema property;
- (g) Financial Indebtedness arising under any performance or similar bond guaranteeing performance by a Transaction Party under any contract entered into in the ordinary course of business:
- (h) Financial Indebtedness arising under a guarantee given to a landlord in respect of a lease entered into by a Transaction Party;
- (i) Financial Indebtedness under finance or capital leases of vehicles, plant, equipment or computers existing at the date of this document; and
- (j) Financial Indebtedness not permitted by the preceding paragraphs and the outstanding principal amount of which does not exceed \$500,000 in aggregate for the Transaction Parties at any time.

PPS Act means the Personal Property Securities Act 2009 (Cth)

PPS property means all property (other than Excluded Property) over which the Borrower or a Security Provider is legally capable under the PPS Act of granting a security interest.

Potential Event of Default means any thing which, with the giving of notice, lapse of time or determination of materiality, will constitute an Event of Default.

Pricing Period means, in relation to an Advance under the Corporate Markets Loan Facility, the period having the duration selected in accordance with clause 6.1 and beginning on the Funding Date in relation to the Advance.

Quarter means each three month period ending on 31 March, 30 June, 30 September and 31 December in each year.

Reading Entertainment Australia Group means, at any time, the Borrower and any subsidiary of the Borrower and Reading Entertainment Australia Group Member means any one of them.

Release Date means the Business Day following the later of:

- (a) the latest of the Expiry Dates of all Current Bank Guarantees; and
- (b) the date on which the Bank is satisfied in its reasonable opinion that it has been paid all amounts which are then or may in the future become due and payable to the Bank under any of the Transaction Documents and that there is no prospect that any amounts which the Bank has received in relation to any of the Transaction Documents will subsequently be made void or be required to be repaid in whole or in part.

Relevant Jurisdiction means Victoria.

Receiver means a receiver or receiver and manager appointed by the Bank under any Transaction Document and any person who derives a right directly

or indirectly from a Receiver.

Reference Banks means each of Australia and New Zealand Banking Group Limited, Commonwealth Bank of Australia and Westpac Banking Corporation, or any other banks or financial institutions determined by the Bank from time to time following consultation with the Borrower.

Regulatory Event means any:

- (a) change in, or introduction of a new, law or other form of regulation;
- (b) change in, or introduction of a new, practice or policy of an Government Body;
- investigation into a Transaction Party or any related entity of a Transaction Party by a Government Body;
- (d) application for or grant of an injunction or order in respect of any Encumbrance, Facility or account held with the Bank made by a Government Body, or
- (e) change in, or introduction of a new, code of practice or custom relating to the provision of the Services which a reasonable and prudent banker would comply with,

whether in Australia or elsewhere, that, in the Bank's good faith opinion, applies in any way to a Transaction Party, or the Service.

Representative of a person means an officer, employee, contractor or agent of that person.

Restatement Deed means the document entitled 'Restatement Deed' executed in December 2015 between the Bank and the Transaction Parties.

Review Event means any event or circumstance described in clause 10.4. Secured Property means all property which, from time to time, is subject to a

Security which forms part of the Collateral Security.

Security means any document or transaction which reserves or creates a Security Interest.

Security Interest means any interest or right which secures the payment of a debt or other monetary obligation or the compliance with any other obligation. It includes any retention of title to any property and any right to set off or withhold payment of any deposit or other money.

Security Provider means each person who gives a Collateral Security (other than a related body corporate of the Bank).

Service means any service the Bank provides to the Borrower under or in relation to a Facility including making or processing any payment or issuing any document.

Subordinated Debt means:

(a) Financial Indebtedness that is or may become owing by the Borrower to Reading International Cinemas, LLC, that is fully subordinated on the

terms set out in the Parent Subordination Agreement; and

(b) Financial Indebtedness that is or may become owing by a Transaction Party to Reading International Inc (or any subsidiary or affiliate of Reading International Inc) that is fully subordinated on substantially the same terms (except for the name and other details of the subordinated lender) as those set out in the Parent Subordination Agreement.

Tax means a tax (including any tax in the nature of a goods and services tax), rate, levy, impost or duty (other than a tax on the net overall income of the Bank) and any interest, penalty, fine or expense relating to any of them.

Termination Date means, in respect of each Facility, the Termination Date set out in schedule 2, or such other date agreed in writing by the parties.

Total Gross Debt means, on any date, all Financial Indebtedness of the Reading Entertainment Australia Group, but excluding any Excluded Financial Indebtedness.

Total Lease Payments means the aggregate amount of all rental expenditure of the Reading Entertainment Australia Group, other than rental expenditure payable to any Transaction Party, calculated in accordance with Accounting Standards, for that period.

Transaction Documents means:

- (a) this document;
- (b) not used;
- (c) each Guarantor Accession Deed:
- (d) the Collateral Security;
- (e) the Parent Subordination Agreement;
- (f) the ISDA Master Agreement dated 17 June 2011 between the Bank and the Borrower, as amended from time to time;
- (g) each deed of consent listed in paragraph 6 of schedule 5 upon it being executed by the relevant parties;
- (h) any agreement relating to the priority of any Security which is a Collateral Security;
- (i) any document which the Borrower and the Bank agree is a Transaction Document for the purposes of this document; and
- (j) each document entered into for the purpose of amending, novating, restating or replacing any of them.

Transaction Parties means the Borrower and each Guarantor.

Trust means, in relation to any Transaction Party that enters into a Transaction Document in the capacity as trustee of a trust, the relevant trust

Trust Deed means, in relation to a Trust, the trust deed or other document which establishes or evidences that Trust.

Trustee means a Transaction Party that enters into a Transaction Document acting as the trustee of a Trust.

Valuation means a valuation of the Freehold Properties or leasehold properties included in the Secured Property addressed to the Bank, by an Approved Valuer in form and substance satisfactory to the Bank in all respects.

Verification Certificate means a certificate in substantially the form set out in schedule 6.

#### 1.2 Construction

Unless expressed to the contrary, in this document:

- (a) words in the singular include the plural and vice versa;
- (b) any gender includes the other genders;
- (c) if a word or phrase is defined its other grammatical forms have corresponding meanings
- (d) "includes" means includes without limitation;
- (e) no rule of construction will apply to a clause to the disadvantage of a party merely because that party put forward the clause or would otherwise benefit from it; and
  - a reference to
    - (i) a person includes a partnership, joint venture, unincorporated association, corporation and a government or statutory body or authority;
    - (ii) a person includes the person's legal personal representatives, successors, assigns and persons substituted by novation;
    - (iii) any legislation includes subordinate legislation under it and includes that legislation and subordinate legislation as modified or replaced;
    - (iv) an obligation includes a representation or warranty and a reference to a failure to comply with an obligation includes a breach of representation or warranty;
    - (v) a right includes a benefit, remedy, discretion or power;
    - (vi) time is to local time in Melbourne;
    - (vii) "\$" or "dollars" is a reference to Australian currency;
    - (viii) this or any other document includes the document as novated, varied or replaced and despite any change in the identity of the parties;
    - (ix) writing includes any mode of representing or reproducing words in tangible and permanently visible form, and includes fax transmissions;
  - (x) any thing (including any amount) is a reference to the whole or any

part of it and a reference to a group of things or persons is a reference to any one or more of them:

- (xi) this document includes all schedules and annexures to it; and
- (xii) a clause, schedule or annexure is a reference to a clause, schedule or annexure, as the case may be, of this document.

#### 1.3 Headings

Headings do not affect the interpretation of this document.

### 1.4 Corporations Act, GST and Accounting Standards

Unless expressed to the contrary:

- (a) "control", "controller", "corporation", "disclosing entity", "holding company", "marketable security", "prospective liability", "public company", "related body corporate" and "subsidiary" each has the meaning which it is defined to have in the Corporations Act;

  (b) "adjustment event", "consideration", "GST", "input tax credit", "supply", "taxable supply" and "tax invoice" each has the meaning which it is defined to have in the A New Tax System (Goods and Services Tax) Act 1999; and
- "economic entity", "entity" and "finance lease" each has the meaning which it has in the Accounting Standards.
- (d) terms have the meanings given to them in the PPS Act.

### Subsisting Events of Default and Potential Events of Default

- (a) An Event of Default subsists if it has occurred and has not been waived by the Bank in accordance with this document or remedied.
- (b) A Potential Event of Default subsists if it exists and has not been waived by the Bank in accordance with this document or remedied.

### Not used

### Inconsistency

If there is any inconsistency between this document and any other Transaction Document, then this document prevails to the extent of that inconsistency.

The Borrower enters into this document in consideration of the Bank agreeing to make the Facility available in accordance with this document.

- 3 Conditions precedent
- 3.1 Not used
- 3.2 Conditions precedent to Advances and Drawings

The obligation of the Bank to make any Advances or Drawings is subject to the further conditions precedent that the Bank is satisfied in its absolute discretion that:

- (a) the representations and warranties set out in clause 8.1 are correct and in all material respects not misleading in any material respect when the Funding Notice is given and on the Funding Date;
- (b) all fees and charges then due and payable in connection with the Facility have been paid (including the Restructure Fee set out in clause 9.1(a)); and
- (c) no Event of Default or Potential Event of Default subsists when the Funding Notice is given and on the Funding Date.
- 3.3 Not used
- 4 Facility
- 4.1 Nature
  - (a) Subject to clauses 3 and 10.2, the Bank will make available:
    - (i) the revolving Corporate Markets Loan Facility under which it will make Advances; and
    - (ii) the Bank Guarantee Facility under which it will issue Bank Guarantees at the request of the Borrower,

in accordance with this document.

(b) The Borrower may request one or more Advances and Drawings in accordance with this clause 4, but so that the Outstanding Accommodation under each Facility does not at any time exceed the relevant Facility Limit.

### 4.2 Purpose

The Borrower must only use Advances and Drawings under each Facility for the relevant purposes set out in schedule 2, and the Borrower must promptly repay to the Bank all Advances and Drawings not used for these purposes.

#### 4.3 Advances and Drawings

- (a) The Borrower may request an Advance or a Drawing by giving a Funding Notice to the Bank by 11.00 am at least one clear Business Day before the date the proposed Advance or Drawing is required.
- (b) An Advance under the Corporate Markets Loan Facility must not be for an amount which, when added to the Outstanding Accommodation (if

any) under that Facility, causes the Facility Limit for that Facility to be exceeded. In determining with an Advance will cause the Facility Limit to be exceeded:

- (i) the amount of all Advances repaid on the Funding Date are excluded from the calculation of the Outstanding Accommodation; and
- (ii) the aggregate amount of all other Advances which the Borrower has requested to be made on the same Funding Date are included in that calculation.
- (c) The Aggregate Amount of a Drawing under the Bank Guarantee Facility must not, when added to the Outstanding Accommodation (if any) under that Facility, cause the Facility Limit for that Facility to be exceeded at any time during the Funding Period. In determining whether the Aggregate Amount of a Drawing will cause the Facility Limit to be exceeded:
  - (i) the Face Value of all Bank Guarantees under a Facility which will mature on the Funding Date for the relevant Drawing are excluded from the calculation of the Outstanding Accommodation, and
  - (ii) the Aggregate Amount of all other Drawings which the Borrower has requested to be made under the same Facility and on the same Funding Date are included in that calculation.
- (d) The Bank is only obliged to make Advances or accept any Drawings during the Availability Period.

#### 4.4 Funding Notices

- (a) A Funding Notice must:
  - (i) be substantially in the form of schedule 7;
  - (ii) be signed by an Authorised Representative of the Borrower;
  - (iii) specify the proposed Funding Date which must be a Business Day during the Availability Period;
  - (iv) specify the amount of the proposed Advance or the Aggregate Amount of the proposed Drawing;
  - (v) specify the duration of the Pricing Period for each Advance; and
  - (vi) in the case of any Drawing, specify whether the Drawing is:
    - (A) to comprise the issue of a new Bank Guarantee, and if so, also specify the date to be shown as the Expiry Date, the person to be named as the Beneficiary and the Face Value of each requested Bank Guarantees; or
    - (B) deemed to comprise an existing bank guarantee that prior to the date of this document has been issued by the Bank at the request of the Borrower and, if so, specify the date shown as the Expiry Date, the person named as the

Beneficiary and the Face Value of that bank guarantee.

- (b) The requirement of a Funding Notice is for the benefit of the Bank. The Bank may waive the requirement at any time and in any manner.
- (c) A Funding Notice is irrevocable from the time of its actual receipt in legible form by the Bank.
- 4.5 Not used
- 4.6 Not used
- 4.7 Not used

#### 4.8 Bank Guarantee Facilities

In the case of the Bank Guarantee Facility on the Funding Date specified in the Funding Notice:

- (a) the Bank must for the purposes of a Drawing contemplated under clause 4.4(a)(vi)(A), issue each Bank Guarantee requested in the Funding Notice in accordance with that Funding Notice; or
- (b) the parties agree that for the purposes of a Drawing contemplated under clause 4.4(a)(vi)(B), the existing bank guarantee referred to in the Funding Notice is deemed to be a Bank Guarantee issued in accordance with the Bank Guarantee Facility and that Funding Notice.

#### Cancellation

The Borrower may cancel the Available Commitment or any part of it (being \$100,000 or an integral multiple of that amount) by giving 30 Business Days' notice to the Bank specifying the amount to be cancelled and the date on which the cancellation takes effect. The cancellation takes effect on the date specified in the notice (which must be a date not earlier than five Business Days after the date the Bank receives the notice).

# 4.10 Market disruption

- (a) If the Bank determines that a Market Disruption Event occurs or has occurred in relation to an Advance, then the Bank will promptly notify the Borrower, and the Interest Rate on that Advance for that Pricing Period will be the rate per annum which is the sum of:
  - (i) the Margin for the Advance; and
  - (ii) the rate notified to the Borrower as soon as practicable and in any event no later than the Business Day before interest is due to be paid in respect of that Pricing Period, to be that which expresses as a percentage rate per annum the cost to the Bank of funding that Advance from whatever source or sources the Bank may reasonably select.
- (b) For the purposes of clause 4.10(a):
  - (i) Market Disruption Event means:
    - (A)

at or about the time on the day (Quotation Day) for the Bank

to determine the Screen Rate for the relevant currency and Pricing Period, the Screen Rate is not available and the Bank is unable to specify another page or service displaying an appropriate rate; or

- (B) in relation to an Advance, before 5pm (local time) on the Business Day after the Quotation Day for the relevant period, the Bank notifies the Borrower, that as a result of market circumstances not limited to the Bank the cost to the Bank of funding the Advance exceeds the Screen Rate.
  - Screen Rate means the rate specified in paragraph (a) of the definition of "Base Rate".

### 4.11 Alternative basis of interest or funding

If a Market Disruption Event occurs and the Bank or the Borrower so requires, the Bank and the Borrower will enter into negotiations (for a period of not more than 30 days) with a view to agreeing a substitute basis for determining the rate of interest or discount.

### 4.12 Pricing Review Events

(ii)

- (a) The Bank has the right to review the pricing applicable to a Facility ( Review ):
  - (i) on or about the third anniversary of the date of this document;
  - (ii) at any time if the Bank reasonably believes that an Event of Default subsists;
  - (iii) at any time:
    - (A) a change occurs in the financial markets which affects financial institutions generally; and/or
    - (B) a general change occurs in the cost of funds in the financial markets in which the Bank raises funds (not being a change resulting from a change in the Bank's credit rating or any other matter relating specifically to the Bank).
- (b) The Bank may request the Borrower to provide information in connection with a Review and the Borrower must provide such information as soon as possible following receipt of the request.

# 4.13 Consequences of a Pricing Review

- a) Following a Review, the Bank may, by giving written notice to the Borrower and/or by way of advertisement in the local or national press:
  - (i) introduce a new fee, charge or premium or change an existing fee, charge or premium (including its amount, the way in which it is calculated and when it is charged); and
  - (ii) change the acceptance margin, line fee, interest rate or yield rate applicable to a Facility including by changing or introducing a margin (including by making the margin positive or negative), or

substituting a different indicator rate for the relevant indicator rate (except where the rate is a fixed rate).

- (b) Where the Bank gives the Borrower notice under clause 4.10(a) by way of advertisement in the local or national press, the Bank will also endeavour to directly notify the Borrower of the change although the Bank will not be precluded from charging the new or adjusted pricing if it does not directly notify the Borrower.
- (c) An introduction or change of a matter specified in clause 4.10(a) takes effect on the date specified in the relevant notice to the Borrower (which must be at least 30 days after the date on which the notice is given to the Borrower).

### 5 Payments

#### 5.1 Not used

### 5.2 Voluntary prepayments

- (a) In relation to any Advance, the Borrower:
  - (i) may prepay any Advance or a part of it (being a minimum of \$100,000 or an integral multiple of that amount) by giving 5 Business Days' notice to the Bank specifying the amount to be prepaid and the date on which the prepayment will be made;
  - ii) may, subject to clause 4.3, redraw any amount prepaid in accordance with this clause 5.2; and
  - (iii) must make any prepayment under this document together with accrued interest on the amount prepaid, any fees payable under clause 9.1 and any Break Costs, but otherwise without premium or penalty.
- (b) The Borrower may reimburse or repay the Face Value in respect of any Current Bank Guarantee by:
  - (i) providing to the Bank, cash collateral (on terms satisfactory to the Bank and subject to clause 10.3) in an amount not less than the Face Value of the Bank Guarantee; or
  - (ii) cancelling that Bank Guarantee by returning the original to the Bank together with written confirmation from the Beneficiary that the Bank has no further liability under that Bank Guarantee.

### 5.3 Indemnity in respect of Bank Guarantees

(a) Without limiting clause 12.1, the Borrower indemnifies the Bank against any liability, loss, cost or expense sustained or incurred in relation to any Bank Guarantee or as a direct or indirect consequence of any claim made or purported to be made under any Bank Guarantee, or anything done by any person who is or claims to be entitled to the benefit of a Bank Guarantee.

- (b) Without limiting clause 5.3(a), the Borrower must pay to the Bank all amounts claimed by or paid to any Beneficiary in relation to any Bank Guarantee (whether or not the Beneficiary was entitled to make that claim or the Bank was required to make that payment), including any payment made by the Bank under clause 10.2(a)(iv)(B).
- (c) The Borrower's obligations under clause 5.4(a) and (b) are absolute and unconditional. They are not affected by any reduction, termination or other impairment by set-off, deduction, abatement, counterclaim, agreement, defence, suspension, deferment or otherwise.
- (d) The Borrower is not released, relieved or discharged from any obligation under this document, nor will such obligation be prejudiced or affected for any reason, including:
  - (i) any falsity, inaccuracy, insufficiency or forgery of or in any demand, certificate or declaration or other document which on its face purports to be signed or authorised under a Bank Guarantee;
  - ii) any failure by the Bank to enquire whether a cable, telex or other notification was inaccurately transmitted, received or given by an unauthorised person (other than where such failure occurs due to the wilful default or fraud of the Bank);
  - (iii) the impossibility or illegality of performance of, or any invalidity of or affecting, any Transaction Document or Bank Guarantee or any other document;
  - (iv) any act of any Government Body or arbitrator including any law, judgment, decree or order at any time in effect in any jurisdiction affecting any Transaction Document or Bank Guarantee or any document delivered under a Transaction Document;
  - v) any failure to obtain any consent, license or other authorisation necessary or desirable in connection with any Transaction Document or any Bank Guarantee; or
  - (vi) any other cause or circumstance, foreseen or unforeseen, whether or not similar to any of the above, affecting any Transaction Document or Bank Guarantee or any transaction under a Transaction Document or Bank Guarantee.
  - (vii) and the Bank need not inquire into any of these matters.
- (e) The Bank is irrevocably authorised and directed by the Borrower to pay immediately against a demand appearing or purporting to be made by or on behalf of a Beneficiary, any sums up to the Face Value of a Bank Guarantee which may be demanded from the Bank from time to time without any reference to or any necessity for confirmation or verification on the part of the Borrower, and notwithstanding any instructions from the Borrower to the contrary.
- (f) The obligations of the Borrower will not be affected or in any way limited by any falsity, inaccuracy, insufficiency or forgery of or in any notice or

demand pursuant to any liability or the failure of the Bank to enquire (other than where such failure arises due to the wilful default or fraud of the Bank) whether any notice or demand has been inaccurately transmitted or received from any cause whatsoever or has been given or sent by an unauthorised person.

#### Mandatory prepayments

- (a) Unless the Bank otherwise agrees, if any of the assets, business or undertaking of any Transaction Party is the subject of any Disposal (other than a Permitted Disposal) the Borrower must apply or ensure is applied an amount equal to the cash or equivalent proceeds received by the Transaction Party from the Disposal net of reasonable transaction costs and Taxes in prepayment of Outstanding Accommodation or (other than in respect of a disposal contemplated by clause 5.4(b)), at the Borrower's election, in permanent reduction of the unused portion of one or more of the Facility Limits.
- If the land and improvements known as 70 Station Road, Indooroopilly Queensland and described in certificate of title 11485156 are sold:
  - (i) unless the Bank otherwise agrees in writing, the Borrower must apply or ensure is applied an amount equal to 50% of the proceeds arising from that sale net of reasonable transaction costs and Taxes in prepayment of the Outstanding Accommodation and/or in permanent reduction of the unused portion of the Facility Limit for the Corporate Markets Loan Facility; and
  - (ii) subject to the prior written consent of the Bank (such consent not to be unreasonably withheld), a Transaction Party may make a Distribution or other payment to Reading International Inc (or any subsidiary or affiliate of Reading International Inc that is outside of the Reading Entertainment Australia Group) of an amount equal to 50% of the proceeds arising from that sale net of reasonable transaction costs and Taxes.
- (c) For the avoidance of doubt the requirements to prepay under clause 5.4(b) will to the extent of the prepayment permanently reduce the Facility Limit of the relevant Facility or Facilities.

Subject to clause 10.2 and clause 10.3, each Borrower must:

- (a) repay the Outstanding Accommodation in respect of each Facility on the Termination Date in respect of that Facility; and
- (b) subject to clause 6, and any other provision in a Transaction Document that provides otherwise, pay any other amounts payable in connection with the Transaction Documents, to the Bank on demand.

#### 5.6 Not used

#### 6 Interest and fees

#### 6.1 Pricing Periods

- (a) Subject to clause 6.1(c), the Pricing Period for each Advance must be a period of 30, 60 or 90 days or six Months or another period agreed by the Bank.
- Subject to clause 6.1(c), the first Pricing Period for an Advance commences on its Funding Date and will have the duration specified in the relevant Funding Notice. Each subsequent Pricing Period for the Advance:
  - (i) commences on the day after the preceding Pricing Period for the Advance expires; and
- (ii) is a period notified by the Borrower to the Bank at least two Business Days before the last day of the current Pricing Period, but if the Borrower does not give notice, is of the same duration as the Pricing Period which immediately precedes it.

### (c) A Pricing Period:

- (i) which would otherwise end on a day which is not a Business Day ends on the next Business Day and a Pricing Period which would otherwise end after the Termination Date ends on the Termination Date. For the avoidance of doubt, if a Pricing Period ends on a day that is not followed by a Business Day, the Bank may extend that Pricing Period accordingly (except where this would be contrary to clause 6.1(c)(ii), in which case the Bank may shorten the Pricing Period); and
- (ii) May be adjusted by the Bank where necessary so that:
  - (A) a Pricing Period starts on a Business Day;
  - (B) all Advances will have the same Pricing Period;
  - (C) a Pricing Period does not end after the Termination Date; and
  - (D) if a new Advance is made during a Pricing Period for an existing Advance, the first Pricing Period for that new Advance ends on the same day as the Pricing Period for the existing Advance.

### 6.2 Payment and rate

- (a) In respect of the Corporate Markets Loan Facility:
  - (i) interest for each day is calculated by applying the Daily Interest Rate to the Advance at the end of that day (excluding any amount to which the Overdue Rate applies); and

- (ii) the Borrower must pay accrued interest in respect of:
  - (A) each Pricing Period, on the First Business Day after the expiry of that Pricing Period; and
  - (B) the last Pricing Period, for the period up to and including the Termination Date, on the Termination Date.
- The Borrower must pay interest on Overdue Money, and such interest must be paid on demand by the Bank.
- (c) The interest rate on Overdue Money will be the Overdue Rate.

#### 6.3 Computation of interest

Interest will:

- (a) accrue from day to day;
- (b) be computed from and including the day when the money on which interest is payable becomes owing to the Bank by the Borrower until but excluding the day of payment of that money; and
- (c) be calculated on the actual number of days elapsed on the basis of a 365 day year.

### 6.4 Capitalisation of interest

The Bank may:

- (a) capitalise, on a monthly or other periodical basis as the Bank determines, any part of any interest which becomes due and payable and interest is payable in accordance with this document on capitalised interest; and
- (b) continue to capitalise interest despite:
  - that as between the Bank and the Borrower the relationship of Bank and customer has ceased;
  - (ii) any composition agreed to by the Bank;
  - (iii) any judgment or order against the Borrower; or
  - (iv) any other thing.

### 6.5 Merger

If the liability of the Borrower to pay to the Bank any money payable under a Transaction Document becomes merged in any deed, judgment, order or other thing, the Borrower must pay interest on the amount owing from time to time under that deed, judgment, order or other thing at the higher of the rate payable under the Transaction Documents and that fixed by or payable under that deed, judgment, order or other thing.

# 7 Payments

# 7.1 Place, manner and time of payment

Each Transaction Party must make payments to the Bank under the Transaction Documents:

- (a) at the address specified in clause 19.3 or at such other place reasonably required by the Bank;
- (b) in a manner reasonably required by the Bank;
- (c) by 11.00 am local time in the place where payment is required to be made; and
- (d) in immediately available funds and without set-off, counter claim, condition or, unless required by law, deduction or withholding.

# 7.2 Gross-up

If a Transaction Party is required by law to deduct or withhold Taxes from any payment it must:

- (a) make the required deduction and withholding;
- (b) pay the full amount deducted or withheld in accordance with the relevant law;
- (c) deliver to the Bank an original receipt for each payment; and
- (d) pay an additional amount with such payment so that, after all applicable deductions or withholdings, the Bank actually receives for its own benefit the full amount which would have been payable to the Bank if no deduction or withholding had been required.

#### 7.3 Appropriation

Subject to any express provision to the contrary in any Transaction Document, the Bank may appropriate any payment towards the satisfaction of any money due for payment by the Borrower in relation to a Transaction Document in any way that the Bank thinks fit and despite any purported appropriation by the Borrower.

### 8 Representations and warranties

### 8.1 Nature

Each Transaction Party represents and warrants that:

- (a) duly incorporated: if it purports to be a corporation, it is duly incorporated in accordance with the laws of its place of incorporation, validly exists under those laws and has the capacity to sue or be sued in its own name and to own its property and conduct its business as it is being conducted;
- (b) capacity: it has capacity unconditionally to execute and deliver and comply with its obligations under the Transaction Documents;

- action taken: it has taken all necessary action to authorise the unconditional execution and delivery of, and the compliance with its obligations under, the Transaction Documents to which it is a party; (c)
- (d) binding obligations: each Transaction Document constitutes the valid and legally binding obligations of, and is enforceable against it by the Bank in accordance with its terms (subject to any necessary stamping or registration and to equitable principles and insolvency laws);
- priority: each Security Interest which each Transaction Document purports to create exists and has the priority which the Bank has agreed to (subject to any necessary stamping and registration);
- authorisations: each authorisation from, and filing and registration with, a Government Body necessary to enable it to unconditionally execute and deliver and comply with its obligations under the Transaction Documents to which it is a party has been obtained, effected and complied with;
- (g) no contravention: the unconditional execution and delivery of, and compliance with its obligations by it under, the Transaction Documents to which it is a party do not:
  - (i) contravene any law to which it or any of its property is subject or any order or directive from a Government Body binding on it or any of its property;
  - (ii) contravene its constituent documents;
  - (iii) contravene any agreement or instrument to which it is a party;
  - (iv) contravene any obligation it has to any other person; or
  - require it to make any payment or delivery in relation to any Financial Indebtedness (other than Excluded Financial Indebtedness) before the scheduled date for that payment or delivery;
- (h) correct information: all information given and each statement made to any Bank by it or at its direction in relation to the Transaction Documents, is correct, complete and not
- (i) full disclosure: it has disclosed to the Bank all information which the Borrower has or has access to and which is relevant to the assessment by the Bank of the nature and amount of the risks undertaken by the Bank becoming a creditor of or taking a Security from it;

  (j) Financial Statements: the Financial Statements of each of Transaction Party given to the Bank under clause 9.3:
- - (i) are a true, fair and accurate statement of their respective financial performance and position and their respective consolidated financial performance and position at the date to which they are prepared; and
  - (ii) have been prepared in accordance with clause 9.2 and 9.3,

except for such departures expressly disclosed in those Financial Statements;

- (k) no change in financial position: there has been no change in the financial performance or position of a Transaction Party since the date to which the last Financial Statements given to the Bank under clause 9.3 were prepared, which has a Material Adverse Effect;
- (I) no related party transaction: no person has contravened or will contravene sections 208 or 209 of the Corporations Act due to a Transaction Party entering into or performing its obligations under a Transaction Document;
- (m) no proceeding: except as notified to the Bank in writing before the date of this document, no litigation, arbitration or administrative proceeding is current, pending or, to the knowledge of the Borrower, threatened, which has, or the adverse determination of which would be likely to have, a Material Adverse Effect;
- (n) no trust: except as notified to the Bank in writing before the date of this document, no Transaction Party enters into a Transaction Document as trustee of any trust;
  - sole owner and no Encumbrances: except as notified to the Bank in writing before the date of this document:
  - (i) each Transaction Party is the sole legal and beneficial owner of the property it purports to own; and
  - (ii) there are no Encumbrances over the property of any Transaction Party other than Permitted Encumbrances;
- (p) no existing default: no Event of Default, Review Event or Potential Event of Default subsists;
- (q) ranking of obligations: each obligation of the Borrower under this document ranks at least pari passu with all unsecured and unsubordinated obligations of the Borrower except obligations mandatorily preferred by law;
- (r) warranties correct: the representations and warranties given by any Transaction Party in any Transaction Document are correct in all material respects and not misleading in any material respect and will be when given or repeated;
- (s) no immunity: each Transaction Party and its property are free of any right of immunity from set-off, proceedings or execution in relation to its obligations under any Transaction Document;
- (t) insurance: the Insurances are enforceable against the relevant insurer in accordance with their terms and are not void or voidable;
- (u) trust provisions: in relation to each Transaction Party which enters into any Transaction Document as trustee of a Trust:
  - (i) the Trustee has power as trustee of the Trust to execute and

perform its obligations under the Transaction Documents;

- (ii) the Trustee, in executing the Transaction Documents and entering into those transactions, have properly performed their obligations to the beneficiaries of the Trust;
- (iii) all necessary action required by the Trust Deed to authorise the unconditional execution and delivery of, and compliance with its obligations under, the Transaction Documents has been taken;
- (iv) the Trustee is the only trustee of the Trust;
- (v) no effective action has been taken to remove the Trustee as trustee of the Trust or to appoint an additional trustee of the Trust;
- vi) (A) the Trustee has a right to be fully indemnified out of the property of the Trust in relation to all of its obligations under the Transaction Documents;
  - (B) the Trustee has not released or disposed of its equitable lien over the property of the Trust which secures that indemnity; and
  - (C) the property of the Trust is sufficient to satisfy that indemnity;
- (vii) the Trustee has complied with all of its obligations as trustee of the Trust in relation to execution of the Transaction Documents;
- (viii) no effective action has been taken or, so far as the Trustee is aware, is contemplated by the beneficiaries of the Trust to terminate the Trust;
- (ix) the Trustee has disclosed to the Bank full details of:
  - (A) the Trust and any other trust or fiduciary relationship affecting the property of the Trust and, without limitation, has given to the Bank copies of any instruments creating or evidencing the Trust; and
  - (B) the Trustee's other trusteeships (if any);
- (x) the Trust is properly constituted and the Trust Deed is not void, voidable or otherwise unenforceable;
- (xi) the rights of the beneficiaries of the Trust in relation to, and their interest in, the property of the Trust are subject to:
  - (A) the rights of the Bank in relation to, and their respective interests in, the property of the Trust; and
  - (B) any rights or interests in the property of the Trust to which the Bank may from time to time be subrogated; and
- (xii) the Trustee:
  - (A) if it is a corporation, is duly incorporated in accordance with the laws of its place of incorporation, validly exists under those laws and has the capacity to sue and be sued in its

own name, to own property and to act as trustee of the Trust;

- (B) if it is natural person, has the capacity to be trustee of the Trust;
- (v) solvency: each Transaction party is not insolvent;
- (w) corporate benefit: each of the Transaction Parties will receive corporate benefit by entering into the Transaction Documents to which they are a party.

### General

- (a) The interpretation of any statement contained in any representation or warranty will not be restricted by reference to or inference from any other statement contained in any other representation or warranty.
- The Borrower acknowledges that the Bank enters into the Transaction Documents in reliance on each representation and warranty.
- (c) Each representation and warranty survives the execution of the Transaction Documents and is deemed to be repeated with reference to the facts and circumstances then existing on the date each Funding Notice is issued, on each Funding Date, on the last day of each Funding Period and on each day that an Annual Compliance Certificate or Interim Compliance Certificate is given.

### General obligations

#### 9.1 Fees

The Borrower must pay to the Bank:

- (a) restructure fee : on or before execution of the Restatement Deed, a non- refundable Restructure Fee of \$30,000;
- (b) Corporate Markets Loan Facility Fee: a non-refundable facility fee on the Facility Limit in respect of the Corporate Markets Loan Facility calculated at 0.95% per annum from the date of the 'Variation Date' under the Restatement Deed, which will:
  - (i) accrue from day to day from the date of this document up to and includir (ii) be payable monthly in arrears, on the first Business Day of each Month; accrue from day to day from the date of this document up to and including the Termination Date;

  - be calculated on the actual number of days elapsed and on the basis of a 365 day year;
- (c) Reset Fee : on the first Business Day of each Pricing Period (other than the first Pricing Period) a non-refundable fee of \$150.00;
- (d) Not used
- (e) Bank Guarantee service fee : on and from the first services fee charge

date following 30 June 2014, a non-refundable fee in respect of each Bank Guarantee of 50% of the applicable Bank Guarantee Margin calculated on the Face Value of the Bank Guarantee, payable on a pro- rata basis half yearly in arrears, with the first payment due six months after the relevant Funding Date of the Bank Guarantee, and subsequent payments due every six months thereafter until the Bank Guarantee Matures or Expires or is cancelled. This fee will be calculated on the actual number of days elapsed and on the basis of a 365 day year, and

(f) Bank Guarantee issuance fee: a non-refundable fee in respect of each Bank Guarantee of 50% of the applicable Bank Guarantee Margin calculated on the Face Value of the Bank Guarantee (or \$125 whichever is greater), payable on the relevant Funding Date of the Bank Guarantee.

#### 9.2 Records

The Borrower must ensure that each Transaction Party:

- prepares and keeps books, accounts and other records in accordance with the law and Accounting Standards; and
- (b) on demand, makes the same available for inspection and copying by the Bank.

### 9.3 Financial Statements and other financial information

The Borrower must give to the Bank:

- (a) annual Financial Statements: as soon as practicable, and in any event within 120 days after the end of each Financial Year the consolidated audited Financial Statements of the Reading Entertainment Australia Group for that Financial Year;
- (b) Quarterly Financial Statements: as soon as practicable, and in any event within 45 days after the end of each Quarter (other than the Quarter ending 31 December) the consolidated unaudited Financial Statements of the Reading Entertainment Australia Group for that Quarter (showing both actual and budget figures);
- (c) group structure diagram: within 120 days after the end of each Financial Year, a group structure diagram in relation to Reading International Inc. and the Reading Entertainment Australia Group which lists all the then Group Members and which contains such other information in relation to the legal relationship between Reading International Inc. and the Reading Entertainment Australia Group Members as the Bank reasonably requires;
- (d) budget: as soon as practicable, and in any event before 31 March for each Financial Year, a consolidated budget for the Reading Entertainment Australia Group for the current Financial Year showing the budgeted profit and loss, balance sheet and cash flow for the Reading Entertainment Australia Group and such other matters customarily dealt with in such budgets;
- (e) deregistration : not used : and

(f) other financial information: promptly on reasonable notice from the Bank, such additional information in relation to the financial condition and the operations of the Borrower and each other Transaction Party as the Bank reasonably requests from time to time.

The Borrower must ensure that all Financial Statements given to the Bank under the Transaction Documents are prepared in accordance with the Corporations Act and the Accounting

If after the date of this document there is a change in the accounting principles or practices referred to in the definition of 'Accounting Standards' and the Bank or the Borrower reasonably considers that, if the change were to apply for the purposes of this document, the change would have a material effect on the Financial Statements or the calculation of the financial ratios in clause 9,7, the Bank and the Borrower shall endeavour to agree mutually acceptable changes to this document so that the accounting change can be adopted for the purposes of this document.

#### 9.4 Other information

The Borrower must give to the Bank:

- (a) other information: on reasonable notice from the Bank, any other information in the possession or under the control of a Transaction Party which in the Bank's reasonable opinion is necessary to verify the Borrower's compliance with any Transaction Document;

  (b) Annual Compliance Certificate: as soon as practicable, and in any event within 120 days after the end of each Financial Year, an Annual Compliance Certificate for that Financial Year signed by at least one director of the Borrower;
- (c) Interim Compliance Certificate: as soon as practicable, and in any event within 45 days after the end of each Quarter (other than the Quarter ending 31 December) an Interim Compliance Certificate for the previous 12 months signed by at least one director of the Borrower;
- (d) tenancy schedule: as soon as practicable, and in any event within 120 days of the end of each Financial Year an updated tenancy schedule for each Freehold Property, including (without limitation) the following details:
  - (i) the name of each tenant;
  - (ii) area let by each tenant;
  - (iii) current passing rent paid by each tenant;
  - (iv) the lease start date;
  - (v) the lease term;
  - (vi) the lease maturity date;
  - (vii) the option term (if any);
  - (viii) rent review details; and

**JA3151** 

- (ix) any other material or special clauses or conditions;
- (e) Valuations: on demand (provided that no more than one demand is made in a Financial Year and the Bank reasonably considers that there has been a material devaluation of the freehold and leasehold interests subject to the Collateral Security), and in any event within 30 days of the end of every 36 month period from the date of this document, a Valuation in respect of each Freehold Property and leasehold interest that is subject to the Collateral Security. Each Valuation is to be at the Borrower's expense, addressed to the Bank, conducted by an Approved Valuer and in a form and substance (other than as to value) reasonably satisfactory to the Bank;
- (f) not used

(c)

- (g) details of any proceeding: full details of any litigation, arbitration, administrative proceeding or native title claim which affects a Transaction Party and which has or the adverse determination of which would be likely to have a Material Adverse Effect, as soon as it is commenced or to the knowledge of the Borrower is threatened; and
- (h) claims: on being notified of it, full details of any event which entitles the Borrower or the Bank to claim more than \$1,000,000 under the Insurances.

#### 9.5 Other financial undertakings

Each Transaction Party must ensure that:

- (a) negative pledge: no Encumbrances exist on its property (including the Secured Property and the land and improvements known as 78 Middleborough Road, Old Burwood Road, Burwood Victoria), except Permitted Encumbrances;
- (b) permitted financial transactions: it does not, without the prior written consent of the Bank:
  - incur any Financial Indebtedness except Permitted Financial Indebtedness;
  - (ii) provide any financial accommodation (excluding trade credit in the ordinary course of business) except Permitted Financial Accommodation;
  - disposals: must not dispose of any of its assets, either in a single transaction or in a series of transactions whether related or not and whether voluntary or involuntary, except Permitted Disposals;
- (d) mergers: a Transaction Party does not:
  - (i) enter into any merger, reconstruction or amalgamation; or
  - (ii) acquire any property or business or make any investment if the property, business or investment is substantial in relation to the relevant Transaction Party.

if it would have or be likely to have a Material Adverse Effect;

- (e) maintain status: it does everything necessary to maintain its corporate existence in good standing and:
  - (i) ensures that it has the right and is properly qualified to conduct its business in all relevant jurisdictions; and
  - (ii) obtains and maintains all Authorisations necessary for the conduct of its business;
  - (iii) comply with all laws affecting it or its business in all relevant jurisdictions
- (f) Distributions: it must not make any Distribution except:
  - (i) Permitted Distributions;
  - (ii) to another Transaction Party; or
  - (iii) with the Bank's prior written consent (such consent not to be unreasonably withheld),

and provided that:

- (iv) no Event of Default subsists; and
- (v) such Distribution will not cause an Event of Default;
- (g) Taxes: must
  - (i) promptly pay when they become due for payment (or reimburse the Bank on demand for) all Taxes payable by it from time to time other than Taxes being contested in good faith where it has made adequate provisioning;
  - (ii) not transfer any Tax losses to any person other than to the Borrower in connection with the preparation of consolidated annual Financial Statements or in connection with the Reading Entertainment Australia Group's tax consolidation arrangements; and
  - (iii) not become a member of a consolidated group for the purposes of Part 3-90 of the Income Tax Assessment Act 1936 and the Income Tax Act 1997 including any amendments thereto (including any amendments made by the Now Business Tax (Consolidation Act (No. 1)) 2002 and the Now Business Tax System (Consolidation, Value Shifting, Damages and other Measures) Act 2002) other than in accordance with a Tax Sharing Agreement or otherwise on terms approved by the Bank;
- (h) Guarantor accession: if at any time:
  - (i) there is a new wholly-owned member of the Reading Entertainment Australia Group that earns any revenue, holds an asset or is otherwise not dormant; or
  - (ii) a wholly-owned member of the Reading Entertainment Australia

Group (as at the date of this document that is not a Guarantor) generates any revenue, acquires an asset or ceases to be dormant.

promptly, and in any event within 45 days, it procures the execution of:

- (iii) a Guarantor Accession Deed; and
- an all assets fixed and floating charge in favour of the Bank, in such form as the Bank requires, (iv)

by such of those wholly-owned subsidiaries as are required to rectify that situation and provides to the Bank any documents or evidence in relation to any such acceding entity as the Bank may reasonably consider necessary in respect to the entering into, validity and enforceability of the accession documents;

- Major developments: in respect of any major development projects to be undertaken by the Transaction Parties (that are outside of the budgeted capital expenditure that has been disclosed to the Bank): (i)
  - (i) the Bank is provided with development budgets and other informat
    (ii) the Bank is given the first right of refusal in relation to the funding of the project; the Bank is provided with development budgets and other information reasonably requested by the Bank; and
- (j) Major acquisitions: in respect to any acquisitions or investments in assets to be undertaken by the Transaction Parties, the Bank's written consent is obtained for (and prior to) the purchase of:
  - (i) any freehold title or ground lease with a remaining tenor of 25 years or more and a consideration greater than \$50,000,000; and
  - the purchase of any other operating business assets with a consideration greater than \$25,000,000.
- (k) Management Fees: the aggregate Management Fees paid by the Transaction Parties in a Financial Year does not exceed the amount set out in the 'Amount' column corresponding to that Financial Year set out in the following table (without the Bank's prior written consent):

Financial Year Ending	Amount
31 December 2015	\$5,250,000
31 December 2016	\$5,512,500
31 December 2017	\$5,788,125
31 December 2018	\$6,077,531
31 December 2019	\$6,381,408

No Management Fees may be paid while an Event of Default subsists;

(I) Burwood property:

- (i) no Encumbrances exist on the land and improvements known as 78 Middleborough Road, Old Burwood Road, Burwood Victoria (except Permitted Encumbrances); and
- (ii) it does not, without the prior written consent of the Bank, incur any Financial Indebtedness in respect of (or secured by) the land and improvements known as 78 Middleborough Road, Old Burwood Road, Burwood Victoria (except Permitted Financial Indebtedness);
- (m) Preservation and protection of Security: it does everything necessary or reasonably required by the Bank to:
  - (i) keep the Secured Property in good repair and in good working order,
  - (ii) promptly pay when they become due for payment (or reimburse the Bank on demand for) all Taxes payable in respect of the Secured Property;
  - (iii) preserve and protect the value of the Secured Property as a whole; and
  - (iv) protect and enforce its title and the Bank's title as mortgagee to the Secured Property.

### 9.6 Insurance

- (a) Subject to the provisions of the Transaction Documents, the Borrower must effect and maintain insurance over and in relation to the Secured Property, the business operations of the Group (including business interruption) and for public liability with insurers, for amounts, against risks and on terms and conditions:
  - (i) that the Bank reasonably requires; or
  - (ii) if the Bank does not notify the Borrower of its requirements, that a prudent and reasonable owner of the Secured Property would effect and maintain, including insurance for full replacement value on a reinstatement basis.
  - Subject to the provisions of the Transaction Documents, the Borrower must give to the Bank on demand a certificate in form and substance satisfactory to the Bank from the insurer to the effect that the required Insurances are current and no premium is overdue.

### 9.7 Financial ratios

- (a) The Borrower must ensure that:
  - (i) Fixed Charges Cover Ratio : at each Calculation Date, the Fixed Charges Cover Ratio for the Calculation Period ending on that date is not less than 1.70 times.
  - (ii) Leverage Ratio: at each Calculation Date, the Leverage Ratio for the Calculation Period ending on that date is less than or equal to

3 25 times

- (iii) Loan to Value Ratio: at each Calculation Date, the Loan to Value Ratio for the Calculation Period ending on that date is less than or equal to 70%.
- (b) A financial ratio or amount to be determined under clause 9.7(a) must be tested or determined by reference to the most recently prepared. Financial Statements. The calculation of any amounts on a consolidated basis must be made in accordance with the requirements of the Accounting Standards relating to the consolidation of entities.

#### 9.8 Environment

- (a) Each Transaction Party must ensure that at all times all practical and reasonable steps that can be taken and measures and precautions that can be adopted are taken or adopted by each Transaction Party to ensure that:
  - (i) all persons, things and activities of any kind on or using the Land comply with all Environmental Laws and any consent, permit, approval, licence, authorisation, certification, order or direction granted or issued under any Environmental Law;
  - (ii) if there is any non-compliance with any Environmental Law or any consent, permit, approval, licence, authorisation, certification, order or direction granted or issued under any Environmental Law.
    - (A) the impact on the Land and the environment is minimised; and
    - (B) steps are taken as quickly as possible to rectify the non-compliance, eliminate or reduce any liability arising from the non-compliance and to ensure the non-compliance does not recur;
  - (iii) it or any person on the Land does not:
    - (A) allow onto or permit to exist on the Land any Contaminant; or
    - (B) allow a Contaminant to escape or be released into the environment,
      - if to do so would be in breach of any Environmental Law or any consent, permit, approval, licence, authorisation, certification, order or direction granted or issued under any Environmental Law; and
  - (iv) if any Contaminant is discovered on or affecting the Land (other than a Contaminant which is safely stored in accordance with lawful authority) or, without lawful authority, escapes or is released from the Land into the environment:
    - (A) the impact on the Land and the environment is minimised;

- (B) steps are taken as quickly as possible to safely contain the Contaminant and to remove the Contaminant from the environment or the Land or reduce the levels of the Contaminant to a level required or recommended by the relevant Government Body as safe and in either case to eliminate or reduce any liability arising from the Contaminant and do all things necessary to restore the Land and the environment.
- (b) If there is any non-compliance under clauses 9.8(a)(i), (ii) or (iii) or any Contaminant is discovered or the Borrower has reason to believe that there is some Contaminant on the Land requiring action to be taken under clause 9.8(a)(iv), the Borrower must immediately notify the Bank.
- (c) If there is or the Bank has reason to believe that there may be any
  - non-compliance under clauses 9.8(a)(i), (iii) or (iii) or any Contaminant is discovered or the Bank has reason to believe that there is some Contaminant on the Land requiring action to be taken under clause 9.8(a)(ii), the Borrower, at the request of the Bank, must procure and furnish to the Bank, in a form acceptable to the Bank, an Environmental Assessment Report in relation to the Land and any operations conducted on it.
- (d) The Borrower indemnifies the Bank from and against all:
  - (i) Environmental Liability; and
  - (ii) damages, losses, outgoings, costs, charges or expenses suffered or incurred by the Bank in respect of any action, claim or demand made or brought in respect of or otherwise arising from or in connection with any breach of any Environmental Law in relation to the Land.
- (e) The Borrower must immediately notify the Bank of:
  - (i) the existence of any Contaminant on or adjacent to or affecting the Land; and
  - the receipt by any Transaction Party of any notice, order or direction: (ii)
    - (A) to clean up any Contaminant on the Land; or
    - (B) alleging any breach of Environmental Law.
- If requested by the Bank, the Borrower must provide the Bank with a copy of each environmental consent, permit, approval, licence, authorisation, certification, order and direction relating to the Land together with confirmation that: (f)
  - (i) it is complying with the terms and conditions of each consent, permit, approval, licence, authorisation, certification, order and direction; and

- (ii) it has renewed each consent, permit, approval, licence, authorisation, certification, order and direction as appropriate
- (g) The Borrower must:
  - (i) when reasonably required by the Bank, obtain or permit the Bank to obtain an Environmental Assessment Report from a person approved by the Bank in relation to the Land; and
  - (ii) promptly comply with any reasonable recommendation contained in any Environmental Assessment Report relating to compliance with Environmental Law in relation to the Land and obtain any consent, permit, approval, licence, authorisation, certification, order and direction required in order to comply with that recommendation.

### 9.9 No default

The Borrower must ensure that an Event of Default does not occur.

### 9.10 Obligations of Trustees

If a Transaction Party is a Trustee the Borrower must ensure that it:

- (a) ensures that the property of the Trust is not mixed with any other property;
- (b) complies with its obligations as trustee of the Trust;
- (c) does not release, dispose of or otherwise prejudice its right of indemnity against, and equitable lien over, the property of the Trust and its right of indemnity (if any) against the beneficiaries of the Trust in relation to any money owing to the Bank;
- (d) at the Bank's request:
  - (i) exercises its right of indemnity against, and equitable lien over, the property of the Trust and its right of indemnity (if any) against the beneficiaries of the Trust in relation to any money owing to the Bank; and
  - (ii) assigns to the Bank those indemnities and that equitable lien and otherwise facilitates the subrogation of the Bank to those indemnities and that equitable lien;
- (e) does not, if the Trust is a unit trust, consent to or register the transfer of units in the Trust or cancel, repurchase, redeem or issue any units in the Trust;
- (f) ensures that:
  - (i) another person is not appointed as trustee of the Trust;
  - (ii) the Trust is not terminated or its terms varied;
  - (iii) the Trustee does not resign and is not removed or replaced as trustee of the Trust;
  - (iv) the property of the Trust is not resettled;

- (v) the capital of the property of the Trust is not distributed at any time; and
- (vi) income of the Trust is not distributed to anyone other than a Transaction Party while an Event of Default or Potential Event of Default subsists;
- (g) prepares and keeps full and true records and books of accounts of the Trust and makes them available for inspection and copying by the Bank on demand; and
  - does not default in performing or observing its obligations under the Transaction Documents.

### 9.11 Release for Permitted Disposals

The Bank must on request from (and at the cost of) a Transaction Party release from the Collateral Security that part of the Secured Property that is the subject of a Permitted Disposal (other than a Permitted Disposal of the kind referred to in paragraph (a) of that term's definition).

# 10 Events of Default

### 10.1 Nature

(h)

Each of the following is an Event of Default (whether or not caused by anything outside the control of any Transaction Party):

- a) non-payment: a Transaction Party does not pay on the due date any principal, interest and fees due for payment by it under a Transaction Document in accordance with the relevant Transaction Document;
- (b) other non-compliance: a Transaction Party does not comply with any other obligation under a Transaction Document and if that default is capable of rectification:
  - (i) it is not rectified within five Business Days (or any other longer period agreed by the Bank) after its occurrence; or
  - (ii) the Transaction Party does not during that period take all action which in the Bank's reasonable opinion is necessary to rectify that default;
- (c) untrue warranty: a representation, warranty or statement made or deemed to be made by a Transaction Party in a Transaction Document is untrue or misleading in any material respect or a reply by a Transaction Party to a requisition made by, or on behalf of, the Bank is untrue or misleading in any material respect;
- (d) void document: a Transaction Document is void, voidable or otherwise unenforceable by the Bank or is claimed to be so by a Transaction Party;
- (e) compliance unlawful: it is unlawful for a Transaction Party to comply with any of its obligations under a Transaction Document or it is claimed to be so by a Transaction Party;

**JA3159** 

- (f) loss of priority: a Security Interest created by or purportedly created by a Collateral Security does not have or ceases to have the priority which it purports to have under the relevant Transaction Document or becomes ineffective to secure the payment of the money or compliance with the obligations which it purports to secure, otherwise than by any act of the Bank;
- (g) Insolvency Event: an Insolvency Event occurs in relation to a Transaction Party;
- (h) authorisation ceasing: an Authorisation from a Government Body necessary to enable:
  - (i) a Transaction Party to comply with its obligations under a Transaction Document or carry on its principal business or activity;
  - (ii) a Transaction Party to carry on its principal business or activity; or
  - (iii) the Bank to exercise its rights under a Transaction Document, is withheld or ceases to be in full force and effect and, in the case of
  - clause 10.1(h)(i), would have a Material Adverse Effect;
- (i) Material Adverse Effect: an event or series of events whether related or not, including any material adverse change in the property or financial condition of a Transaction Party, occurs which has a Material Adverse Effect;
- (j) cross default:
  - (i) Financial Indebtedness (other than Excluded Financial Indebtedness) of a Transaction Party in excess of \$500,000 becomes due for payment before its stated maturity other than by the exercise of an option of the Transaction Party to pay it before its maturity;
  - (ii) a Transaction Party fails to pay when due for payment (or within any applicable grace period) any Financial Indebtedness (other than Excluded Financial Indebtedness) in excess of \$500,000;
  - (iii) an obligation by a person to a Transaction Party to provide financial accommodation or to acquire or underwrite Financial Indebtedness (other than Excluded Financial Indebtedness) in excess of \$500,000 ceases before its stated maturity other than by the exercise of an option of the Transaction Party to cancel that obligation; or
  - (iv) a marketable security issued by a Transaction Party and having a face value over \$500,000 is required to be redeemed or repurchased before its stated maturity other than by the exercise of an option of the issuer to redeem or repurchase;
- k) cessation of business: a Transaction Party ceases or threatens to cease to carry on its business or a substantial part of its business;
- (I) enforcement of other Security: a person who holds a Security over

property of a Transaction Party exercises a right under that Security against the property to recover any money the payment of which is secured by that Security or enforce any other obligation the compliance with which is secured by it;

- (m) undertaking: an undertaking given to the Bank (or its lawyers) by or on behalf of a Transaction Party (or its lawyers) is not honoured in accordance with its terms and if capable of rectification, is not rectified within three Business Days (or any other longer period agreed by the Bank) after its occurrence;
- (n) reduction of capital: if a Transaction Party is a corporation:
  - (i) it reduces or takes any action to reduce its capital other than by the redemption of redeemable preference shares;
  - ii) it passes or takes any action to pass a resolution of the type referred to in section 254N of the Corporations Act;
  - (iii) it:
    - (A) buys or takes any action to buy, or
    - (B) financially assists (within the meaning of section 260A of the Corporations Act) or takes any action to financially assist any person to acquire, shares in itself or in a holding company of it,
- (o) investigation: if a Transaction Party is a corporation, an investigation is instituted under the Corporations Act or other legislation into, or an inspector is appointed to investigate, its affairs, which would have a Material Adverse Effect;
- (p) environmental claim: a Government Body takes any action, there is a legally valid claim or there is a legally enforceable requirement for expenditure or for cessation or alteration of activity under an Environmental Law, which, in the reasonable opinion of the Bank, would have a Material Adverse Effect;
- (q) Management Fees : the aggregate Management Fees paid by the Transaction Parties in a Financial Year exceed the amount specified for that Financial Year under clause 9.5(k) (without the Bank's prior written consent); or
- (r) Trust: if a Transaction Party is a Trustee:
  - (i) the Trustee ceases to be the trustee or the only trustee of the Trust or any action is taken for the removal of the Trustee as trustee of the Trust, or for the appointment of another person as trustee in addition to the Trustee;
  - (ii) an application or order is sought or made in any court, which is not withdrawn or dismissed within ten Business Days, for:
    - (A) the property of the Trust to be administered by the court; or

- (B) an account to be taken in relation to the Trust: or
- (iii) non-compliance by the Trustee with its obligations as trustee under the Trust Deed which has a Material Adverse Effect.

#### 10.2 Effect of Event of Default

- (a) If an Event of Default subsists the Bank may at any time by notice to the Borrower do any or all of the following:
  - (i) cancel Facility: cancel any or all of the Facilities or any part of a Facility, specified in the notice;
  - (ii) accelerate: make so much of the Outstanding Accommodation which is not then immediately due and payable, any unpaid accrued interest or fees and any other money owing by the Borrower to the Bank in relation to the Transaction Documents either:
    - (A) payable on demand; or
  - (B) immediately due for payment;
  - (iii) Not used
  - (iv) Bank Guarantees:
    - (A) by notice to the Borrower require the Borrower to pay immediately to the Bank the aggregate of the Face Values for all Current Bank Guarantees as at the date of the notice, together with any unpaid accrued interest or fees and any other money (including all Indemnity Amounts) owing by the Borrower to the Bank in relation to the Transaction Documents;
    - (B) pay the Beneficiaries of any one or more of the Current Bank Guarantees the amount agreed between the Bank and the relevant Beneficiary sufficient to obtain from the Beneficiary an unconditional release of the Bank's obligations under the relevant Bank Guarantee on terms satisfactory to the Bank (acting reasonably).
  - (v) engage consultants: at the cost of the Borrower, appoint (or require the Borrower to appoint) such accountancy, financial management and other consultants as the Bank may nominate to investigate the business affairs and financial condition of any Transaction Party and whether each Transaction Party has complied with each Transaction Document to which it is a party and to make recommendations relating to the manner in which the Transaction Party carries on its business. Each Transaction Party agrees to provide all assistance and information required by the consultants (including making alfancial records available and giving access to all premises and records) to enable the consultants to conduct their examination promptly, completely and accurately. No Transaction Party is obliged to accept the

recommendations of any consultant, and the Bank will assume no liability with respect to any actions a Transaction Party takes, or does not take, as a result of those recommendations: or

- (vi) treasury related transactions: if there are any Hedging Transactions or treasury related transactions in existence between the Bank and the Borrower (Open Positions ) then:
  - (A) the Bank may close out the Open Positions, by entering into opposite positions for the balance of the unexpired term, or by such other means as may be usual in the relevant market. Any such close out must be at market rates prevailing at the time;
  - (B) any costs incurred by the Bank in closing out Open Positions must be paid by the Borrower to the Bank immediately upon demand by the Bank;
  - (C) any gain derived from the closing out of the Open Positions will be credited to the Borrower and set off against the Amount Owing; and
  - (D) the Bank will give the Borrower reasonable particulars of the manner of close out of the Open Positions and the basis of calculation of any amounts payable by or to the relevant Borrower arising from that close out.
- (b) On receipt of a notice under clause 10.2(a)(ii)(A) or 10.2(a)(ii)(B), the Borrower must immediately pay in full the amounts referred to in that notice.

# 10.3 Cash Cover Account regarding Bank Guarantees

- (a) The Bank must credit so much of the money paid by the Borrower under clause 10.2(a)(iv)(A) which the Bank appropriates towards the Face Values of Current Bank Guarantees to an account maintained by the Bank for this purpose ( Cash Cover Account ).
- (b) The following provisions apply to the Cash Cover Account:
  - (i) the account will be in the name of the Borrower,
  - (ii) despite the Cash Cover Account being in the name of the Borrower, until the Release Date the money held in the account is not owed by the Bank to the Borrower and the Borrower is not entitled to withdraw or be paid any of that money (including interest credited to the account);
  - (iii) the Bank must credit to the account interest at the Cash Cover Rate from time to time and that interest will be credited to the account monthly and on the Release Date; and
  - (iv) without limiting this clause 10.3, the Bank may apply any amounts from time to time held in the account towards payment of any amounts due and payable from time to time to the Bank under any

Transaction Document.

(c) On the Release Date, the Bank must pay to the Borrower the credit balance of the Cash Cover Account.

#### 10.4 Review Events

Each of the following is a Review Event (whether or not caused by anything outside the control of any Transaction Party):

- (a) there is an Insolvency Event in respect of Reading International Inc; and
- (b) a Change of Control occurs in relation to any Transaction Party.

#### 10.5 Reviews

- (a) In addition to any other review rights the Bank has under this document, the Bank may conduct a review of any Facility following a Review Event.
- (b) If a Review Event has occurred, then, at any time or from time to time:
  - (i) the Bank may change any of the conditions applying to the Facility including, but not limited to, increasing or otherwise varying the fees payable in connection with the Facility; and/or
  - (ii) the Bank may terminate the Facility. If the Bank terminates the Facility, the Termination Date occurs on the date 30 days after the date the Bank notifies the Borrower that it wishes to terminate the Facility.
- (c) The Bank may not change any of the conditions applying to the Facility unless it has first given 30 days prior notice to the Borrower of the intended change.
- (d) If the Bank gives notice of any change to the conditions of any Facility and the Borrower refuses to accept the changes before the end of the period of notice, then at the end of that period, the Facility will become repayable within 30 days of any demand by the Bank.
- (e) Nothing in this clause affects the Bank's rights if any Event of Default occurs.

# 11 Costs and expenses

# 11.1 Interpretation

A reference to "costs and expenses" in a Transaction Document includes legal costs and expenses on a full indemnity basis.

#### 11.2 Nature

The Borrower must on demand pay and if paid by the Bank reimburse to the Bank:

- (a) the Bank's reasonable costs and expenses relating to:
  - (i) any Valuation obtained for the purposes of any Transaction

**JA3164** 

#### Document:

- (ii) the negotiation, preparation, execution, stamping and registration of the Transaction Documents or any document contemplated by them;
- (iii) any consent, request for consent (whether or not given), communication or waiver of any right, or the variation, replacement or discharge of any Transaction Document or any document contemplated by it,
- (iv) the enforcement or attempted enforcement or the preservation of any rights of the Bank under the Transaction Documents;
- (v) the occurrence of any Event of Default or Potential Event of Default; and
  - the lodgment or removal of any Encumbrance on the Secured Property by any person; and
- (b) subject to clause 18.14(d), any Taxes and registration or other fees (including fines and penalties relating to the Taxes and fees) which are payable or are assessed by a relevant Government Body or other person to be payable in relation to the Transaction Documents or any document or transaction contemplated by them.

#### 11.3 Remuneration

(vi)

The Bank, any Receiver and any Attorney must be remunerated by the Borrower for any services rendered by them in relation to the enforcement of any right under the Transaction Documents. The rate of the remuneration and the manner of payment will be that determined by the Bank, acting reasonably.

# 12 Indemnities

#### 12.1 Nature

The Borrower indemnifies the Bank on demand against any liability, loss, cost or expense (including Break Costs) caused or contributed to by:

- (a) any failure by any Transaction Party to comply with any obligation under any Transaction Document;
- (b) any Event of Default or Potential Event of Default;
- (c) the enforcement or attempted enforcement of any right by the Bank, any Receiver or any Attorney under the Transaction Documents;
- any Drawing requested by the Borrower not being granted by the Bank for any reason other than a default by the Bank;
- (e) any payment not being made by the Borrower in accordance with any Transaction Document; or
- any act by the Bank in reliance on any communication purporting to be from the Borrower or to be given on behalf of the Borrower.

#### 12.2 Representatives

The Borrower indemnifies each Receiver and Attorney and their respective Representatives and the Representatives of the Bank against any liability, loss, cost and expense caused by anything the Bank is indemnified against under clause 12.1 and the Bank holds the benefit of this clause 12.2 on trust for those persons.

# 12.3 Currency deficiency

If there is any deficiency between:

- an amount payable by a Transaction Party under a Transaction Document which is received by the Bank in a currency other than the currency payable under the Transaction Document because of a judgment, order or otherwise; and
- (b) the amount produced by converting the payment received from the currency in which it was paid into the currency in which it was agreed to be paid either directly or through a currency other than that in which it was agreed to be paid,

the Borrower must pay to the Bank the deficiency and any loss, costs or expenses resulting from it.

#### 12.4 Independence and survival

Each indemnity in a Transaction Document is a continuing obligation, separate and independent from the other obligations of the Borrower and survives the termination of that Transaction Document.

### 12.5 Accounting for transactions

- (a) The Borrower irrevocably authorises the Bank to open such accounts as the Bank requires in connection with a Facility.
- (b) The Borrower irrevocably authorises the Bank to debit from any account in the name of the Borrower (including an account the Bank opens in the Borrower's name) any amounts payable by the Borrower in relation to that Facility or account, including interest, costs, Taxes, enforcement expenses and any amount payable under an indemnity.

  (c) If the Borrower authorises the Bank to debit any amount from an account, the Bank can debit that amount from that account even if it causes the account to become overdrawn. Alternatively, if there are insufficient cleared funds in that account, the Borrower authorises the Bank to debit that amount from any account of the Borrower the Bank decides, including an account the Bank opens in the Borrower's name.
- (d) Where the Bank debits an account in the name of the Borrower, opened by:
  - the Borrower, the Borrower must pay the Bank interest (including default interest if applicable) on any debit balance in accordance with the terms of that account;

- (ii) the Bank, the Borrower must pay the Bank interest on the overdrawn balance of that account at the Overdue Rate applying to the relevant Facility or, if there is none, in accordance with the terms normally applied by the Bank to accounts of that type; or
- (iii) either the Borrower or the Bank, the overdrawn balance of the account in excess of the applicable Facility Limit is immediately payable without further notice.
- (e) Unless otherwise provided, the Bank may apply any payment under or in connection with this document towards satisfying obligations under this document as the Bank sees fit.
- (f) Where the Bank is authorised to debit an amount from an account under this document, it can do so without prior notice

# 12.6 Liability for Regulatory Events

- The Borrower acknowledges that the Services may be interrupted, prevented, delayed or otherwise adversely affected by a Regulatory Event.
- (b) To the extent permitted by Law:
  - (i) the Bank is not liable for any loss incurred by a Borrower or any other person if an event described in clause 12.6(a) occurs, irrespective of the nature or cause of that loss, and the Bank has no obligation to contest any Regulatory Event or to mitigate its impact on the Borrower or the Bank; and
  - (ii) the Borrower releases the Bank from all liability in connection with any loss incurred by a Borrower or any other person if an event described in clause 12.6(a) occurs.
  - To the extent that the Bank's liability cannot be excluded, the Bank's liability is limited to the cost of having the Service supplied again.
- (d) The Bank may use and disclose to any other financial institution or agency, any information about any Borrower, the Services or any person connected with it or the Services, for any purpose which the Bank, or any other financial institution, considers appropriate or necessary in connection with any Regulatory Event or the Services and this may result in information being transmitted overseas.
- (e) The Borrower agrees to provide information to the Bank about it, the Services or any person connected with it or the Services on request, and to promptly procure any consents the Bank requires to give effect to clause 12.6(d).
- 13 Goods and Services Tax
- 13.1 Taxable supply

(a)

(a) If GST is payable by the Bank on any supply made under a Transaction

Document, the Borrower must pay to the Bank an amount equal to the GST payable on the supply.

- (b) That amount must be paid at the same time that the consideration for the supply is to be provided under the Transaction Document and must be paid in addition to the consideration expressed elsewhere in the Transaction Document.
- (c) On receiving that amount from the Borrower, the Bank must provide the Borrower with a tax invoice for the supply.

#### 13.2 Adjustment events

If an adjustment event arises in relation to a supply made by the Bank to the Borrower under a Transaction Document, a corresponding adjustment must be made between the Bank and the Borrower in relation to any amount paid to the Bank by the Borrower under clause 13.1 and payments to give effect to the adjustment must be made.

#### 13.3 Payments

If the Borrower is required under a Transaction Document to pay for or reimburse an expense or outgoing of the Bank or is required to make a payment under an indemnity in relation to an expense or outgoing of the Bank, the amount to be paid by the Borrower is the sum of:

- (a) the amount of the expense or outgoing less any input tax credit in relation to that expense or outgoing that the Bank is entitled to; and
- (b) if the Bank's recovery from the Borrower is in relation to a taxable supply, an amount equal to the GST payable by the Bank in relation to that recovery.

#### 14 Increased costs

If the Bank determines that:

- (a) the cost to it of providing, funding or maintaining the Facility is increased;
- (b) an amount payable to the Bank or the effective return to the Bank under a Transaction Document is reduced;
- the effective return to the Bank under any Transaction Document as a proportion of the capital of the Bank is reduced; or
- (d) the Bank must make a payment or forego any interest or other return calculated by reference to any amount received or receivable by it from any Transaction Party under a Transaction Document,

# because of:

(e) any law, regulation or Government Body directive or request (whether or not having the force of law) introduced or made after the date of this document, including those relating to taxation, capital adequacy or reserve requirements or banking or monetary controls; or

(f) any change in the interpretation or application of any of them.

the Borrower must, within two Business Days after a demand by the Bank, pay to the Bank the amount which, in the Bank's reasonable opinion, will compensate the Bank for the increased cost, reduction, payment or foregone interest or other return.

# 15 Illegality

### 15.1 Prepayment

If because of any change after the date of this document in:

- (a) a law, regulation or a Government Body directive or request which is legally enforceable or compliance with which is in accordance with the practice of responsible Banks in the relevant jurisdiction; or
- (b) the interpretation or application of any of them,

the Bank determines that it is or it will become impossible or illegal or contrary to that Government Body directive or request for:

- (c) the Bank to fund, provide or maintain the Facility or otherwise comply with its obligations under the Transaction Documents; or
- (d) a person from whom the Bank has raised or proposes to raise money in relation to the Facility to fund, provide or maintain that money,

the Borrower must, within five Business Days after receipt of a notice from the Bank to do so, pay the amount referred to in clause 10.2(a)(ii)(A) or 10.2(a)(ii)(B) as if that notice were a notice under clause 10.2(a)(ii)(A) or 10.2(a)(ii)(B).

### 15.2 Facility terminated

The Bank's obligation to make Advances or Drawings under this document terminates on the giving of a notice under clause 15.1.

### 16 Guarantee and indemnity

### 16.1 Guarantee

- (a) Each Guarantor unconditionally and irrevocably guarantees the payment to the Bank of the Guaranteed Money.
- (b) If the Borrower does not pay the Guaranteed Money on time and in accordance with the Transaction Documents, then the Guarantors agree to pay the Guaranteed Money on demand from the Bank.
- (c) A demand may be made at any time and from time to time and whether or not the Bank or the Bank has made demand on the Borrower or any other Transaction Party.

# 16.2 Nature of guarantee

- The guarantee in clause 16.1 is a continuing obligation despite any intervening payment, settlement or other thing and extends to all of the Guaranteed Money. (a)
- As between each Guarantor and the Bank (but without affecting the obligations of any other Transaction Party) each Guarantor is liable under this document in relation to the Guaranteed Money as a sole and principal debtor and not as surety. (b)

### 16.3 Indemnity

- (a) Each Guarantor indemnifies the Bank against any liability or loss arising and any costs it suffers or incurs:
  - (i) if a Transaction Party does not, is not obliged to or is unable to pay the Guaranteed Money in accordance with the Transaction Documents;
  - (ii) if a Guarantor is not obliged to pay the Bank an amount under
  - (iii) if the Bank is obliged, or agrees, to pay an amount to a trustee in bankruptcy or liquidator (of an insolvent person) in connection with a payment by a Transaction Party under or in connection with a Transaction Document; (iv) if a Guarantor defaults under the Guarantee in clause 16.1; or
  - (v) in connection with any person exercising, or not exercising, rights under the Guarantee in clause 16.1. Each Guarantor agrees to pay amounts due under this indemnity immediately on demand from the Bank.

### (b) 16.4 Reinstatement of rights

- (a) Following an Insolvency Event in respect of a Transaction Party, a person may claim that a transaction (including a payment) in connection with this Guarantee or the Guaranteed Money is void or voidable.
- (b) If a claim is made and upheld, conceded or comprised:
  - (i) the Bank is immediately entitled as against the Guarantors to the rights in respect of the Guaranteed Money to which it was entitled immediately before the transaction; and
  - on request from the Bank, each Guarantor agrees to do anything (including signing any document) to restore to the Bank any Security Interest (including this Guarantee) held by it from the Guarantors immediately before the transaction. (ii)

### 16.5 Rights of the Bank are protected

Rights given to the Bank under this Guarantee (and each Guarantor's liabilities under it) are not affected by any act or omission by the Bank or by anything else that might otherwise affect them under law or otherwise, including:

- (a) the fact that it varies or novates any agreement under which the Guaranteed Money is expressed to be owing, such as by increasing the Facility Limit or extending the term;
- (b) the fact that it releases any Transaction Party or gives it a concession, such as more time to pay;
- (c) the fact that a Transaction Party opens an account with it;
  - the fact that it releases, loses the benefit of or does not obtain any Security Interest;
- (e) the fact that it does not register any Security Interest which could be registered;
- (f) the fact that it releases any person who gives a guarantee or indemnity in connection with any Transaction Party's obligations (including under clause 16.13);
- (g) the fact that a person becomes a Guarantor after the date of this document (including under clause 16.14);
- (h) the fact the obligations of any person who guarantees any Transaction Party's obligations (including under this Guarantee) may not be enforceable:
- (i) the fact that any person who was intended to guarantee any Transaction Party's obligations does not do so or does not do so effectively;
- (j) changes in the membership, name or business of any person; or
- (k) the fact that a person who is a co-surety or co-indemnifier for payment of the Guaranteed Money is discharged under an agreement or by operation of law.

# 16.6 No merger

- (a) This Guarantee does not merge with or adversely affect, and is not adversely affected by, any of the following:
  - (i) any other guarantee, indemnity, or Security Interest, or other right or remedy to which the Bank is entitled; or
  - (ii) a judgment which the Bank obtains against the Guarantors in connection with the Guaranteed Money or any other amount payable under this Guarantee.
- (b) The Bank may still exercise rights under this Guarantee as well as under the judgment, other guarantee, indemnity, Security Interest, or other right or remedy.

# 16.7 Extent of Guarantor's obligations

If more than one person is named as "Guarantor", each of them is liable for all the obligations under this Guarantee both individually and jointly with any one or more other persons named as "Guarantor".

#### 16.8 Guarantor's rights are suspended

As long as any of the Guaranteed Money remains unpaid, the Guarantor may not, without the Bank's consent:

- (a) reduce its liability under this Guarantee by claiming that it or any other Transaction Party or any other person has a right of set-off or counterclaim against the Bank;
- (b) exercise any legal right to claim to be entitled to the benefit of another guarantee, indemnity, or Security Interest given in connection with the Guaranteed Money or any other amount payable under this Guarantee;
- (c) claim an amount from another Transaction Party, or another guaranter of the Guaranteed Money (including a person who has signed this document as a "Guaranter"), under a right of indemnity in respect of this guarantee; or
- (d) claim an amount in the insolvency of a Transaction Party or of another guarantor of the Guaranteed Money (including a person who has signed this document as a "Guarantor").

#### 16.9 Guarantor's right of proof limited

Each Guarantor agrees not to exercise a right of proof after an event occurs relating to the insolvency of a Transaction Party or another guarantor of the Guaranteed Money (including a person who has signed this document as a "Guarantor") independently of an attorney appointed under clause 16.12.

#### 16.10 No set-off against assignees

If the Bank assigns or otherwise deals with its rights under this Guarantee, the Guarantors may not claim against any assignee (or any other person who has an interest in this Guarantee) any right of set-off or other right the Guarantors have against the Bank.

#### 16.11 Suspense account

The Bank may place in a suspense account any payment it receives from the Guarantors if there is currently an Insolvency Event, or an Insolvency Event is likely to occur, in relation to any Transaction Party, but must apply it towards satisfying the Guaranteed Money within six months unless the winding up of the relevant Guarantor has commenced.

#### 16.12 Right to prove

- (a) The Guarantor irrevocably appoints the Bank and each of its Authorised Representatives individually as its attorney and agrees to formally approve all action taken by an attorney under this clause 16.
- Each attorney may, at any time while any Guaranteed Money is outstanding:
  - (i) do anything which a Guarantor may lawfully do to exercise their right of proof in respect of a Transaction Party after an Insolvency Event occurs in respect of such Transaction Party. These things

may be done in the Guarantor's name or the attorney's name and they include signing and delivering documents, taking part in legal proceedings and receiving any dividends arising out of the right of proof;

- (ii) delegates its powers (including this power) and may revoke a delegation; and
- (iii) exercise its powers even if this involves a conflict of duty and even if it has a personal interest in doing so.
- (c) The attorney need not account to a Guarantor for any dividend received on exercising the right of proof under clause 16.12(i) except to the extent that any dividend remains after the Bank has received all of the Guaranteed Money and all other amounts payable under the Guarantee.

#### 16.13 Release of Guarantors

- (a) The Bank must, at the Borrower's cost, execute any release documentation in respect of the Bank's rights under clause 16.
- (b) As between the Transaction Parties and the Bank, the Bank is not obliged to consent to a release unless required to do by the terms of another Transaction Document.
- (c) The rights and obligations of the remaining Guarantors under the Guarantee in clause 16.1 will continue in full force and effect despite the release of a Guarantor under this clause 16.13.

#### 16.14 New Guarantors

If a Subsidiary of any Transaction Party is required by the terms of a Transaction Document to become a Guarantor, the Borrower must ensure that such subsidiary executes a Guarantor Accession Deed as a new Transaction Party.

#### 16.15 Consideration

Each Guarantor acknowledges having executed this document in return for the Bank entering into the Transaction Documents at the request of the Guarantor and other valuable consideration.

#### 16.16 New Guarantors

- (a) A person automatically becomes a party to this document as a Guarantor and Transaction Party (after the date of this document) by signing and delivering to the Bank a Guarantor Accession Deed and doing anything else which the Bank reasonably requests to ensure the enforceability of that person's obligations as a Guarantor.
- (b) Each of the other parties to his document irrevocably appoints the Bank as its agent to sign on its behalf any Guarantor Accession Deed.
- (c) The execution of a Guarantor Accession Deed will not operate to release any party from its obligations under any Transaction Document.

#### 17 Attorney

#### 17.1 Appointment

If and for so long as an Event of Default occurred and is continuing, the Borrower irrevocably appoints the Bank its attorney with the power:

- (a) at any time to:
  - (i) do everything which in the Attorney's reasonable opinion is necessary or expedient to enable the exercise of any right of the Bank in relation to the Transaction Documents;
  - (ii) not used;
  - (iii) complete the Transaction Documents to which it is a party; and
  - (iv) appoint its directors, officers, employees and solicitors as substitutes and otherwise delegate its powers to any of them (except this power of delegation); and
- (b) at any time after a notice is given under clause 10.2(a)(ii)(A) or 10.2(a)(ii)(B), to do all acts and things which the Borrower is obliged to do under the Transaction Documents or which in the Attorney's opinion are necessary or expedient to enable the exercise of any right of the Bank in relation to the Transaction Documents.

#### 17.2 Not used

#### 17.3 General

- (a) Any Attorney may exercise any right solely for the benefit of the Bank, even if the exercise of the right constitutes a conflict of interest or duty.
- (b) The Borrower by this document ratifies anything done or not done by the Attorney pursuant to the power of attorney.
- (c) The power of attorney is granted:
  - (i) to secure the compliance by the Borrower with its obligations to the Bank under the Transaction Documents and any proprietary interests of the Bank under the Transaction Documents; and
  - (ii) for valuable consideration (receipt of which is acknowledged) which includes entry into of this document by the Bank at the Borrower's request.

#### 18 General

#### 18.1 Set-off

The Bank may set off any money due for payment by the Bank to the Borrower, whatsoever, including any money in any currency held by the Bank for the account of the Borrower in any place, against any money due for payment by the Borrower to the Bank under a Transaction Document.

#### 18.2 Bank's certificate

- (a) A certificate by the Bank relating to any amount owing under a Transaction Document or as to its opinion in relation to any matter under any Transaction Document is prima facie evidence against the Borrower of the matters certified unless proven incorrect or there is a manifest error.
- (b) The Bank is not obliged to give the reasons for its determination or opinion in relation to any matter under any Transaction Document. Any certification, determination or opinion relating to an amount must contain reasonable detail as to how the amount was calculated.
- (c) A determination or an opinion of an Authorised Representative of the Bank which is given to the Borrower or otherwise expressed or acted on by the Bank as being a determination or an opinion of the Bank will be deemed to be a determination or opinion of the Bank.

#### 18.3 Supervening legislation

Any present or future legislation which operates:

- (a) to lessen or vary in favour of the Borrower any of its obligations in connection with the Transaction Documents; or
- (b) to postpone, stay, suspend or curtail any rights of the Bank under the Transaction Documents,

is excluded except to the extent that its exclusion is prohibited or rendered ineffective by law.

#### 18.4 Time of the essence

Time is of the essence as regards any obligations of the Borrower or any date or period determined under the Transaction Documents, and if any date or period is altered by agreement between the parties, time is of the essence as regards such altered date or period.

#### 18.5 Business Days

- (a) If the day on or by which anything, other than making a payment, must be done by the Borrower under a Transaction Document is not a Business Day, that thing must be done on or by the preceding Business Day.
- (b) If a payment would otherwise be due on a day which is not a Business Day it will be due on the immediately following Business Day. However,

if this would result in the payment being due in the month after the original due day or after the Termination Date it will be due on the immediately preceding Business Day.

- If anything, including making a payment, is to be done by the Borrower on or by a particular day and it is done:

  (i) after the time by which a Transaction Document states it must be done or, if the Transaction Document does not state a time, after 4.00 pm in the place where it is to be done; or
  - (ii) on a day which is not a Business Day,
  - it will be deemed to have been done at 9.00 am on the next Business Day.

#### 18.6 Confidentiality

- The Bank must keep any information or document relating to a Transaction Party confidential. However, the Bank may disclose to any person any information or document relating to a Transaction Party:
  - (i) where permitted in a Transaction Document;
  - (ii) to another party to a Transaction Document;
  - (iii) to a potential transferee, assignee, participant or sub-participant of the Bank's interests under a Transaction Document or to any other person who is considering entering into contractual relations with it in connection with a Transaction Document;
  - (iv) to the Bank's related bodies corporate and shareholders, or to any employee, banker, lawyer, auditor or other consultant of the Bank, its related bodies corporate or its shareholders;
  - (v) to the professional advisers or consultants of any party involved in connection with any Facility who are bound by a duty or obligation of confidence;
  - (vi) if required by law or by any Government Body or stock exchange;
  - (vii) in connection with any legal proceedings relating to a Transaction Document or a document delivered under or in relation to a Transaction Document;
  - (viii) if the information or document is in the public domain; or
  - (ix) with the consent of the Borrower (which must not be unreasonably withheld or delayed).
- (b) Subject to paragraph (c), the Transaction Parties shall keep confidential and not disclose to any other person the terms of the Transaction Documents.
- (c) However, the Transaction Parties and any officers or employees of each Transaction Party may disclose such information:
  - (i) with the prior written consent of the Bank;

- (ii) to the extent required by any applicable law or regulation;
- (iii) to the extent it reasonably deems necessary in connection with any actual or contemplated proceedings or a claim with respect to this clause 18.6; or
- (iv) to the extent permitted by clause 18.6(a) (other than paragraph (iii)) as if each reference in that clause to the 'Bank' were to a 'Transaction Party' and each reference to the 'Borrower' were to the 'Bank'.
- (d) The Bank and the Transaction Parties agree that:
  - neither of them will disclose information of the kind mentioned in section 275(1) of the PPS Act; and
  - (ii) this document does not create a Security Interest.
- (e) This clause 18.6 survives the termination of this document.
- (f) The Bank acknowledges that:
  - (i) information provided from time to time by the Transaction Parties to the Bank may constitute confidential non-public information; and
  - (ii) trading in marketable securities of Reading International Inc while in possession of the information referred to clause 18.6(f)(i) will violate United States' federal securities
- (g) The Bank agrees to:
  - (i) take reasonable precautions to maintain the confidentiality of the information referred to in clause 18.6(f)(i); and
  - (ii) advise any party to whom the information referred to in clause 18.6(f)(i) is disclosed that it may not trade in the marketable securities of Reading International Inc while in the possession of such information.
- (h) This clause 18.6 will not be deemed to restrict the provision of information by any party to the Internal Revenue Service of the United States of America.

#### 18.7 Exchange rate

Subject to any express provision to the contrary, if for the purposes of a Transaction Document it is necessary to convert one currency into another currency, the conversion must be effected using an exchange rate selected by the Bank acting reasonably and in accordance with it usual practices.

#### 18.8 Records as evidence

The Bank may maintain records specifying:

- (a) payments made by the Bank for the account of a Transaction Party under a Transaction Document;
- (b) payments by a Transaction Party for the account of the Bank under a Transaction Document; and

**JA3177** 

- (c) interest, fees, charges, costs and expenses payable in relation to the Transaction Documents.
- and those records will against the Borrower constitute prima facie evidence of the matters set out in them.

#### 18.9 Further assurances

The Borrower must promptly execute all documents and do all things that the Bank from time to time reasonably requires to:

- (a) effect, perfect or complete the provisions of each Transaction Document or any transaction contemplated by it;
- (b) establish the priority of or reserve or create any Security Interest contemplated by or purported to be reserved or created by a Transaction Document; and
- (c) stamp and register each Transaction Document in any relevant jurisdiction and by any person that the Bank thinks fit.

#### 18.10 Amendment

This document may only be varied or replaced by a document executed by the parties.

#### 18.11 Waiver and exercise of rights

- (a) A right in favour of the Bank under a Transaction Document, a breach of an obligation of the Borrower under a Transaction Document or an Event of Default can only be waived by an instrument signed by the Bank. No other act, omission or delay of the Bank constitutes a waiver binding, or estoppel against, the Bank.
- (b) A single or partial exercise or waiver by the Bank of a right relating to a Transaction Document does not prevent any other exercise of that right or the exercise of any other right.
- (c) The Bank and its Representatives are not liable for any loss, cost or expense of the Borrower caused or contributed to by the waiver, exercise, attempted exercise, failure to exercise or delay in the exercise of a right and the Bank holds the benefit of this clause 18.11 on trust for itself and its Representatives.

#### 18.12 Rights cumulative

The rights of the Bank under the Transaction Documents are cumulative and in addition to its other rights.

#### 18.13 Approval and consent

Except where a Transaction Document expressly provides otherwise, the Bank may conditionally or unconditionally give or withhold any consent under a Transaction Document and is not obliged to give its reasons for doing so.

#### 18.14 Assignment

(a) The Borrower must not dispose of or Encumber any right under the

Transaction Documents without the consent of the Bank.

- (b) The Bank may assign any of its rights or novate, sub-participate, sell-down or transfer by whatever form or otherwise deal with any or all of its rights and obligations under any Transaction Document without the consent of, or notice to, the Borrower.
- (c) If an Event of Default subsists then, in order to facilitate the Bank to deal with its rights and obligations, the Bank may (but is not obliged to), from time to time, separate and sever any of its rights (or any part of any of its rights) described in a notice given by the Bank to the Borrower from its other rights and obligations under any Transaction Document. Any such notice is effective on the time of delivery to separate and sever the rights described in the notice so that:
  - (i) those rights and obligations are independent from, and may be assigned (including at law), novated, sub-participated, sold-down, transferred or otherwise dealt with separately from, any other of the rights and obligations of the Bank under that Transaction Document;
  - (ii) those rights and obligations may be exercised differently from any other rights and obligations of the Bank under that Transaction Document; and
  - (iii) the Outstanding Accommodation in respect of those rights may be calculated separately from the other Outstanding Accommodation.
- (d) If the Bank assigns its rights or transfers its rights and obligations under this document or any other Transaction Document, no Transaction Party will be required to pay any net increase in the aggregate amount of costs, Taxes, fees or charges which is a direct consequence of the assignment or transfer.

#### 18.15 Counterparts

This document may consist of a number of counterparts and, if so, the counterparts taken together constitute one document.

#### 18.16 Sovereign immunity

The Borrower irrevocably waives any immunity that it or its property has from:

- (a) set-off;
- (b) legal, arbitral or administrative proceedings;
- (c) any process or order of any court, administrative tribunal or arbitrator for the satisfaction or enforcement of a judgment, order or arbitral award or for the arrest, detention or sale of any property; or
- (d) service on it of any process, judgment, order or arbitral award,

on the grounds of sovereignty or otherwise under any law of any jurisdiction where any proceedings may be brought or enforced in relation to any Event of Default under a Transaction Document.

#### 18.17 Governing law and jurisdiction

- (a) This document is governed by and is to be construed in accordance with the laws applicable in the Relevant Jurisdiction.
- (b) Each party irrevocably and unconditionally submits to the non-exclusive jurisdiction of the courts exercising jurisdiction in the Relevant Jurisdiction and any courts which have jurisdiction to hear appeals from any of those courts and waives any right to object to any proceedings being brought in those courts.

#### 18.18 Telephone recording

The Borrower consents to the Bank recording any telephone conversations between it and the Bank in relation to any Facility that are customarily recorded in the finance industry or where the Borrower is notified prior to the commencement of the telephone conversation and such recordings being used in any arbitral or legal proceedings and any telephone recording remains the Bank's sole property at all times.

#### 18.19 Legal advice

The Borrower acknowledges that, except as expressly set out in a Transaction Document:

- (a) none of the Bank or any of its advisers have given any representation or warranty or other assurance to it in relation to any Transaction Document or the transactions contemplated by any Transaction Document, including as to Tax or other effects;
- (b) It has not relied on the Bank or any of its advisers or on any conduct (including any recommendation) by the Bank or any of its advisers; and
- (c) it has obtained its own independent financial, Tax and legal advice.

#### 18.20 Further assurances

Whenever the Bank requests a Transaction Party to do anything:

- (a) to ensure any Transaction Document (or any security interest (as defined in the PPS Act) or other Security Interest, right or power under any Transaction Document) is fully effective, enforceable and perfected with the contemplated priority;
- (b) for more satisfactorily assuring or securing to the Bank the property the subject of any such security interest or other Security in a manner consistent with the Transaction Documents; or
  - for aiding the exercise of any right or power in any Transaction Document,

the Transaction Party shall do it promptly at its own cost. This may include obtaining consents, getting documents completed and signed, supplying information, delivering documents and evidence of title and executed blank transfers, and giving possession or control with respect to any Secured Property.

#### 18.21 Exclusion of certain provisions

Where there is a Security Interest under any Transaction Document:

- (a) to the extent permitted, sections 142 and 143 of the PPS Act are excluded in full and will not apply to that Security Interest and the Bank need not comply with sections 95, 118, 121(4), 125, 130, 132(3)(d), and 132(4) of the PPS Act; and
- (b) each Transaction Party waives its right to receive from the Bank any notice required under s157 of the PPS Act or the provisions of the PPS Act referred to in s144 of the PPS Act, except section 135.

This does not affect any rights a person has or would have other than by reason of the PPS Act and applies despite any other clause in any Transaction Document.

#### 18.22 Notice of changes

Each Transaction Party agrees to notify the Bank at least 14 days before:

- (a) a Transaction Party (or if the Transaction Party is trustee of a Trust or a partner of a partnership, the Trust or the partnership) changes its name;
- (b) any ABN, ARBN or ARSN allocated to a Transaction Party (or if the Transaction Party is trustee of a Trust or a partner of a partnership, the Trust or the partnership) changes, is cancelled or otherwise ceases to apply to it (or if it does not have an ABN, ARBN or ARSN, one is allocated, or otherwise starts to apply, to it), or
- (c) the Borrower becomes trustee of a trust, or a partner in a partnership, which is not expressly contemplated in the Transaction Documents.

#### 19 Notices

#### 19.1 General

A notice, demand, certification, process or other communication relating to a Transaction Document must be in writing in English and may be given by an Authorised Representative of the sender.

#### 19.2 How to give a communication

In addition to any other lawful means, a communication may be given by being:

- (a) personally delivered;
- (b) left at the party's current address for notices;
- (c) sent to the party's current address for notices by pre-paid ordinary mail or, if the address is outside Australia, by pre-paid airmail; or
- (d) sent by fax to the party's current fax number for notices.

**JA3181** 

#### 19.3 Particulars for delivery of notices

(a) The particulars for delivery of notices are initially:

#### Transaction Parties:

As set out in schedule 1. Bank:

Address: Level 28, 500 Bourke Street, Melbourne, Victoria 300

Fax: 1300 889 390

Attention: Andrew Tham

(b) Each party may change its particulars for delivery of notices by notice to each other party.

#### 19.4 Communications by post

Subject to clause 19.6, a communication is given if posted:

- (a) within Australia to an Australian address, three Business Days after posting; or
- (b) in any other case, ten Business Days after posting.

#### 19.5 Communications by fax

Subject to clause 19.6, a communication is given if sent by fax when the sender's fax machine produces a report that the fax was sent in full to the addressee. That report is conclusive evidence that the addressee received the fax in full at the time indicated on that report.

#### 19.6 After hours communications

If a communication is given:

- (a) after 5.00 pm in the place of receipt; or
- (b) on a day which is a Saturday, Sunday or bank or public holiday in the place of receipt,

#### it is taken as having been given at 9.00 am on the next day which is not a Saturday, Sunday or bank or public holiday in that place.

#### 19.7 Process service

Any process or other document relating to litigation, administrative or arbitral proceedings relating to a Transaction Document may be served on a party to this document by any method contemplated by this clause 19.7 or in accordance with any applicable law.

#### READING INTERNATIONAL, INC. - LIST OF SUBSIDIARIES

#### Subsidiary (Jurisdiction of Incorporation)

READING INTERNATIONAL, INC. – LIST OF S
Subsidiary (Jurisdiction of Incorporation)
A.C.N. 143 633 96 Pty Ltd (Australia)
AHGP, Inc. (Delaware)
Angellia Film Center, LLC (Delaware)
Angellia Film Center Mosilos, LLC (Nevada)
Angellia Film Center (Dalla), Inc. (Texas)
Angellia Film Center (Dalla), Inc. (Texas)
Angellia Film Center (Plano) LP (Nevada)
Australia Country Cinemas Pty Ltd (Australia)
Bayou Cinemas LP (Delaware)
Bogart Hoflings Ltd (New Zealand)
Burwood Developments Pty Ltd (Australia)
Bayou Cinemas LP (Delaware)
Dogart Hoflings Ltd (New Zealand)
Cindel Agricultur, Loc. (Callfornia)
Cindel Cinemas (Nevada)
Cindel Agricultur, Loc. (Nevada)
Cindel Cinemas (Ltc (Nevada)
Consolidated Cinemas Repole, LtC (Nevada)
Darmelle Enterprise Limited (New Zealand)
Dierro Street Hospitality, Lt. (Celaware)
Lepon Cinemas LtC (Nevada)
Liberty Theatres, LtC (Nevada)
Liberty Theatres, LtC (Nevada)
Liberty Theatres, LtC (Nevada)
Liberty Theatres, LtC (Nevada)
Movieland Cinemas (RDJ Ltd (New Zealand)
Newmarket Properties N. 2 Pty Ltd (Australia)
Newmar

RDI Employee Investment Fund LLC (California)
Reading Arthouse Limited (New Zealand)
Reading Authum Ply Ld (Australia)
Reading Authum Ply Ld (Australia)
Reading Authum Ply Ld (Australia)
Reading Cantonia Leaning (ReiK) Ply Ld (Australia)
Reading Cantonia Leaning (ReiK) Ply Ld (Australia)
Reading Canton Park Ply Ld (Australia)
Reading Cantell Corporation (Delaware)
Reading Cantell Corporation (Delaware)
Reading Cinemas Courteny Central Ldd (New Zealand)
Reading Cinemas Management Ply Ldd (Australia)
Reading Cinemas Management Ply Ldd (Australia)
Reading Cinemas of Parto Rico, Inc. (Puerto Rico)
Reading Cinemas Ply Ldd (Australia)
Reading Cinemas Ply Ldd (Australia)
Reading Cinemas Ply Ldd (Australia)
Reading Concilidated Holdings, (Hawaii), Inc. (Hawaii)
Reading Concilidated Holdings, (Hawaii), Inc. (Hawaii)
Reading Concilidated Holdings, (Hawaii), Inc. (Hawaii)
Reading Concilidated Holdings, Inc. (Newada)
Reading Concilidated Holdings, Inc. (Newada)
Reading Danchelm Ply Ldd (Australia)
Reading Danchelm Ply Ldd (Australia)
Reading Danchelm Ply Ldd (Australia)
Reading Entertainment Australia Ply Lld (Australia)
Reading International Services Company (California)
Reading International Services Company (California)
Reading Management NZ Limited (New Zealand)
Reading Monagement NZ Limited (New Zealand)
Reading Monagement NZ Limited (New Zealand)
Reading Properties Lld (New Zealand)
Reading Properties New Zealand Lld (New Zealand)
Reading Reading Properties Lld (Ne

Rialto Cinemas Ltd (New Zealand)
Rialto Distribution Ltd (New Zealand)
Rialto Distribution Ltd (New Zealand)
Rialto Entertainment Ltd (New Zealand)
Reec Ltd. (Rew Zealand)
Reec Ltd. (Remsylvania)
Rydal Equipment Co. (Pennsylvania)
S Note Liquidation Company, Ltd. (Newada)
Salia Apartment Marangement Ltd (New Zealand)
Salia Apartments Marangement Ltd (New Zealand)
The Port Reading Railtond Company (New Jersey)
The Theatre At Legacy LP (Texas)
Tobrooke Hoblings Ltd (New Zealand)
Trans-Pacific Finance Fund I, Ltd. (Delaware)
Trenton-Princeton Traction Company (New Jersey)
Twin Ctica Cinemas, Inc. (Delaware)
US Development Ltd. (Newada)
US Development Ltd. (Newada)
Washington and Frendilin Railway Company (Pennsylvania)
Westlakes Cinema Per Ltd (Australia)
Westlakes Cinema Per Ltd (Australia)
Wilmington and Northern Railroad Company (Pennsylvania)

Consent of Independent Registered Public Accounting Firm

We have issued our reports dated April 29, 2016, with respect to the consolidated financial statements, schedule and internal control over financial reporting included in the Annual Report of Reading International, Inc. on Form 10-K for the year ended December 31, 2015. We hereby consent to the incorporation by reference of said reports in the Registration Statements of Reading International, Inc. on Forms S-8 (File No. 333-112669, effective January 21, 2004, File No. 333-167101, effective May 26, 2010) and on Form S-3 (File No. 333-162581, effective Cetober 20, 2009).

/s/ GRANT THORNTON LLP Los Angeles, California April 29 , 2016

#### CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

#### I. Ellen M. Cotter , certify that:

- 1) I have reviewed this Form 10-K of Reading International, Inc.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4) The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5) The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's Board of Directors (or persons performing the equivalent functions):
  - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial
  - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ Ellen M. Cotter

Preside nt and Chief Executive Officer

(Principal Executive Officer)

April 29, 2016

#### CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

#### I. Devasis Ghose, certify that:

- 1) I have reviewed this Form 10-K of Reading International, Inc.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4) The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5) The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's Board of Directors (or persons performing the equivalent functions):
  - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ Devasis Ghose Devasis Ghose Chief Financial Officer and Treasurer

Chief Financial Officer and Treasure (Principal Financial Officer) April 29, 2016

2450AC950CAC010AC

EXHIBIT 32.1

CERTIFICATION PURSUANT TO SECTION 996 OF THE SARBANES-OXLEY ACT OF 2002

[18 U.S.C. SECTION 1350]

In connection with the accompanying Annual Report of Reading International, Inc. (the "Company") on Form 10-K for the fiscal year ended December 31, 2015 (the "Report"), I, Ellen M. Cotter, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that:

- 1. The Report fully complies with the requirements of Section 13(a) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Ellen M. Cotter Ellen M. Cotter President and Chief Ex ecutive Officer (Principal Executive Officer) April 29, 2016

CERTIFICATION PURSUANT TO SECTION 996 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the accompanying Annual Report of Reading International, Inc. (the "Company") on Form 10-K for the fiscal year ended December 31, 2015 (the "Report"), I, Devasis Ghose, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that:

- 1. The Report fully complies with the requirements of Section 13(a) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Js/ Devasis Ghose Devasis Ghose Chief Financial Officer (Principal Financial Officer) April 29 , 20 16

## **JA3191**

# Confidential - Filed Under Seal

**EXHIBIT 2** 

## **JA3192**

# Confidential - Filed Under Seal

**EXHIBIT 3** 

# **EXHIBIT 4**

#### UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

### FORM 8-K

### CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported):		March 10, 2016	
Re	eading International, Inc.		
7,000	f registrant as specified in its cl	narter)	
	4 (m/a **		
Nevada	1-8625	95-3885184	
(State or other jurisdiction of incorporation)	(Commission File Number)	(IRS Employer Identification No.)	
6100 Center Drive, Suite 900, Los Angeles, California		90045	
(Address of principal ex	recutive offices)	(Zip Code)	
Registrant's telephone number, including area code:		(213) 235-2240	
	Not applicable.		
(Former name or fo	rmer address, if changed since	last report.)	
Check the appropriate box below if the I obligation of the registrant under any of the		o simultaneously satisfy the filing	
[] Written communications pursuant to Rul	e 425 under the Securities Act	(17 CFR 230.425)	
[] Soliciting material pursuant to Rule 14a-	12 under the Exchange Act (17	CFR 240.14a-12)	
[] Pre-commencement communications pur 2(b))	rsuant to Rule 14d-2(b) under the	ne Exchange Act (17 CFR 240.14d-	
[ ] Pre-commencement communications pur 4(c))	rsuant to Rule 13e-4(c) under the	ne Exchange Act (17 CFR 240.13e-	

#### Item 1.01 Entry into a Material Definitive Agreement.

#### New Compensatory Arrangements for Executive and Management Employees

See Item 5.02 below with respect to certain new compensation arrangements for executive and management employees and outside directors of Reading International, Inc. ("Reading," "Registrant" or the "Company").

#### Amendment to 2010 Stock Incentive Plan

On March 10, 2016, Reading's Board of Directors approved an amendment to the 2010 Stock Incentive Plan to permit the award of restricted stock units.

The foregoing description of the amendment to the 2010 Stock Incentive Plan is qualified in its entirety by reference to the provisions of the amendment to the 2010 Stock Incentive Plan as exhibit 10.1 to this Current Report on Form 8-K, which is incorporated herein by reference.

## Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers

#### Item 5.02 (c)

#### Andrzej Matyczynski

On March 10, 2016, the Company's Board of Directors (the "Board") appointed Andrzej Matyczynski, 63, as Executive Vice President—Global Operations.

From May 11, 2015 until March 10, 2016, Andrzej Matyczynski has acted as corporate advisor to the Company. Mr. Matyczynski served as our Chief Financial Officer and Treasurer from November 1999 until May 11, 2015 and Corporate Secretary from May 10, 2011 to October 20, 2014. Prior to joining our Company, he spent 20 years in various senior roles throughout the world at Beckman Coulter Inc., a U.S. based multi-national. Mr. Matyczynski earned a Master's Degree in Business Administration from the University of Southern California.

See Item 5.02(e) below with respect to the compensation arrangements for Mr. Matyczynski.

#### **Margaret Cotter**

On March 10, 2016, the Board appointed Margaret Cotter, 48, as Executive Vice President-Real Estate Management and Development-NYC.

Margaret Cotter has been a Director of the Company since September 27, 2002, and on August 7, 2014 was appointed Vice Chairperson of our Board. Ms. Cotter is the owner and President of OBI, LLC ("OBI"), which has, since 2002, managed our live-theater operations. Pursuant to the OBI management arrangement, Ms. Cotter also served as the President of Liberty Theaters, LLC, the subsidiary through which we own our live theaters. Operating and overseeing these properties for over 16 years, Ms. Cotter contributes to the strategic direction for our developments. Until her appointment on March 10, 2016, while she received management fees through OBI, Ms. Cotter received no compensation for her duties as President of Liberty Theaters, LLC, other than the right to participate in our Company's medical insurance program. Ms. Cotter, through OBI and Liberty Theaters, LLC, managed the real estate which houses each of our four live theaters in Manhattan and Chicago. Based in New York, Ms. Cotter secures leases, manages tenancies, oversees maintenance and regulatory compliance of these properties and heads up the re-development process with respect to these properties and our Cinemas 1, 2 & 3 property. Ms. Cotter is also a theatrical producer who has produced shows in Chicago and New York and a board member of the League of Off-Broadway Theaters and Producers. Ms. Cotter, a former Assistant District Attorney for King's County in Brooklyn, New York, graduated from Georgetown University and Georgetown University Law Center. She is the sister of Ellen M. Cotter, a director and our President and Chief Executive Officer, and James J. Cotter, Jr., a director. Ms. Margaret Cotter is a Co-Executor of her father's estate, which is the record owner of 427,808 shares of our Class B Voting Stock (representing 25.5% of such Class B voting Stock). Ms. Margaret Cotter is also a Co-Trustee of the James J. Cotter, Sr. Trust, which is the record owner of 696,080 shares of Class B Voting Common Stock (representing an additional 44.0% of such Class B Stock). In addition, with her direct ownership of 804,173 shares of Class A Stock and 35,100 shares of Class B Stock and her positions as Co-Executor of her father's estate and Co-Trustee of the James J. Cotter, Sr. Trust, Ms. Cotter is a significant stockholder in our Company.

In connection with her appointment and employment as Executive Vice President of the Company, the Company's Audit and Conflicts Committee authorized the mutual termination of the Theater Management Agreement dated January 1, 2002, between the Company's subsidiary, Liberty Theaters, Inc. (predecessor to Liberty Theaters, LLC) and OBI, LLC, an entity wholly-owned by Ms. Cotter, (the "Theater Management Agreement"). The termination agreement is currently being negotiated by OBI, LLC and Liberty Theaters, LLC and finalized, will be filed on Form 8-K. While Ms. Cotter is the President of Liberty Theaters, LLC, Liberty Theaters, LLC is being separately represented in these negotiations and the final termination agreement will be subject to the review and approval of our Audit and Conflicts Committee.

The Compensation Committee and the Audit and Conflicts Committee each approved additional consulting fee compensation to Margaret Cotter totaling \$200,000 for services rendered by her to the Company in recent years outside of the scope of the Theater Management Agreement, including, but not limited to: (i) predevelopment work on the Company's Union Square and Cinemas 1, 2 & 3 properties, (ii) management of the New York properties, and (iii) management of Union Square tenant matters. The Compensation Committee also noted, when considering this additional consulting fee, that OBI, LLC had agreed to include as a part of its termination agreement with the Company certain waivers and releases including the termination of any rights it might have to receive compensation with respect to any show continuing at any of our theaters after the date of such termination.

The Theater Management Agreement generally provided for the payment of a combination of fixed and incentive fees for the management of our four live theaters. Historically, these fees have equated to approximately 21% of the net cash flow generated by these properties. We currently estimate that fees to be paid to OBI for 2015 will be approximately \$390,000. We paid \$397,000 and \$401,000 in fees with respect to 2014, and 2013, respectively. We also reimbursed OBI for certain travel expenses.

As Executive Vice-President Real Estate Management and Development - NYC, Ms. Cotter will continue to be responsible for the management of our live theater assets and business, will continue her role heading up the pre-redevelopment of our New York Properties and will become our senior executive responsible for the actual redevelopment of our New York properties.

Ms. Cotter's compensation as Executive Vice-President was set as part of the extensive executive compensation process described in Item 5.02(e) below. For 2016, Ms. Cotter's base salary will be \$350,000, she will have a short term incentive target bonus opportunity of \$105,000 (30% of her base salary), and she was granted a long term incentive of a stock option for 19,921 shares of Class A common stock and 4,184 restricted stock units under the Company's 2010 Stock Incentive Plan, as amended, which long term incentives vest over a four year period.

#### Item 5.02(e)

#### **Compensation Arrangements**

#### Background

The Executive Committee ("Executive Committee") of the Board of Directors (the "Board"), upon the recommendation of our Chief Executive Officer, requested the Compensation Committee to evaluate the Company's compensation policy for executive officers and outside directors and to establish a plan that encompasses sound corporate practices consistent with the best interests of the Company. The Compensation Committee undertook to review, evaluate, revise and recommend the adoption of new compensation arrangements for executive and management officers and outside directors of the Company. In January 2016, the Compensation Committee retained the international compensation consulting firm of Willis Towers Watson as its advisor in this process and also relied on the Company's legal counsel, Greenberg Traurig, LLP.

Going forward, the Board of Directors has adopted a formal charter for our Compensation Committee a copy of which has been posted on our website, www.ReadingRDI.com.

#### **Executive Compensation**

From late January to late February 2016, the Compensation Committee met five separate times with Willis Towers Watson, the Chief Executive Officer, and legal counsel. Except for the first meeting, each meeting exceeded three hours and was fully focused on the assessments

**JA3197** 

discussed above. As part of its engagement Willis Towers Watson reviewed the Company's compensation paid to executive and management officers by position, in light of each person's duties and responsibilities. Willis Towers Watson then compared the top executive and management positions at the Company to (i) executive compensation paid by a peer group and (ii) two surveys, the 2015 Towers Watson Data Services Top Management Survey Report and the 2015 Mercer MBD Executive Compensation Survey, in each case, identified by office position and duties performed by the officer. The peer group utilized by Willis Towers Watson included the following companies:

Arcadia Realty Trust
Associated Estates Realty Corp.
Carmike Cinemas Inc.
Cedar Realty Trust Inc.
Charter Hall Group
EPR Properties
Vicinity Centres
IMAX Corporation

Inland Real Estate Corp.
Kite Realty Group Trust
Marcus Corporation
Pennsylvania Real Estate Investment Trust
Ramco-Gershenson Properties Trust
Urstadt Biddle Properties Inc.
Village Roadshow Ltd.

Willis Towers Watson selected the above peer group because (i) the companies included US and Australian based companies reflecting the Company's geographic operations and (ii) the companies were comparable to the Company based on revenue.

The executive pay assessment prepared by Willis Towers Watson measured the executive and management compensation paid by the Company against compensation paid by peer group companies and the companies listed in the two surveys based on the 25 th, 50 th and 75 th percentile of such peer group and surveyed companies. The 50 th percentile was the median compensation paid by such peer group and surveyed companies to executives performing similar responsibilities and duties.

In its report to the Compensation Committee, Willis Towers Watson noted that for Company executive officers:

- Base salaries in the aggregate were generally in the competitive zone of the market (1% below the market 50 th percentile), with certain notable exceptions on position by position review;
- Total cash compensation (base salary and cash bonus) in the aggregate was 26% below the 50 th percentile; and
- Total compensation (base salary, cash bonus and long term incentive awards) in the aggregate was 40% below the 50 <sup>th</sup> percentile.

The Willis Towers Watson assessment compared the base salary, the short term incentive (cash bonus) and long term incentive (equity awards) of the peer and surveyed companies to the base salary, short term incentive and long term incentive provided to executives by the Company. The assessment concluded that except in a few positions, the Company was generally competitive in base salary, however, the Company was not competitive when short term incentives and long term incentives were included in the total compensation paid to the Company

executives and management. The base salary paid by the Company to the following officers was below the 25 <sup>th</sup> percentile: President and CEO. The total compensation (base salary, short term incentive and long term incentive) paid by the Company to the following officers was below the 25 <sup>th</sup> percentile: President and CEO, EVP Global Operations, and Chief Legal Officer. The only officer at or above the 50 <sup>th</sup> percentile for total compensation was the Company's Chief Financial Officer and Treasurer who had joined the Company in May 2015 and which was the product of a negotiated armslength employment agreement.

As a result of the foregoing factors Willis Towers Watson recommended that the Company:

- In determining salary increases, the Company should consider the executive's current competitive position, performance, role, and retention risk.
- Implement a formal annual incentive opportunity for all executives.
- Implement a regular annual grant program for long-term incentives.

The foregoing observations and recommendations were studied, questioned and thoroughly discussed by the Compensation Committee, Willis Towers Watson and legal counsel over each of the four in-person meetings (the first of the five meetings mentioned above was conducted by telephone and was focused on engagement of advisors). Among other things, the Compensation Committee discussed and considered the recommendations made by Willis Towers Watson regarding the establishment of a compensation program with the long term objective to target the compensation program with the market 50 <sup>th</sup> percentile of our peer companies, however, for 2016, to target our compensation program to the 25 <sup>th</sup> percentile of our peer companies. In each case, however, the Compensation Committee's final decision or recommendation, as the case may be, as to total compensation to an individual would be based on the individual's actual performance and specific circumstances and to adjust base salary, short term incentive, and long term incentive accordingly.

The Compensation Committee recommended, and the Board subsequently adopted, a compensation philosophy for the Company's management team members to:

- Attract and retain talented and dedicated management team members;
- Provide overall compensation that is competitive in its industry;
- Correlate annual cash incentives to the achievement of its business and financial objectives;
   and
- Provide management team members with appropriate long-term incentives aligned with stockholder value.

As part of the compensation philosophy the Company's compensation focus will be to (1) drive the Company's strategic plan on growth, (2) align officer and management performance with the interests of the Company's stockholders, and (3) encourage retention of officers and management team members.

In furtherance of the compensation policy and as a result of the extensive deliberations, including consideration of the Willis Towers Watson recommendations, the Compensation Committee adopted an executive and management officer compensation structure for 2016 consisting of:

- A base salary comparable with job description and industry standard.
- A short term incentive or cash bonus plan based on a combination of factors including
  individual performance against corporate goals as well as overall corporate and division
  performance. Target bonus to be denominated as a percent of base salary with specific goals
  weightings and pay-out ranges).
- A long term incentive or equity awards inline with job description, performance, and industry standards.

The Compensation Committee's intention is that the compensation structure approved for 2016 will remain in place indefinitely. However, it will review performance and results after the first year and thereafter and evaluate from time to time whether enhancements, changes or other compensation structures are in the Company's and it stockholders best interests.

Reflecting the new approach, the Compensation Committee established (i) annual base salaries at levels that it believed (based heavily on the data provided by Willis Towers Watson) are generally competitive with executives in our peer group and in other comparable publicly-held companies as described in the executive pay assessment prepared by Willis Towers Watson, and (ii) short term incentives in the form of discretionary annual cash bonuses based on the achievement of identified goals and benchmarks. Long-term incentives in the form of employee stock options and restricted stock units will be used as a retention tool and as a means to further align an executive's long-term interests with those of the Company's stockholders, with the ultimate objective of affording our executives an appropriate incentive to help drive increases in stockholder value.

The Compensation Committee will evaluate both executive performance and compensation to maintain the Company's ability to attract and retain highly-qualified executives in key positions and to assure that compensation provided to executives remains competitive when compared to the compensation paid to similarly situated executives of companies with whom we compete for executive talent or that we consider comparable to our company.

#### Role of Chief Executive Officer in Compensation Decisions

In connection with the implementation of the new compensation structure, the Compensation Committee conducted the thorough review of executive compensation discussed above. The Compensation Committee engaged in extensive discussions with and considered with great weight the recommendations of the Chief Executive Officer as to compensation for executive and management team members other than for the Chief Executive Officer.

In connection with consideration of 2015 performance bonuses for members of management, the Chief Executive Officer prepared and submitted recommendations for each of the executive and management team members, other than her own. In considering these recommendations, the Compensation Committee had the benefit of its extensive deliberations as well as the data provided by Willis Towers Watson. In executive session, the Compensation Committee approved a 2015 performance bonus for the Chief Executive Officer. At the Compensation Committee's February 17, 2016 meeting, it approved recommendations to the Board for its February 18, 2016 meeting, at which time the Board approved the same. The Board

approval covered certain officers including the five officers set forth below. In addition, our Chief Executive Officer discussed recommendations for other management team members but the Compensation Committee and Board agreed that such positions were within the scope of the Chief Executive Officer's authority and did not require the Compensation Committee or Board approval.

The Compensation Committee expects to perform an annual review of executive compensation, generally in the first quarter of the year following the year in review, with a presentation by the Chief Executive Officer regarding each element of the executive compensation arrangements. At the Compensation Committee's direction, the Chief Executive Officer prepared an executive compensation review for each executive officer (other than the Chief Executive Officer), as well as the full executive team, which included recommendations for:

- 2016 Base Salary
- A proposed year-end short -term incentive in the form of a target cash bonus based on the achievement of certain objectives; and
- A long-term incentive in the form of stock options and restricted stock units for the year under review.

As part of the compensation review, the Chief Executive Officer may also recommend other changes to an executive's compensation arrangements such as a change in the executive's responsibilities. The Compensation Committee will evaluate the Chief Executive Officer's recommendations and, in its discretion, may accept or reject the recommendations, subject to the terms of any written employment agreements.

The Compensation Committee met in executive session without our Chief Executive Officer to consider the Chief Executive Officer's compensation, including base salary, cash bonus and equity award, if any. Prior to such executive sessions, the Compensation Committee interviewed the Chief Executive Officer to obtain a better understanding of factors contributing to the Chief Executive Officer's compensation. With the exception of these executive sessions of the Compensation Committee, as a rule, our Chief Executive Officer participated in all deliberations of the Compensation Committee relating to executive compensation. However, the Compensation Committee also asked our Chief Executive Officer to be excused for certain deliberations with respect to the compensation recommended for Margaret Cotter, the sister of the Chief Executive Officer.

In conjunction with the year-end annual compensation review, or as soon as practicable after the year-end, our Chief Executive Officer will recommend to the Compensation Committee the Company objectives and other criteria to be utilized for purposes of determining cash bonuses for certain senior executive officers. The Compensation Committee, in its discretion, may revise the Chief Executive Officer's recommendations. At the end of the year, the Compensation Committee, in consultation with the Chief Executive Officer, will review each performance goal and determine the extent to which the officer achieved such goals. In establishing performance goals, the Compensation Committee expects to consider whether the goals could possibly result in an incentive for any executives to take unwarranted risks in our Company's business and intend to seek to avoid creating any such incentives.

#### **Base Salaries**

The Compensation Committee reviewed the executive pay assessment prepared by Willis Towers Watson and other factors and engaged in extensive deliberation and then recommended the following 2016 base salaries (the 2015 base salaries are shown for comparison purposes) for the following officers; the Board approved the recommendations of the Compensation Committee on March 10, 2016: the President and Chief Executive Officer, Chief Financial Officer and the persons identified and Named Executive Officers in the Company's proxy statement dated November 10, 2015 other than our prior Chief Executive Officers James J. Cotter, Sr. and James J. Cotter, Jr:

Name	Title	2016 Base Salary(4)	2015 Base Salary(4)
Ellen Cotter (1)	President and Chief Executive Officer	\$450,000	\$402,000
Devasis Ghose (2)	Chief Financial Officer	400,000	400,000
Andrzej Matyczynski (3)	EVP Global Operations	336,000	312,000
Robert F. Smerling	President, US Cinemas	375,000	350,000
Wayne Smith	Managing Director, ANZ	A\$370,000	A\$365,360

- Ellen M. Cotter was appointed Interim President and Chief Executive Officer on June 12, 2015 and President and Chief Executive Officer on January 8, 2016.
- (2) Devasis Ghose was appointed Chief Financial Officer on May 11, 2015. Mr. Ghose is the only executive officer that is a party to an employment agreement.
- (3) Andrzej Matyczynski was the Company's Chief Financial Officer until May 11, 2015 and thereafter he acted as corporate advisor to the Company. He was appointed EVP-Global Operations on March 10, 2016.
- (4) All dollars are US dollars except the salary for Wayne Smith is reported in Australian dollars Short Term Incentives

The Short Term Incentives authorized by the Compensation Committee and the Board provides the Company's executive officers and other management team members, who are selected to participate, with an opportunity to earn an annual cash bonus based upon the achievement of certain Company financial goals, division goals and individual goals, established by the Company's Chief Executive Officer and approved by the Compensation Committee and the Board of Directors (in future years, under the Compensation Committee Charter approved by the Board on March 10, 2016, the Compensation Committee will have full authority to approve

these matters). Specifically, a participant in the short term incentive plan will be advised of his or her annual potential target bonus expressed as a percentage of the participant's base salary and by dollar amount. The participant will be eligible for a short term incentive bonus once the participant achieves goals identified at the beginning of the year for a threshold target, the potential target or potential maximum target bonus opportunity. The bonus will vary depending upon the achievements made by the individual participants, the division and the corporation. Corporate goals will include levels of earnings before interest, depreciation, taxes and amortization ("non-GAAP Operating Income") and property development milestones. Division goals will include levels of division cash flow and division milestones and individual goals will include specific unique performance goals specific to the individual's position in the Company. Each of the corporate, division and individual goals carries a different percentage weight in determining the officer's or other team member's bonus for the year.

For 2016, executive officers will have an annual bonus opportunity expressed and determined as a percent of their base salary. This approach also was a recommendation of the Willis Towers Watson report to the Compensation Committee and provided points of reference for our Compensation Committee to compare short-term incentive opportunities for our executive and management team to those in peer and competitor companies.

Ms. Ellen Cotter, President and Chief Executive Officer, has a potential target bonus opportunity of 95% of Base Salary, or \$427,500 at target. Of that potential target bonus opportunity, a threshold bonus of \$213,750 may be achieved based upon Ms. Cotter's achievement of her performance goals and the Company's achievement of corporate goals as discussed above, a potential maximum target of \$641,250 is based on achieving performance goals approved by the Chairman of the Compensation Committee. Ms. Cotter's aggregate annual bonus opportunity can range from \$0 to \$641,250. Mr. Devasis Ghose, Chief Financial Officer, has a potential target bonus opportunity of 50% of Base Salary, or \$200,000 at target, which is based on achievement of his performance goals and the Company's achievement of corporate goals, as discussed above. Mr. Ghose's aggregate annual bonus opportunity can range from \$0 to \$300,000 (the maximum potential target if additional performance goals are met by Mr. Ghose). Mr. Andrzej Matyczynski, EVP - Global Operations, has a target bonus opportunity of 50% of Base Salary, or \$168,000 at target, which is based on achievement of his performance goals, the Company's achievement of corporate goals and certain divisional goals. Mr. Matyczynski's aggregate annual bonus opportunity can range from \$0 to \$252,000 (the maximum potential target if additional performance goals are met by Mr. Matyczynski). Mr. Robert Smerling, President, US Cinemas, has a target bonus opportunity of 30% of base pay, or \$112,500 at target, which is based on achievement of his performance goals, the Company's achievement of corporate goals and certain divisional goals. Mr. Smerling's aggregate annual bonus opportunity can range from \$0 to \$168,750 (the maximum potential target if additional performance goals are met by Mr. Smerling). Mr. Wayne Smith, Managing Director, ANZ, has a target bonus opportunity of 40% of Base Salary, or A\$148,000 at target, which is based on achievement of his performance goals, the Company's achievement of corporate goals and certain divisional goals. Mr. Smith's aggregate annual bonus opportunity can range from A\$0 to A\$222,000 (the maximum potential target if additional performance goals are met by Mr. Smith). The positions of other management team members have target bonus opportunities ranging from 20% to 30% of Base Salary based on achievement certain goals. The highest level of achievement,

participants may be eligible to receive up to a maximum of 150% of his or her target bonus amount.

#### Long-Term Incentives

Long-Term incentives will utilize the equity-based plan under the Company's 2010 Incentive Stock Plan, as amended (the "2010 Plan"). For 2016, executive and management team participants will receive awards in the following forms: 50% time-based restricted stock units and 50% non-statutory stock options. The grants of restricted stock units and options will vest ratably over a four (4) year period with 1/4 the vesting on each anniversary date of the grant date.

On March 10, 2016 the following grants were made:

Name	Title	Dollar Amount of Restricted Stock Units (1)	Dollar Amount of Non-Statutory Stock Options (1)
Ellen Cotter	President and Chief Executive Officer	\$150,000	\$150,000
Devasis Ghose (2)	Chief Financial Officer	0	0
Andrzej Matyczynski	EVP Global Operations	37,500	37,500
Robert F. Smerling	President, US Cinemas	50,000	50,000
Wayne Smith	Managing Director, ANZ	27,000	27,000

<sup>(1)</sup> The number of shares of stock to be issued will be calculated using the Black Scholes pricing model as of the date of grant of the award.

All long-term incentive awards will be subject to other terms and conditions set forth in the 2010 Plan and award grant.

#### **Director Compensation**

The Compensation Committee also undertook a review of outside director compensation. The process followed was similar to that in scope and approach used by the Compensation Committee in considering executive compensation as described above. The meetings of the Compensation Committee evaluating executive compensation in most cases also considered

<sup>(2)</sup> Mr. Devasis Ghose was awarded 100,000 non-statutory stock options vesting over a 4 year period on Mr. Ghose's commencement of employment on May 11, 2015.

outside director compensation, although a substantial majority of those meetings were focused on executive compensation considerations.

Willis Towers Watson reviewed and presented to the Compensation Committee the competitiveness of the Company's outside director compensation. The Company's outside director compensation was compared to the compensation paid by the 15 peer companies (identified above in the Executive Compensation discussion). Wills Towers Watson's key findings were:

- The Company's annual board retainer was slightly above the 50 th percentile while the total cash compensation paid to outside directors was close to the 25 th percentile.
- Due to minimal annual director equity grants at the Company, total direct compensation to the Company's outside directors was the lowest among the peer group.
- The Company should consider increasing its committee cash compensation and annual director equity grants to be in line with peer practices.

The foregoing observations and recommendations were studied, questioned and thoroughly discussed by the Compensation Committee, Willis Towers Watson and legal counsel at three of the four in-person meetings (the first of the five meetings mentioned above was conducted by telephone and was focused on engagement of advisors). Among other things, the Compensation Committee discussed and considered the recommendations made by Willis Towers Watson regarding director retainer fees and equity awards for directors. Following discussion the Committee recommended to the Board that:

- The Board retainer currently paid to outside directors not be changed
- The committee chair retainers be increased to \$20,000 for the Audit Committee and the Executive Committee and \$15,000 for the Compensation Committee.
- The committee member fees be increased to \$7,500 for the Audit and Executive Committees and to \$5,000 for the Compensation Committee.
- The Lead Director fee be increased to \$10,000.
- The annual equity award value to directors be \$60,000 as a fixed dollar value (using Black-Scholes), that the equity award be restricted stock units and that such restricted stock units have a twelve month vesting period.
- The Board approved additional special compensation to be paid to certain directors for extraordinary services provided to the Company and devotion of time in providing such services as follows:
  - o Guy W. Adams, \$50,000
  - o Edward L. Kane, \$10,000
  - Douglas J. McEachern, \$10,000

The recommendations of the Compensation Committee with respect to outside director compensation were presented and approved by the Board on May 10, 2016. Board compensation was made effective for the year 2016 and equity grants were made on March 10, 2016 based upon the closing of the Company's Class A Common Stock on such date.

#### Item 8.01 Other Events.

#### Separation Agreement - William D. Ellis

On February 18, 2016, William D. Ellis submitted his resignation as General Counsel and Corporate Secretary of the Company. On March 11, 2016, Reading entered into an agreement with Mr. William D. Ellis, pursuant to which Mr. Ellis will be available to advise the Company on matters on which he previously worked until December 31, 2016. Mr. Ellis' last day was March 11, 2016.

The foregoing description of the Separation Agreement is qualified in its entirety by reference to the provisions of the Separation Agreement filed as exhibit 10.3 to this Current Report on Form 8-K, which is incorporated herein by reference.

#### Item 9.01 Financial Statements and Exhibits.

#### (d) Exhibits

10.1	Amendment to 2010 Stock Incentive Plan approved March 10, 2016.
10.2	Compensatory Arrangements for Certain Executive Officers and Management.
10.3	Separation Agreement dated as of March 11, 2016, by and between William D. Ellis and Reading International, Inc.

#### **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Reading International, Inc.

Date: March 15, 2016 /s/ Devasis Ghose

Name: Devasis Ghose

Title: Chief Financial Officer

### READING INTERNATIONAL, INC. FIRST AMENDEMNT TO THE 2010 STOCK INCENTIVE PLAN

This First Amendment (the "Amendment") to the Reading International, Inc. 2010 Stock Incentive Plan (the "Plan"), is made and shall be effective as of this  $10^{\text{th}}$  day of March , 2016 (the "Effective Date").

#### RECITALS

WHEREAS, the stockholders of Reading International, Inc. (the "Company") approved the Plan on May 13, 2010 at the annual meeting of stockholders in accordance with the recommendation of the board of directors; and

WHEREAS, the Plan provides for awards of stock options, restricted stock, bonus stock, and stock appreciation rights to eligible employees, directors, and consultants;

WHEREAS, the Company believes that it would be be in the best interests of the Company and its stockholders to permit awards of restricted stock units;

WHEREAS, NASDAQ rules do not require stockholders to approve an amendment to an equity incentive plan if the amendment relates to adding restricted stock units as long as the Plan provides for the award of restricted stock;

WHEREAS, the Plan provides for the award of restricted stock;

NOW, THEREFORE, in accordance with Section 12 of the Plan, the Plan is amended as follows as of the Effective Date:

#### **AMENDMENTS**

- 1. Section 2(y) the definition of "Rule 16b-3" is hereby renumbered as Section 2(z).
- 2. Section 2(z) the definition of "Securities Act" is hereby renumbered as Section 2(aa).
- 3. Section 2(aa) the definition of "Stock Award" is hereby renumbered as Section 2(bb).
- 4. Section 2(bb) the definition of "Service" is hereby renumbered as Section 2(cc).
- Section 2(cc) the definition of "Stock Award Agreement" is hereby renumbered as Section 2(dd).
- 6. Section 2(dd) the definition of "<u>Ten Percent Stockholder</u>" is hereby renumbered as Section 2(ee).
- 7. Section 2(y) the definition of "Restricted Stock Units" is hereby added.

#### Page 1 of 1

"Restricted Stock Units" means a Stock Award which may be earned in whole or in part upon the passage of time or the attainment of performance criteria established by the Board and which may be settled for Common Stock, other securities or cash or a combination of Common Stock, other securities or cash as established by the Board.

- 8. Section 2(bb) of the Plan is hereby deleted and replaced in its entirety by the following: "Stock Award" means any right granted under the Plan, including an Option, a stock bonus, a right to acquire restricted stock, a restricted stock unit and a stock appreciation right granted under the Plan, whether singly, in combination or in tandem, to a Participant by the Board pursuant to such terms, conditions, restrictions and/or limitations, if any, as the Board may establish.
- 9. Section 7(d) is hereby added to the Plan as follows: Restricted Stock Units. Each restricted stock unit agreement shall be in such form and shall contain such terms and conditions as the Board shall deem appropriate. The terms and conditions of the restricted stock unit agreements may change from time to time, and the terms and conditions of separate restricted stock unit agreements need not be identical, but each restricted stock unit agreement shall include (through inclusion or incorporation of provisions hereof by reference in the agreement or otherwise) the substance of each of the following provisions:
- Consideration. A restricted stock unit may be awarded upon the passage of time, the attainment of performance criteria or the satisfaction or occurrence of such other events as established by the Board.
- ii. Vesting Generally. At the time of the grant of a restricted stock unit, the Board may impose such restrictions or conditions to vesting, and/or the acceleration of the vesting, of such restricted stock unit as it, in its sole discretion, deems appropriate. Vesting provisions of individual restricted stock units may vary.
- iii. **Termination of Service**. In the event that a Participant's Service terminates, any or all of the restricted stock units held by the Participant that have not vested as of the date of termination under the terms of the restricted stock unit agreement shall be forfeited to the Company in accordance with the restricted stock unit agreement, except as otherwise provided in the applicable restricted stock unit agreement.
- iv. **Transferability**. A restricted stock unit shall be subject to similar transfer restrictions as awards of restricted stock, except that no shares are actually awarded to a Participant who is granted restricted stock units on the date of grant, and such Participant shall have no rights of a stockholder with respect to such restricted stock units until the restrictions set forth in the restricted stock unit agreement have lapsed. Restricted stock units may be transferred to any trust established by a Participant for the benefit of the Participant, his or her spouse, and/or any one or more lineal descendants.

Page 2 of 2

- Voting, Dividend & Other Right . Holders of restricted stock units will not be entitled v. to vote or to receive the dividend equivalent rights in respect of the restricted stock units at the time of any payment of dividends to stockholders on Common Stock until they become owners of the Common Stock pursuant to their restricted stock unit agreement. If the applicable restricted stock unit agreement specifies that a Participant will be entitled to dividend equivalent rights, (i) the amount of any such dividend equivalent right shall equal the amount that would be payable to the Participant as a stockholder in respect of a number of shares equal to the number of vested restricted stock units then credited to the Participant, and (ii) any such dividend equivalent right shall be paid in accordance with the Company's payment practices as may be established from time to time and as of the date on which such dividend would have been payable in respect of outstanding shares of Common Stock (and in accordance with Section 409A of the Code with regard to awards subject thereto); provided that no dividend equivalents shall be currently paid on restricted share units that are not yet vested.
- 10. Except as modified hereby, the provisions of the Plan shall remain in full force and effect, and the Plan may be restated, as amended hereby, in its entirety.

Page 3 of 3

#### Compensatory Arrangements for Executive and Management Employees

#### Background

The Executive Committee ("Executive Committee") of the Board of Directors (the "Board") of Reading International, Inc. (("Reading," "Registrant" or the "Company"), upon the recommendation of our Chief Executive Officer, requested the Compensation and Stock Options Committee (the "Compensation Committee") to evaluate the Company's compensation policy for executive officers and outside directors and to establish a plan that encompasses sound corporate practices consistent with the best interests of the Company. The Compensation Committee undertook to review, evaluate, revise and recommend the adoption of new compensation arrangements for executive and management officers and outside directors of the Company. In January 2016, the Compensation Committee retained the international compensation consulting firm of Willis Towers Watson as its advisor in this process and also relied on the Company's legal counsel, Greenberg Traurig, LLP.

#### **Executive Compensation**

From late January to late February 2016, the Compensation Committee met five separate times with Willis Towers Watson, the Chief Executive Officer, and legal counsel. As part of its engagement Willis Towers Watson reviewed the Company's compensation paid to executive and management officers by position, in light of each person's duties and responsibilities. Willis Towers Watson then compared the top 12 executive and management positions at the Company to (i) executive compensation paid by a peer group and (ii) two surveys, the 2015 Towers Watson Data Services Top Management Survey Report and the 2015 Mercer MBD Executive Compensation Survey, in each case, identified by office position and duties performed by the officer. The peer group utilized by Willis Towers Watson included the following companies:

Arcadia Realty Trust
Associated Estates Realty Corp.
Carmike Cinemas Inc.
Cedar Realty Trust Inc.
Charter Hall Group
Epr Properties
Vicinity Centres
IMAX Corporation

Inland Real Estate Corp.
Kite Realty Group Trust
Marcus Corporation.
Pennsylvania Real Estate Investment Trust
Ramco-Gershenson Properties Trust
Urstadt Biddle Properties Inc.
Village Roadshow Ltd.

Willis Towers Watson selected the above peer group because (i) the companies included US and Australian based companies reflecting the Company's geographic operations and (ii) the companies were comparable to the Company based on revenue.

The executive pay assessment prepared by Willis Towers Watson measured the executive and management compensation paid by the Company against compensation paid by peer group companies and the companies listed in the two surveys based on the  $25^{\text{ th}}$ ,  $50^{\text{ th}}$  and  $75^{\text{ th}}$  percentile of such peer group and surveyed companies. The  $50^{\text{ th}}$  percentile was the median compensation

paid by such peer group and surveyed companies to executives performing similar responsibilities and duties.

In its report to the Compensation Committee, Willis Towers Watson noted that for Company executive officers:

- Base salaries in the aggregate were generally in the competitive zone of the market (1% below the market 50 th percentile), with certain notable exceptions on position by position review;
- Total cash compensation (base salary and cash bonus) in the aggregate was 26% below the 50 th percentile; and
- Total compensation (base salary, cash bonus and long term incentive awards) in the aggregate was 40% below the 50 <sup>th</sup> percentile.

The Compensation Committee, recommended, and the Board subsequently adopted, a compensation philosophy for the Company's management team members to:

- Attract and retain talented and dedicated management team members;
- Provide overall compensation that is competitive in its industry;
- Correlate annual cash incentives to the achievement of its business and financial objectives;
- Provide management team members with appropriate long-term incentives aligned with stockholder value.

As part of the compensation philosophy the Company's compensation focus will be to (1) drive the Company's strategic plan on growth, (2) align officer and management performance with the interests of the Company's stockholders, and (3) encourage retention of officers and management team members.

In furtherance of the compensation policy and as a result of the extensive deliberations, including consideration of the Willis Towers Watson recommendations, the Compensation Committee adopted an executive and management officer compensation structure for 2016 consisting of:

- A base salary comparable with job description and industry standard.
- A short term incentive or cash bonus plan based on a combination of factors including
  individual performance against corporate goals as well as overall corporate and division
  performance. Target bonus to be denominated as a percent of base salary with specific goals
  weightings and pay-out ranges).
- A long term incentive or equity awards inline with job description, performance, and industry standards.

The Compensation Committee's intention is that the compensation structure approved for 2016 will remain in place indefinitely. However, it will review performance and results after the first year and thereafter and evaluate from time to time whether enhancements, changes or other compensation structures are in the Company's and it stockholders best interests.

**JA3212** 

Reflecting the new approach, the Compensation Committee established (i) annual base salaries at levels that it believed (based heavily on the data provided by Willis Towers Watson) are generally competitive with executives in our peer group and in other comparable publicly-held companies as described in the executive pay assessment prepared by Willis Towers Watson, and (ii) short term incentives in the form of discretionary annual cash bonuses based on the achievement of identified goals and benchmarks. Long-term incentives in the form of employee stock options and restricted stock units will be used as a retention tool and as a means to further align an executive's long-term interests with those of the Company's stockholders, with the ultimate objective of affording our executives an appropriate incentive to help drive increases in stockholder value.

The Compensation Committee will evaluate both executive performance and compensation to maintain the Company's ability to attract and retain highly-qualified executives in key positions and to assure that compensation provided to executives remains competitive when compared to the compensation paid to similarly situated executives of companies with whom we compete for executive talent or that we consider comparable to our company.

#### Role of Chief Executive Officer in Compensation Decisions

In connection with the implementation of the new compensation structure, the Compensation Committee conducted the thorough review of executive compensation discussed above. The Compensation Committee engaged in extensive discussions with and considered with great weight the recommendations of the Chief Executive Officer as to compensation for executive and management team members other than for the Chief Executive Officer.

In connection with consideration of 2015 performance bonuses for members of management, the Chief Executive Officer prepared and submitted recommendations for each of the executive and management team members, other than her own. In considering these recommendations, the Compensation Committee had the benefit of its extensive deliberations as well as the data provided by Willis Towers Watson. In executive session, the Compensation Committee approved a 2015 performance bonus for the Chief Executive Officer. At the Compensation Committee's February 17, 2016 meeting, it approved recommendations to the Board for its February 18, 2016 meeting, at which time the Board approved the same. The Board approval covered certain officers including the five officers set forth below. In addition, our Chief Executive Officer discussed recommendations for other management team members but the Compensation Committee and Board agreed that such positions were within the scope of the Chief Executive Officer's authority and did not require the Compensation Committee or Board approval.

The Compensation Committee expects to perform an annual review of executive compensation, generally in the first quarter of the year following the year in review, with a presentation by the Chief Executive Officer regarding each element of the executive compensation arrangements. At the Compensation Committee's direction, the Chief Executive Officer prepared an executive compensation review for each executive officer (other than the Chief Executive Officer), as well as the full executive team, which included recommendations for:

2016 Base Salary

- A proposed year-end short -term incentive in the form of a target cash bonus based on the achievement of certain objectives; and
- A long-term incentive in the form of stock options and restricted stock units for the year under review.

As part of the compensation review, the Chief Executive Officer may also recommend other changes to an executive's compensation arrangements such as a change in the executive's responsibilities or a change in title. The Compensation Committee will evaluate the Chief Executive Officer's recommendations and, in its discretion, may accept or reject the recommendations, subject to the terms of any written employment agreements.

The Compensation Committee met in executive session without our Chief Executive Officer to consider the Chief Executive Officer's compensation, including base salary, cash bonus and equity award, if any. Prior to such executive sessions, the Compensation Committee interviewed the Chief Executive Officer to obtain a better understanding of factors contributing to the Chief Executive Officer's compensation. With the exception of these executive sessions of the Compensation Committee, as a rule, our Chief Executive Officer participated in all deliberations of the Compensation Committee relating to executive compensation. However, the Compensation Committee will ask the Chief Executive Officer to be excused for certain deliberations with respect to the compensation recommended for Margaret Cotter, the sister of the Chief Executive Officer.

In conjunction with the year-end annual compensation review, or as soon as practicable after the year-end, our Chief Executive Officer will recommend to the Compensation Committee the Company objectives and other criteria to be utilized for purposes of determining cash bonuses for certain senior executive officers. The Compensation Committee, in its discretion, may revise the Chief Executive Officer's recommendations. At the end of the year, the Compensation Committee, in consultation with the Chief Executive Officer, will review each performance goal and determine the extent to which the officer achieved such goals. In establishing performance goals, the Compensation Committee expects to consider whether the goals could possibly result in an incentive for any executives to take unwarranted risks in our Company's business and intend to seek to avoid creating any such incentives.

#### **Base Salaries**

The Compensation Committee reviewed the executive pay assessment prepared by Willis Towers Watson and other factors and engaged in extensive deliberation and then recommended the following 2016 base salaries (the 2015 base salaries are shown for comparison purposes) for the following officers; the Board approved the recommendations of the Compensation Committee on March 10, 2016: the President and Chief Executive Officer, Chief Financial Officer and the persons identified and Named Executive Officers in the Company's proxy statement dated November 10, 2015 other than our prior Chief Executive Officers James J. Cotter, Sr. and James J. Cotter, Jr:

Name	Title	2016 Base Salary (4)	2015 Base Salary (4)
Ellen Cotter (1)	President and Chief Executive Officer	\$450,000	\$402,000
Devasis Ghose (2)	Chief Financial Officer	400,000	400,000
Andrzej Matyczynski (3)	<b>EVP Global Operations</b>	336,000	312,000
Robert F. Smerling	President, US Cinemas	375,000	350,000
Wayne Smith	Managing Director, ANZ	A\$370,000	A\$365,360

- Ellen M. Cotter was appointed Interim President on June 12, 2015 and President and Chief Executive Officer on January 8, 2016.
- (2) Devasis Ghose was appointed Chief Financial Officer on May 11, 2015. Mr. Ghose is the only executive officer that is a party to an employment agreement.
- (3) Andrzej Matyczynski was the Company's Chief Financial Officer until May 11, 2015 and thereafter he acted as corporate advisor to the Company. He was appointed EVP-Global Operations on March 10, 2016.
- (4) All dollars are in US dollars except the salary for Wayne Smith is reported in Australian dollars.

#### Short Term Incentives

The Short Term Incentives authorized by the Compensation Committee and the Board provides the Company's executive officers and other management team members, who are selected to participate, with an opportunity to earn an annual cash bonus based upon the achievement of certain Company financial goals, division goals and individual goals, established by the Company's Chief Executive Officer and approved by the Compensation Committee and the Board of Directors (in future years, under the Compensation Committee Charter approved by the Board on March 10, 2016, the Compensation Committee will have full authority to approve these matters). Specifically, a participant in the short term incentive plan will be advised of his or her annual potential target bonus expressed as a percentage of the participant's base salary and by dollar amount. The participant will be eligible for a short term incentive bonus once the participant achieves goals identified at the beginning of the year for a threshold target, the potential target or potential maximum target bonus opportunity. The bonus will vary depending upon the achievements made by the individual participants, the division and the corporation. Corporate goals will include levels of earnings before interest, depreciation, taxes and amortization ("non-GAAP Operating Income") and property development milestones. Division goals will include levels of division cash flow and division milestones and individual goals will

include specific unique performance goals specific to the individual's position in the Company. Each of the corporate, division and individual goals carries a different percentage weight in determining the officer's or other team member's bonus for the year.

For 2016, executive officers will have an annual bonus opportunity expressed and determined as a percent of their base salary. This approach also was a recommendation of the Willis Towers Watson report to the Compensation Committee and provided points of reference for our Compensation Committee to compare short-term incentive opportunities for our executive and management team to those in peer and competitor companies.

Ms. Ellen Cotter, President and Chief Executive Officer, has a potential target bonus opportunity of 95% of Base Salary, or \$427,500 at target. Of that potential target bonus opportunity, a threshold bonus of \$213,750 may be achieved based upon Ms. Cotter's achievement of her performance goals and the Company's achievement of corporate goals as discussed above, a potential maximum target of \$641,250 is based on achieving performance goals approved by the Chairman of the Compensation Committee. Ms. Cotter's aggregate annual bonus opportunity can range from \$0 to \$641,250. Mr. Devasis Ghose, Chief Financial Officer, has a potential target bonus opportunity of 50% of Base Salary, or \$200,000 at target, which is based on achievement of his performance goals and the Company's achievement of corporate goals, as discussed above. Mr. Ghose's aggregate annual bonus opportunity can range from \$0 to \$300,000 (the maximum potential target if additional performance goals are met by Mr. Ghose). Mr. Andrzej Matyczynski, EVP - Global Operations, has a target bonus opportunity of 50% of Base Salary, or \$168,000 at target, which is based on achievement of his performance goals, the Company's achievement of corporate goals and certain divisional goals. Mr. Matyczynski's aggregate annual bonus opportunity can range from \$0 to \$252,000 (the maximum potential target if additional performance goals are met by Mr. Matyczynski). Mr. Robert Smerling, President, US Cinemas, has a target bonus opportunity of 30% of base pay, or \$112,500 at target, which is based on achievement of his performance goals, the Company's achievement of corporate goals and certain divisional goals. Mr. Smerling's aggregate annual bonus opportunity can range from \$0 to \$168,750 (the maximum potential target if additional performance goals are met by Mr. Smerling). Mr. Wayne Smith, Managing Director, ANZ, has a target bonus opportunity of 40% of Base Salary, or A\$148,000 at target, which is based on achievement of his performance goals the Company's achievement of corporate goals and certain divisional goals approved by the Compensation Committee. Mr. Smith's aggregate annual bonus opportunity can range from A\$0 to A\$222,000 (the maximum potential target if additional performance goals are met by Mr. Smith). The positions of other management team members have target bonus opportunities ranging from 20% to 30% of Base Salary based on achievement of certain goals. The highest level of achievement, participants may be eligible to receive up to a maximum of 150% of his or her target bonus amount.

#### Long-Term Incentives

Long-Term incentives will utilize the equity-based plan under the Company's 2010 Incentive Stock Plan, as amended (the "2010 Plan"). For 2016, executive and management team participants will receive awards in the following forms: 50% time-based restricted stock units and 50% non-statutory stock options. The grants of restricted stock units and options will vest ratably over a four (4) year period with 1/4 they vesting on each anniversary date of the grant date.

On March 10, 2016 the following grants were made:

Name	Title	Dollar Amount of Restricted Stock Units (1)	Dollar Amount of Non-Statutory Stock Options (1)
Ellen Cotter	President and Chief Executive Officer	\$150,000	\$150,000
Devasis Ghose (2)	Chief Financial Officer	0	0
Andrzej Matyczynski	EVP Global Operations	37,500	37,500
Robert F. Smerling	President, US Cinemas	50,000	50,000
Wayne Smith	Managing Director, ANZ	27,000	27,000

<sup>(1)</sup> The number of shares of stock to be issued will be calculated using the Black Scholes pricing model as of the date of grant of the award.

All long-term incentive awards will be subject to other terms and conditions set forth in the 2010 Plan and award grant.

<sup>(2)</sup> Mr. Devasis Ghose was awarded 100,000 non-statutory stock options vesting over a 4 year period on Mr. Ghose's commencement of employment on May 11, 2015.

#### Separation and Release Agreement

This Separation and Release Agreement (the "Agreement") is entered into as of March 11, 2016, by and between William D. Ellis ("Executive" or "you") and Reading International, Inc., a Nevada corporation ("Reading" or the "Company").

#### RECITALS:

WHEREAS, pursuant to that certain e mployment a greement dated October 20, 2014, as amended (t he "Employment Agreement"), Executive was hired by the Company in the capacity of General Counsel for a th ree (3) year ter m to end on October 20, 2017 (the "Employment Term");

Whereas, o n or about February 18, 2016, Executive gave notice to the Company that he was resigning from his employment under the Employment Agreement with the Company;

Whereas, the Company is willing to accept Executive's resignation, but desires to have the benefit of Executive's continued assistance and cooperation on Company matters as needed after his resignation, as described below;

Whereas, Executive and the Company agree that Executive 's resign ation will be effective March 11, 2016.

NOW, THEREFORE, in consideration of the mutual covenants hereinafter set forth, the parties hereto agree as follows:

- Resignation. Effective March 11, 2016 (the "Resignation Date"), you hereby re sign as the Company's General Counsel and Corporate Secretary and from any and all other positions that you hold as an officer, director, manager and/or employee of the Company and its various direct and indirect subsidiaries. Your status as a corporate officer, director, manager, employee or any fiduciary position with the Company and all affiliates will end on the Resignation Date.
- 2. Compensation . In exchange for the your continued cooperation and assistance, the Release provided below, and for the performance by you of your other obligations under this Agreement the Company hereby waives any rights it might have against you with respect to your early termination of your obligations under the Employment Agreement, including but not limited to any claim for breach of the Employment Agreement, and agrees to the following compensation and benefit treatment:

#### 2.1 Payments.

(a) Base Salary; Accrued Obligations. On March 11, 2016, you will receive payment for any accrued and unused vacation, your accrued but unpaid base salary through the Resignation Date, and reimbursement of unreimbursed business expenses for which substantiation has been submitted (or for which substantiation will be submitted, for charges on your corporate credit card already incurred but for which you do not

207672282 v2

receive a bill until after the Resignation Date) in accordance with the Company's policies and procedures (collectively, the "Accrued Obligations"). You acknowledge and agree that as of the date of this Agreement, you have no hours of accrued and unused vacation.

- (b) Compensation. So long as you are in material compliance with your obligations under this Agreement, you will be entitled to nineteen equal payment s in the amount of \$10,790, each payable semi-monthly beginning on March 30, 2016 and continuing on each of the Company's regular pay day thereafter until December 31, 2016 (the "Pay ments"). The Payments will be subject to applicable required tax withholding (if any).
- 2.2 Equity Awards . So long as you continue satisfy in all material respects your obligations to the Company under this Agreement, twenty thousand (20,000) of the employee stock options granted to you pursuant to Section 4 of your Employment Agreement shall continue to vest on October 20, 2016. No further options shall vest under that grant. This provision shall be interpreted consistent with and supplementary to the stock option agreement.

#### 2.3 Benefits

- (a) COBRA. Y ou will be offered the opportunity to receive continuation coverage for yourself and your eligible dependents under the Company's medical and dental plan s pursuant to the provisions of the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended ("COBRA") following the Resignation Date, provided you timely elect and pay for such coverage.
- (b) Key Man Insurance. The Company shall pay the premiums on your key man insurance policy through December 31, 2016.
- O ther Compensation Matters. Notwithstanding anything to the contrary contained in this Agreement (including the Release set forth in Section 6 hereof), you hereby acknowledge that, in connection with your resignation when you cease to be an employee of the Company, you will not be entitled to receive from the Company or an affiliate (i) any severance or other pay ments or benefits, except as provided for in this Section 2, or (ii) any retiree termination welfare benefits (other than health care continuation coverage that you may be entitled to elect pursuant to Section 4980B of the Code).

2

- 3. Nondisparagement. You agree to refrain from making any false or misleading statements or comments about the Company and any of its respective affiliates, their officers, directors, personnel, or any of their products and services. You agree to refrain from making any disparaging remarks to any person (other than comments to your immediate family members or advisers that are made on a confidential basis and are not repeated or published by such persons) about the Company and any of its respective affiliates, their officers, directors, their respective personnel, and their respective products and services; except to the extent otherwise required by applicable law. The Company agrees to refra in from making any false, misleading, or disparaging statements about you to any person outside the Company (other than comments to advisers of the Company that are made on a confidential basis and are not repeated or published by such persons) except to the extent otherwise required by applicable law. The Company has no obligation under Section 3 with respect to James J. Cotter, Jr.
- 4. Cooperation . In order to ensure a smooth transition from Executive's duties and responsibilities as the Company's General Counsel and Secretary, and taking into consideration Executive's schedule and other commitments, Executive agrees to provide reasonable assistance to and cooperation with Company following the Resignation Date in connection with any Company matters for which Executive had knowledge or responsibility while employed by Company. Further, if Company is involved in any legal action or investigation on or after Executive's Resignation Date relating to events which occurred during Executive's employment, Executive will cooperate with the Company to the fullest extent reasonably possible (taking into consideration Executive's schedule and other commitments) in the preparation, prosecution, or defense of the Company's case, including, but not limited to, required travel, appearances and testimony, the execution of affidavits or documents or providing information reasonably requested by the Company. As part of his cooperation and assistance pursuant to this Section 4, Executive agrees that he will take and/or promptly return phone calls and promptly respond to emails or other communications from the Company or its representatives, and will make himself available to meet with the Company or its representatives in person at its Los Angles offices or other location in Los Angeles County upon reasonable request by the Company. Company will reimburse Executive for reasonable pre-approved out-of-pocket expenses incurred in providing such assistance and cooperation to the Company. Executive agrees that in providing such services, Executive will be serving as an attorney for the Company, and that any communications between the Company (or any of its counsel) and Executive shall be subject to the attorney-client and attorney work product privilege. Executive acknowledges that he has no right or authority to waive any attorneyclient or attorney work product privilege belonging to the Company and/or any of its affiliates, and that he shall not provide any information in violation of such privileges. Executive further agrees that he shall not meet or otherwise communicate with any counter-party or any representative of any counter party to any litigation in which the Company (or any of its officers or directors) is a party, whether or not nominal, without the prior written consent of the Company.

- 5. Public Comment. Prior to issuing any press release or SEC filing (e.g. Form 8-K) regarding your resignation, the Company agrees to give Executive 24 hours, with the opportunity to review and comment on the written draft release or SEC filing, notice prior to the requisite filing date.
- 6. Release. You hereby acknowledge that the Company's obligations under Section 2 hereof are in excess of any payments or benefits to which you are entitled under law, contract or otherwise and are contingent upon your timely performance of your obligations under this Agreement in all material respects, and the release of claims set forth in this Section 6 (the "Release"). For purposes of this Section 6, "Release d Parties" include the Company and its affiliated companies and their officers, directors, managers, s tock holders, employees, agents, representatives, plans, trusts, administrators, fiduciaries, insurance companies, attorneys, successors, and assigns.
  - 6.1 You, on behalf of yourself and your personal and legal representatives, heirs, executors, successors and assigns, hereby acknowledge full and complete satisfaction of, and fully and forever waive, release, and discharge the Release d Parties from any and all claims, causes of action, demands, liabilities, damages, obligations, and debts (collectively referenced as "Claims"), of every kind and nature, whether known or unknown, suspected or unsuspected, that you hold as of the date you sign this Agreement, or at any time previously held against any Release d Party, arising out of any matter whatsoever (with the exception of breach of this Agreement). This release specifically includes, but is not limited to, any and all Claims:
    - (a) Arising out of or in any way related to your employment with or separation of employment from the Company, or any contract or agreement between you and the Company or the termination thereof;
    - (b) Arising out of or in any way related to any treatment of Executive by any of the Release d Parties, which shall include, without limitation, any treatment or decisions with respect to hiring, placement, promotion, discipline, work hours, assignment of or change in duties or responsibilities, demotion, transfer, termination, compensation, performance review, or training; any statements or alleged statements by the Company or any of the Release d Parties regarding Executive, whether oral or in writing; any damages or injury that Executive may have suffered, including without limitation, emotional or physical injury, compensatory damages, or lost wages; or employment discrimination, which shall include, without limitation, any individual or class claims of discrimination on the basis of age, disability, sex, race, religion, national origin, citizenship status, marital status, sexual preference, or any other basis whatsoever.
    - (c) Arising under or based on the Equal Pay Act of 1963 (EPA); Title VII of the Civil Rights Act of 1964, as amended (Title VII); Section 1981 of the Civil Rights Act of 1866 (42 U.S.C. §1981); the Civil Rights Act of 1991

(42 U.S.C. §1981a); the Americans with Disabilities Act of 1990, as amended (ADA); the Family and Medical Leave Act of 1993, as amended (FMLA); the Genetic Information Nondiscrimination Act of 2008 (GINA); the National Labor Relations Act (NLRA); the Worker Adjustment and Retraining Notification Act of 1988 (WARN); the Uniform Services Employment and Reemployment Rights Act (USERRA); the Rehabilitation Act of 1973; the Occupational Safety and Health Act (OSHA); the Employee Retirement Income Security Act of 1974 (ERISA) (except claims for vested benefits, if any, to which you are legally entitled); the False Claims Act; Title VIII of the Corporate and Criminal Fraud and Accountability Act, as amended (18 U.S.C. §1514A) (Sarbanes-Oxley Act); the federal Whistleblower Protection Act and any state whistleblower protection statute(s); the California Fair Employment and Housing Act or any other federal, state or local law relating to employment or discrimination in employment or any other fair employment practices statute(s) of any state, in all cases arising out of or relating to your employment by Reading or investment in Reading or your services as an officer or employee of Reading or its subsidiaries, or otherwise relating to the termination of such employment or services.

- (d) Arising under or based on any other federal, state, county or local law, statute, ordinance, decision, order, policy or regulation prohibiting employment discrimination, providing for the payment of wages or benefits, or otherwise creating rights or claims for employees; any and all claims alleging breach of public policy, the implied covenant of good faith and fair dealing, or any express, implied, oral or written contract, handbook, manual, policy statement or employment practice, including, but not limited to, the Employment Agreement or Amendment thereto; constructive discharge; misrepresentation; defamation; libel; slander; interference with contractual relations; intentional or negligent infliction of emotional distress; invasion of privacy; assault; battery; fraud; negligence; harassment; retaliation; or wrongful discharge; and
- (e) Arising under or based on the Age Discrimination in Employment Act of 1967 ("ADEA"), as amended by the Older Workers Benefit Protection Act ("OWBPA"), and alleging a violation thereof by any Release d Party, at any time prior to the date you sign this Agreement.
- 6.2 You agree that, except as set forth in this Agreement, you are not entitled to any payment or benefits from any of the Release d Parties, including, but not limited to, any payments or benefits under any plan, program or agreement with any Release d Party, including, but not limited to, the Employment Agreement or Amendment thereto.
- 6.3 You agree that, this Agreement extinguishes all claims and charges that you could have raised against any of the Released Parties, whether known to you or not. You expressly waive all rights and benefits under Section 1542 of the California

Civil Code and any similar law of any state or territory of the United States. Section 1542 of the California Civil Code provides as follows:

- "A general release does not extend to claims which the creditor does not know or suspect to exist in his or her favor at the time of executing the release, which if known by him or her must have materially affected his or her settlement with the debtor."
- 6.4 You hereby represent that you know of no claim that you have that has not been released by this Section 6. You further represent and warrant that you have not assigned or subrogated any of your rights, claims or causes of action, including any claims referenced in this Agreement, or authorized any other person or entity to assert such claim or claims on your behalf, and you agree to indemnify and hold the Company harmless against any assignment of said rights, claims and/or causes of action.
- 6.5 Nothing contained in this Release will (i) release any claim that cannot be waived under applicable law, (ii) release your rights to any benefits under any employee welfare benefit plan of the Company, the 401(k) Plan or with respect to the right to elect health care continuation under COBRA, (iii) release any entitlement to or with respect to indemnification which you may have pursuant to agreement, the Company's bylaws, any policy of insurance maintained by the Company or otherwise under law, or (iv) be construed to release your rights under this Agreement or be construed to prohibit or restrict you in any manner from bringing appropriate proceedings to enforce this Agreement. You acknowledge that your execution of this Agreement terminates any claims you previously held to any and all compensation and employee benefits, other than those specifically identified in this Agreement.
- 6.6 By signing this Agreement, you represent that you have not commenced or joined in any claim, charge, action or proceeding whatsoever against any of the Release d Parties arising out of or relating to any of the matters set forth in this Section 6. You further represent that you will not be entitled to any personal recovery in any action or proceeding that may be commenced on your behalf arising out of the matters released hereby.
- 7. Release of ADEA Claims. You expressly acknowledge and agree that this Agreement includes a release of all claims which you have or may have under the Age Discrimination in Employment Act, as amended ("ADEA"). The following terms and conditions apply to and are part of the release of ADEA claims under this Agreement:
  - You have been advised to consult with an attorney before signing this Agreement;
  - (b) You are not releasing any rights or claims under the ADEA that may arise after the date on which you execute this Agreement;

- (c) You have twenty-one (21) days from the date you are presented with this Agreement to decide whether or not to sign this Agreement, although you may choose to sign the Agreement at any time earlier;
- (d) You have seven (7) days after signing this Agreement to revoke this Agreement (the "Revocation Period"), and this Agreement will not be effective until that Revocation Period has expired;
- (e) To revoke this Agreement, you must deliver written notice of revocation by hand, overnight delivery, or confirmed facsimile signed by you and received by the Company to the attention of Ellen Cotter, President & CEO, no later than the seventh (7th) day of the Revocation Period. If no such revocation occurs, the General Release and this Agreement will become effective on the eighth (8th) day following your execution of this Agreement. You further acknowledge and agree that, in the event that you revoke this Agreement, it will have no force or effect; and
- (f) You hereby acknowledge and agree that you are knowingly and voluntarily releasing your rights and claims only in exchange for consideration (something of value) in addition to anything of value to which you are already entitled.
- Restrictive Covenants; Arbitration; Surviving Provisions . You acknowledge and agree that Sections 8 (Non-Disclosure), 9 (Remedies), and 12 (Data), and 13 (Arbitration) of the Employment Agreement shall remain in effect after your resignation and termination of your employment, and are expressly incorporated herein. You further agree that any disputes related to this Agreement, or breach therefor, including the arbitrability of such dispute or controversy, shall be determined and settled by arbitration pursuant to the procedures set forth in Section 13 of the Employment Agreement. Further, the provisions of the Company policies that relate to trade secrets, confidential and proprietary information and non-solicitation of employees will survive the termination of your employment and are incorporated in this Section 8 by reference. Payments to you or on your behalf under Section 2.1(b), 2.2, and 2.3(b) will be conditioned on your continued compliance with the provisions of these provisions and the provisions of this Agreement. In the event of any violation by you of these provisions or the provisions of this Agreement, no further payments will be made under Section 2.1(b) or 2.3(b) and no vesting of any unvested equity awards will occur under Section 2.2, and your right to any unpaid payments under Section 2.1(b) and 2.3(b) and any unvested equity awards under Section 2.2 will be forfeited.

#### 9. General Provisions .

9.1 <u>Severability</u>. It is the desire and intent of the parties that the provisions of this Agreement will be enforced to the fullest extent permissible. In the event that any one or more of the provisions of this Agreement will be held to be invalid, illegal or unenforceable, the validity, legality and enforceability of the remainder of this Agreement will remain valid and enforceable and continue in full force and effect

to the fullest extent consistent with law. Moreover, if any one or more of the provisions contained in this Agreement is held to be excessively broad as to duration, scope, activity or subject, such provisions will be construed by limiting and reducing them so as to be enforceable to the maximum extent compatible with applicable law.

- 9.2 <u>No Admission</u>. By entering into this Agreement, the parties do not admit to, and expressly deny, any wrongdoing.
- 9.3 Return of Property. You agree to return to the Company, on or prior to the Resignation Date, all files, records, documents, reports, computers and other property of the Company in your possession or control, including, but not limited to, any documents or other materials containing c onfidential i nformation, and you further agree that you will not keep, transfer or use any copies or excerpts of the foregoing items. Executive will be permitted to copy and remove any electronic files on the computer or cell phone that contain his personal information (but not any confidential information or proprietary Company information or data), including contact information. Executive understands and agrees that following his resignation, the Company shall have the right to access and review any files on his Company-provided computer, and to open and review any emails received at his Company email address.
- 9.4 Notices. Unless otherwise specified in this Agreement, a ny and all notices, requests, demands and other communications provided for by this Agreement will be in writing and will be effective when delivered in person, consigned to a reputable national or international courier service (including Federal Express), and addressed to you at your last known address on the books of the Company (which is 1995 Monte Vista Street, Pasadena, CA 91107) or, in the case of the Company, at the Company's principal place of business (which is 6100 Center Drive, Suite 900, Los Angeles, CA 90045), attention of the CEO of the Company, or to such other address as either party may specify by notice to the other actually received.
- 9.5 <u>Successors and Assigns</u>. This Agreement is personal to you and, without the prior written consent of the Company, will not be assignable by you otherwise than by will or the laws of descent and distribution. This Agreement will inure to the benefit of and be enforceable by your legal representatives. This Agreement will inure to the benefit of and be binding upon the Company and its successors and assigns.
- 9.6 Governing Law; Captions; Amendment. This Agreement will be governed by, and construed in accordance with, the laws of the State of California, without reference to principles of conflict of laws. The captions of this Agreement are not part of the provisions hereof and will have no force or effect. This Agreement may not be amended or modified except by a written agreement executed by the parties hereto or their respective successors and legal representatives.

- 9.7 Code Section 409A Compliance. The Company and you each hereby affirm that it is their mutual view that the provision of payments and benefits described or referenced herein are either exempt from or in tended to be in compliance with the requirements of Section 409A of the Code and the Treasury regulations relating thereto ("Section 409A ") and that each party's tax reporting will be completed in a manner consistent with such view. The Company and you each agree that upon the Resignation Date, you will experience a "separation from service" for purposes of Section 409A. A ny payments that qualify for the "short-term deferral" exception or another exception under Section 409A will be paid under the applicable exception. For purposes of the limitations on nonqualified deferred compensation under Section 409A of the Code, each payment of compensation under this Agreement will be treated as a separate payment of compensation. Notwithstanding anything to the contrary in this Agreement, all reimbursements and in-kind benefits provided under this Agreement will be made or provided in accordance with the requirements of Section 409A of the Code, including, where applicable, the requirement that (x) the amount of expenses eligible for reimbursement, or in kind benefits provided, during a calendar year may not affect the expenses eligible for reimbursement, or in kind benefits to be provided, in any other calendar year; (y) the reimbursement of an eligible expense will be made no later than the last day of the calendar year following the year in which the expense is incurred; and (z) the right to reimbursement or in kind benefits is not subject to liquidation or exchange for another benefit. Neither the Company nor its affiliates will be liable in any manner for any federal, state or local income or excise taxes (including but not limited to any taxes under Sections 409A of the Code), or penalties or interest with respect thereto, as a result of the payment of any compensation or benefits hereunder or the inclusion of any such compensation or benefits or the value thereof in your income.
  - You acknowledge and agree that the Company will not be responsible for any additional taxes or penalties resulting from the application of Section 409A.
- 9.8 Withholding. Notwithstanding any other provision of this Agreement, the Company may withhold from amounts payable under this Agreement all amounts that are required to be withheld, including, but not limited to, federal, state, local and foreign taxes to be withheld by applicable laws or regulations, but will only take such withholdings to the minimum extent permissible under applicable laws or regulations.
- 9.9 <u>Preparation of Agreement</u>. This Agreement will be interpreted in accordance with the plain meaning of its terms and not strictly for or against any of the parties hereto. Regardless of which party initially drafted this Agreement, it will not be construed against any one party, and will be construed and enforced as a mutually-prepared document.
- 9.10 <u>Entire Agreement</u>. This Agreement constitutes the entire agreement between you and the Company with respect to the subjects addressed herein, and together with the provisions that survive your resignation and termination of your employment as specified in this Agreement, this Agreement supersede all prior agreements,

understandings and representations, written or oral, with respect to those subjects, including, but not limited to the, Employment Agreement and Amendment thereto . Without limiting the generality of the foregoing, you acknowledge that the Employment Agreement and Amendment thereto will be terminated upon the effectiveness of this Agreement, except as specified in this Agreement.

- 9.11 <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which will be deemed an original, and which together will be deemed to be one and the same instrument.
- 9.12 <u>Waiver of Breach</u>. Any waiver of any breach of this Agreement shall not be construed to be a continuing wavier or consent to any subsequent breach on the part of you or of the Company.
- 10. **Consultation w ith Attorney; Voluntary Agreement.** You understand and agree that you have the right and have been given the opportunity to review this Agreement and, specifically, the Release set forth in Section 6 above, with an attorney of your choice. You also understand and agree that you are under no obligation to consent to the Release. You acknowledge that you have read this Agreement and the Release and understand their terms and that you enter into this Agreement freely, voluntarily, and without coercion.

#### READ CAREFULLY BEFORE SIGNING

THIS SEPARATION AND RELEASE AGREEMENT INCLUDES A RELEASE OF ALL KNOWN AND UNKNOWN CLAIMS AND A WAIVER OF YOUR RIGHTS UNDER THE AGE DISCRIMINATION IN EMPLOYMENT ACT AS WELL AS OTHER FEDERAL, STATE AND LOCAL LAWS PROTECTING EMPLOYEE RIGHTS. IF YOU SIGN THIS AGREEMENT, YOU ARE WAIVING ALL OF YOUR RIGHTS TO ASSERT ANY CLAIMS UNDER THESE LAWS. PLEASE READ THIS AGREEMENT CAREFULLY AND SEEK THE ADVICE OF AN ATTORNEY REGARDING THE LEGAL EFFECT OF SIGNING THIS AGREEMENT.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement on the day and year written opposite their signature .

"E xecutive"		
Date: William D. Ellis , an individual		<u></u>
" <i>Compan</i> y" Reading International, Inc.		
Date:		By:
	11	

# **EXHIBIT 5**

```
1
                       DISTRICT COURT
 2
                    CLARK COUNTY, NEVADA
 3
   JAMES J. COTTER, JR.,
    individually and
 5
   derivatively on behalf of)
    Reading International,
   Inc.,
                              ) Case No. A-15-719860-B
 7
            Plaintiff,
                              ) Coordinated with:
      vs.
                              ) Case No. P-14-082942-E
   MARGARET COTTER, et al.,
 9
10
            Defendants.
    and
11
    READING INTERNATIONAL,
12
    INC., a Nevada
    corporation,
13
            Nominal Defendant)
14
15
16
         VIDEOTAPED DEPOSITION OF DOUGLAS MCEACHERN
17
                   TAKEN ON MAY 6, 2016
18
19
20
21
22
23
24
    REPORTED BY:
25
     PATRICIA L. HUBBARD, CSR #3400
```

Page 246 Have you personally ever done any 1 0. 2 analysis to ascertain whether the profitability of the live theaters justifies the compensation that 3 4 has been awarded over the past three years to Margaret Cotter? 5 6 MR. SWANIS: Objection. Form. 7 MR. SEARCY: Objection. Lacks 8 foundation, assumes facts. 9 THE WITNESS: Yes and no. Margaret Cotter has just transitioned 10 from being an outside contractor to an employee of 11 the company. And in connection with that her change 12 in status was reviewed by the audit and conflicts 13 committee of which I chair. 14 We reviewed her management agreement 15 that she succeeded to before she managed -- took 16 over management. Somebody else managed Liberty 17 Theaters before Margaret took it over. 18 Margaret's compensation is 90 percent 19 20 based upon the cash flows of the live theater 21 operation. 22 Now, if we were to terminate that 23 contract with Liberty Theaters, Margaret Cotter, she 24 would be entitled to that same compensation in 25 perpetuity until such time as the shows that were

Litigation Services | 800-330-1112 www.litigationservices.com

Page 247 playing in those theaters ended. 1 2 So her compensation is contractual, not based -- and based upon performance of the theaters, 3 4 not based upon any discretion of the compensation 5 committee. 6 BY MR. NATION: 7 Okay. So, under the contract that 8 pre-existed your coming onto the board --9 Α. Uh-huh. 10 Ο. -- Margaret Cotter is entitled to receive 90 percent of the cash flows from the live 11 theaters as her compensation? 12 13 MR. SWANIS: Objection. Form. 14 MR. SEARCY: Same objection. 15 BY MR. NATION: 16 I'm trying to ask clarification of that. Q. It's not -- 90 percent of the 17 A. compensation she receives is as a result of this 18 19 formula calculation on the cash flow of the live I'm sorry I misspoke. 20 theaters. 21 There's some modest amount of compensation, as I recall, attributable to a monthly 22 23 payment that she gets, but the majority of the 24 compensation comes out of this contract that's been

> Litigation Services | 800-330-1112 www.litigationservices.com

filed with the S.E.C. and is a public record.

25

Page 262 Chicago, there were three in New York. One of them 1 2 in New York was located in the Union Square Building. 3 BY MR. NATION: 5 0. Which theater is that? 6 Α. I don't know the name of it. It was the 7 Union Square Theater. 8 Q. Okay. And Margaret wanted to be in 9 charge of developing the Union Square Theater is 10 your understanding? My understanding is that Margaret has 11 been involved in the Union Square Building as -- the 12 shows and the theater production activities and 13 14 acting as our representative, and in addition on an uncompensated basis worked through the process of 15 getting the Union Square Building through the 16 Landmark Commission, which, by the way, was a 17 12-year period for which she was paid no money to 18 get it entitled and get the building expanded by 19 20 some 25,000 square feet. The mere ability to get that -- and 21 22 these will be rough numbers -- created enormous 23 value in that building by getting it entitled for 24 redevelopment from the Landmark Commission and 25 getting the -- I think we went from 45,000 square

Litigation Services | 800-330-1112 www.litigationservices.com

## DOUGLAS MCEACHERN - 05/06/2016

		DOUGHAD MCHACHHAI 03/00/2010
	1	Page 263 feet to close to 70,000 square feet approval from
	2	that Landmark Commission.
	3	And then the building and safety
	4	group somebody else just recently gave us
	5	permission to continue and go forward with our
	6	plans.
	7	So the enormous amount of value that was
	8	created in that building was Margaret Cotter working
	9	with her father, as I understand it, and getting the
	10	entitlements.
2000	11	MR. NATION: Could you please read me
	12	the question that started that.
	13	(Whereupon the question was read
	14	as follows:
	15	"Question: And Margaret wanted to
	16	be in charge of developing the
	17	Union Square Theater is your
	18	understanding?")
	19	BY MR. NATION:
	20	Q. All right. So, at the time that
	21	picking up our narrative here, at the time that Jim
	22	Cotter came in as C.E.O
	23	A. Junior?
	24	Q. Jim Cotter, Jr., came in as C.E.O
	25	A. Okay.

Litigation Services | 800-330-1112 www.litigationservices.com

# **EXHIBIT 6**

```
1
 2
                       DISTRICT COURT
 3
                    CLARK COUNTY, NEVADA
 4
    JAMES J. COTTER, JR.,
 5
   individually and
    derivatively on behalf of)
    Reading International,
    Inc.,
                                Case No. A-15-719860-B
 7
            Plaintiff,
                                Coordinated with:
 8
       vs.
                              ) Case No. P-14-082942-E
 9
    MARGARET COTTER, et al.,
10
            Defendants.
11
    and
   READING INTERNATIONAL,
12
    INC., a Nevada
13
   corporation,
14
            Nominal Defendant)
15
16
          VIDEOTAPED DEPOSITION OF MARGARET COTTER
17
                   TAKEN ON MAY 12, 2016
18
                         VOLUME I
19
20
21
22
23
24
    REPORTED BY:
25
     PATRICIA L. HUBBARD, CSR #3400
```

100	
1	Q. What else does it encompass, if
2	anything, beyond live theaters?
3	A. The development of the Union Square and
4	the Cinema 1, 2, 3 currently.
5	Q. When you say "currently," is it
6	anticipated that there will be development of other
7	properties in New York?
8	A. Could be.
9	Q. That would be Orpheum, I guess, right?
10	A. The Orpheum and the Minetta Lane.
11	Q. Prior to the time you became executive
12	vice president of real estate in New York at RDI in
13	March of 2016 or in or about March of 2016, how
14	was your compensation at Liberty Theatres calculated
15	or determined?
16	A. I wouldn't be paid unless there was a
17	show inside the theater.
18	So, I would receive, you know, a small
19	amount of money every month if there was a booked
20	show. And then I would receive 20 percent of the
21	cash flow after a certain break-even at year-end.
22	So it was purely incentive based.
23	Q. The was the Union Square property a
24	source in any way of income for you?
25	A. Yes.

Litigation Services | 800-330-1112 www.litigationservices.com

## **JA3238**

# Confidential - Filed Under Seal

**EXHIBIT 7** 

# **EXHIBIT 8**

Confidential - Filed Under Seal

## **JA3240**

# Confidential - Filed Under Seal

**EXHIBIT 9** 

# **EXHIBIT 10**

```
1
                       DISTRICT COURT
 2
                    CLARK COUNTY, NEVADA
 3
 4 JAMES J. COTTER, JR., )
    individually and
 5
   derivatively on behalf of)
    Reading International,
   Inc.,
                             ) Case No. A-15-719860-B
 7
           Plaintiff,
                             ) Coordinated with:
      vs.
                             ) Case No. P-14-082942-E
   MARGARET COTTER, et al.,
9
10
           Defendants.
    and
11
    READING INTERNATIONAL,
12
    INC., a Nevada
    corporation,
13
            Nominal Defendant)
14
15
16
               DEPOSITION OF: EDWARD KANE
17
                 TAKEN ON: MAY 2, 2016
18
19
20
21
22
23
24
    REPORTED BY:
25
     PATRICIA L. HUBBARD, CSR #3400
```

```
Page 72
 1
           0.
                Then we'll go on.
 2
                Directing your attention, Mr. Kane, back
     to your prior testimony regarding your assessment of
 3
 4
     Margaret Cotter's abilities to handle real estate
 5
     development matters, were you of the view on
     June 12, 2015 when Mr. Jim Cotter, Jr., was
 7
     terminated as president and C.E.O. that Margaret
 8
     Cotter was competent to be the senior executive in
 9
     charge of real estate development activities for
10
     RDI?
                Was I confident?
11
           Α.
                Were you -- in June 12, 2015, when Jim
12
           0.
13
     Cotter, Jr., was terminated as president and C.E.O.,
14
     was it your view then that Margaret Cotter was
     competent to be the senior executive at RDI in
15
16
     charge of its real estate development activities in
     New York?
17
           Α.
                Yes.
18
19
                How long before June 12, 2015 did you
           0.
     come to that conclusion?
20
                It evolved over period of time.
21
           Α.
                                                  I can't
22
     say when.
23
                I do know that I was very impressed with
24
     what she had done with the Landmark Commission,
25
     making development of that property possible and
```

Litigation Services | 800-330-1112 www.litigationservices.com

## EDWARD KANE - 05/02/2016

1	of the professional dynamic between Margaret Cotter
2	on the one hand and Jim Cotter, Jr.?
3	MR. SEARCY: Objection. Vague.
4	THE WITNESS: I was reticent to get
5	involved in that dispute. That's that's family.
6	And I was reticent.
7	But I did tell him that he should
8	Margaret wanted to be an employee for purposes of
9	healthcare. And so long as her contract would not
10	change, I saw no reason why she shouldn't. And I
11	told him that.
12	And he kept giving me, "Oh, I can't
13	attend to it, " kind of I think he sent me an
14	email, "I'm busy, I'll tend to it later." It was an
15	irritant.
16	And it's part of why I thought he was
17	bringing litigation into the board room. There was
18	no reason not to do it. In fact the company would
19	benefit from doing it.
20	BY MR. KRUM:
21	Q. How so?
22	A. How so? Because Margaret Cotter had a
23	contract. And if she was terminated, it's my
24	understanding she would continue to get compensation
25	from plays that were in her theaters, including

Litigation Services | 800-330-1112 www.litigationservices.com

## EDWARD KANE - 05/02/2016

9000		EDWARD RANE - 05/02/2016
	1	Page 170 Stomp.
,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	2	And when we made her employee she gave
	3	that up. But that was a lucrative result.
	4	So I think the company benefited
	5	actually from making her an employee.
mil	6	MR. KRUM: What's our next number?
	7	THE REPORTER: 106 105.
	8	MR. KRUM: I'll ask the court reporter
	9	to mark as Exhibit 105 a two-page document bearing
	10	production number GA5234 and 35.
	11	(Whereupon the document referred
	12	to was marked Plaintiffs'
	13	Exhibit 105 by the Certified
	14	Shorthand Reporter and is attached
	15	hereto.)
	16	BY MR. KRUM:
	17	Q. Mr. Kane, I'm going to ask you only
	18	about your March 1 email in the middle of the first
	19	page of Exhibit 105.
	20	Tell me when you've reviewed that to
	21	your satisfaction.
	22	A. Yes.
	23	Q. Do you recognize Exhibit 105?
	24	A. Yes.
	25	Q. It includes a an email you sent on

Litigation Services | 800-330-1112 www.litigationservices.com

```
1
2
3
                       DISTRICT COURT
                    CLARK COUNTY, NEVADA
5
   JAMES J. COTTER, JR.,
   individually and
    derivatively on behalf of)
  Reading International,
    Inc.,
                               Case No. A-15-719860-B
 8
            Plaintiff,
9
                               Coordinated with:
      VS.
                              ) Case No. P-14-082942-E
10
   MARGARET COTTER, et al.,
11
            Defendants.
12
   and
13 READING INTERNATIONAL,
    INC., a Nevada
14 corporation,
15
            Nominal Defendant)
16
17
            VIDEOTAPED DEPOSITION OF EDWARD KANE
18
                   TAKEN ON JUNE 9, 2016
19
                         VOLUME 3
20
21
22
23
    Job No.: 315759
24
    REPORTED BY:
25
    PATRICIA L. HUBBARD, CSR #3400
```

#### EDWARD KANE - 06/09/2016

```
Page 468
 1
                Directing your attention to the portion
           0.
 2
     of your October 14 -- excuse me -- October 24, 2014
     email as part of Exhibit 291 that concerns
 3
     compensation for Margaret --
 4
 5
           Α.
                Uh-huh.
           0.
                -- Margaret was not an employee of the
 7
     company, correct?
 8
           Α.
                Correct.
 9
                So you were assuming that she was soon
           Q.
10
     to be an employee of the company?
           Α.
                She had asked to be an employee of the
11
12
     company, yes.
13
           0.
                What peer group was used -- strike that.
14
                What peer group, if any, was used this
     year in determining Margaret Cotter's compensation
15
16
     package?
                You mean this year 2016?
17
           A.
18
           0.
                Yes.
                I or Ellen and I or Guy Adams and I -- I
19
           Α.
     don't remember who -- went back to Towers Watson and
20
     said, "We're not happy." And I was not happy, I
21
22
     told them, with the peer group. There were
23
     companies in there whose -- theater companies,
24
     cinema companies whose income -- what we say gross
25
     receipts were ten times ours. And I wanted a peer
```

Litigation Services | 1.800.330.1112 www.litigationservices.com